

# City of Grand Island

Monday, December 30, 2013 Special Meeting

# Item F-1

#9463 - Consideration to Amend and Restate the City of Grand Island, Nebraska Police Officers' Retirement System Plan and Trust

**Staff Contact: Jaye Monter, Finance Director** 

# Council Agenda Memo

**From:** Jaye Monter, Finance Director

Meeting: December 30, 2013

**Subject:** Consideration to Amend and Restate the City of Grand

Island, Nebraska Police Officer's and Firefighter's

Retirement System Plan and Trust

Item #'s: F-1 & F-2

**Presenter(s):** Jaye Monter, Finance Director

## **Background**

On December 17, 2013 Council postponed Ordinances 9463 and 9464 to the December 30, 2013 Special Council meeting in order to allow Council more time to review all restated documents and to allow both pension committees time to meet and review all restated documents with representatives from the law firm who prepared the restated documents.

The retirement plan documents are scheduled to be restated every five years pursuant to IRS Revenue Procedure 2007-44. A restatement must incorporate into the plan documents all changes in the federal tax and state retirement plan laws and regulations since the last restatement.

On December 23, 2013 a joint meeting of the Police and Fire Pension Committees was held for the purposes of reviewing the restated plans. City staff, Thomas McKeon and Will Bradshaw of Fitzgarald, Schorr, Barmettler and Brennan of Omaha were available and answered questions posed by committee members. The Police Pension Committee unanimously approved a resolution expressing their support for the plan. The Fire Pension Committee did not have a quorum present but Chairman Scott Kuehl stated for the record his approval as well.

# **Discussion**

Attached is a synopsis of the changes since the plans were unanimously approved by Council with Ordinances 9195 and 9196 on October 28, 2008. The synopsis and plan document changes were prepared by Thomas McKeon, Attorney at law from Fitzgerald,

Schorr, Barmettler & Brennan, P.C., and L.L.O. in Omaha. Adoption of these changes will insure required compliance of each plan.

Failure to restate the retirement plan documents results in disqualification of the plan and loss of related tax benefits. In simple terms, this means all employees who participate in these plans would lose the tax advantage and would need to pay taxes on their retirement contributions and investment earnings. Disqualification may also result in employer assessed penalties from the IRS.

A copy of both 120 page plan documents and both 23 page Adoption Agreements along with red-line versions is available at the City Clerk's office for review.

### **Alternatives**

It appears that the Council has the following alternatives concerning the issue at hand. The Council may:

- 1. Move to approve
- 2. Refer the issue to a Committee
- 3. Postpone the issue to future date
- 4. Take no action on the issue

## **Recommendation**

City Administration recommends that the Council approve to amend and restate the Police Officer's and Firefighter's Retirement System Plan and Trust.

## **Sample Motion**

Move to approve to amend and restate the Police Officer's and Firefighter's Retirement System Plan and Trust.

# Municipal Retirement Plans 2013 Restatement

#### Synopsis of Changes

The IRS has placed all retirement plans on a staggered restatement schedule. Under the schedule, governmental plans were required to be restated to incorporate all amendments and changes in tax and other laws and related guidance, and if desired filed with the IRS for a determination letter, by January 31, 2009; and then every five years thereafter. In accordance with this schedule, the plan documents again have been restated to incorporate all amendments and changes in tax and other laws and related guidance, and if desired filed with the IRS for a determination letter, by January 31, 2014. Following is a synopsis of some of the primary changes to the restated basic plan document and accompanying adoption agreements since they were last restated.

1. Incorporate changes to tax laws, including the Pension Protection Act of 2006, Heroes Earnings Assistance and Relief Tax Act of 2008 ("HEART Act") and the Worker, Retiree, and Employer Recovery Act of 2008 ("WRERA") and IRS guidance.

#### 2. Changes include:

- Permissible retirement dates for distributions pursuant to federal law.
- Reference revised qualified plan IRS correction procedure.
- Federal law changes to applicable mortality table and interest rate for benefit calculations and limitations.
- HEART Act of 2008 provisions regarding treatment of plan participants who die or become disabled while performing qualified military service.
- Additional revisions regarding:
  - Benefit limitation pursuant to Internal Revenue Code section 415.
  - Expanded rollovers between plans pursuant to IRS guidance.
  - Domestic relations orders not disqualified by certain timing issues.
- Specify basis for exemption of governmental plans from certain qualified plan requirements.
- 3. Update various plan provisions consistent with latest IRS guidance regarding desired language as found in IRS listings of required modifications and cumulative list.
- 4. Incorporate changes made by the Nebraska Legislature in 2012 to statutory provisions governing police retirement systems (LB 1082).
- 5. Other revisions to eliminate superseded provisions or integrate the foregoing changes into the adoption agreements and basic plan document.

337168-1

## City of Grand Island Police and Firefighter Plans Summary/Origin of Changes 2013

**ADOPTION AGREEMENT.** Numbering tracks adoption agreement.

#### A. ESTABLISHMENT, EFFECTIVE DATE, ETC.

(2) **Effective Date**. General effective date of restatement, subject to other effective dates specifically stated in documents.

Origin: Amendment No. 1, Section 3.1.1 – "Plan Years beginning after December 31, 2003"

(4) **Retirement Committee**: Current retirement committee members specified.

#### B, ELIGIBILITY REQUIREMENTS

- (4) **Plan Entry Date**:
  - (b) Clarify plan entry date provisions. Not an issue for police and fire plans providing for immediate plan entry upon employment.

#### C. MANDATORY EMPLOYEE CONTRIBUTIONS

(1) **Amount of Contribution**.

(a)(i) through (iii) L.B. 1082 (2012). Police Plan employee contribution increased 6% to 7% from 2013 to 2014.

Origin: Amendment No. 2, Article 2, Section III.A(2) (a) through (c)

#### E. ALLOCATION OF CITY CONTRIBUTIONS AND FORFEITURES

- (1) **Amount of Contribution.** 
  - (a) LB 1082 (2012). Employer contribution equal to 100% of employee contribution.

Origin: Amendment No. 2, Article 2, Section III.B

(7) "Recognized Compensation" Minor edits to clarify interrelationship of sections (7) and (8) of the adoption agreement and definitions in Section 1 of the basic plan document. Not an issue for police and fire plans that use statutory definition of "salary," as required.

#### H. ROLLOVERS

(1) Rollover contributions by Participants include after-tax contributions, if elected.

Origin: <u>IRS Listing of Required Modifications</u> (10/2011) #51 (Section 4)

#### I. VESTING OF REGULAR ACCOUNTS

#### (2) **Employer Accounts**

(iv) <u>Seven Year Vesting</u> (*Police*): L.B. 1082 (2012). Police plan vesting schedule reduced from 10 to 7 years beginning 7/1/2012.

#### Origin: Amendment No. 2, Article 2, Section IV.D

(4) L.B. 1082 (2012). 100% Vesting upon attaining age 60 *while employed as a police officer*.

#### Origin: Amendment No. 2, Article 2, Section IV.D

(6) In-Service Distributions. In-service distributions before normal retirement date not allowed per federal law. Not an issue for police and fire plans that require termination of employment for distributions.

Origin: Amendment No. 1, Sections 2.1(c), 2.4 and 8.1

**N. ACCELERATED DISTRIBUTIONS.** Change made to reflect prohibition on inservice distributions described in I(6) described above. Not an issue for police and fire plans that require termination of employment for distributions.

Origin: Amendment No. 1, Sections 2.1(c), 2.4 and 8.1 - change reflects prohibition on in-service distributions

**P. MANDATORY DISTRIBUTION ALTERNATIVES.** Clarify consistency of fire and police plans in cashing out vested account balances that are less than \$3,500.

**BASIC MUNICIPAL EMPLOYEES PLAN AND TRUST AGREEMENT.** Numbering tracks basic plan document.

1.1.22 Normal Retirement Date. Normal retirement date must not be less than 55 except if substantially all plan participants are qualified public safety employees.

Origin: Treasury Regulations Section 1.401(a)-1(b)(2)(i) & (v)

1.1.31 Regular Interest. Delete reference to January 1, 1984 per LB 1082

Origin: Amendment No. 2, Article 2, Section II(A)

- 1.1.32 Regular Pay
  - (a) Police Plans. Changes to definition per LB 1082.

Origin: Amendment No. 2, Article 2, Section II(B)

- 2.1 Initial Entry into Plan. Clarifying changes made to plan entry date provisions. Not an issue for police and fire plans that provide immediate plan entry upon employment.
- 3.6 Limitation on Allocations and Benefits (Internal Revenue Code Section 415)

3.6.3,

3.6.6 &

3.6.9(b), (d) Corrections/Definitions for IRC Section 415 Limitations

Origin: IRS Revenue Procedure 2013-12

IRS Listing of Required Modifications (10/2011) #31, Sections

2.3 and 4.7

3.6.9(i) Annual Benefit.

Origin: Amendment No. 1, Section 3.1.1

- I. Benefit Forms Not Subject to section 417(e)(3)
  - a. Limitation Years Beginning Before July 1, 2007.
  - b. Limitation Years Beginning On or After July 1, 2007.

Both "a" and "b": Reference applicable mortality table now found in Appendix B

*Origin:* Amendment No. 1, Section 3.2.2(b) & 3.2.3(b)

II. Benefit Forms Subject to Section 417(e)(3)

a. Annuity Starting Date in Small Plans for Plan Years . . .

Origin: Amendment No. 1, Section 3.3.2

b. Annuity Starting Date in Plan Years Beginning After 2005

Origin: Amendment No. 1, Section 3.3.3

Also, moved the paragraph following subsection "b" to Appendix B, Section B(1).

c. Annuity Starting Date in Plan Years Beginning in 2004 or 2005

Origin: Amendment No. 1, Section 3.3.4(b) and following

- 3.6.9(n) Maximum Permissible Benefit
  - II. Adjustment for Defined Benefit Dollar Limitation for Benefit Commencement Before Age 62 or after Age 65.
    - a. Adjustment of defined benefit dollar limitation for benefit commencement before age 62.
      - II. Limitation Years Beginning On or After July 1, 2007.
        - C. Notwithstanding . . . the age –adjusted dollar limit . . . shall not decrease on account of . . . .

Origin: IRS Listing of Requirement Modifications (10/2011) #40, Section 6.9(b)(i)(II)(c)

3.7 Code Section 436 Benefit Restrictions. Benefit restrictions under Internal Revenue Code Section 436 that apply if a plan is underfunded.

Origin: Amendment No. 1, Sections 2.1(d),2.5 & Article XII

(a) Governmental Plan Exception. Governmental plans are exempt from requirements of Internal Revenue Code Section 436.

Origin: Amendment No. 1, Article XII, Section (a)(2)(B)

3.7.1 2008 and 2009 Plan Years. Internal Revenue Code Section 436 provisions for 2008 and 2009.

Origin: Amendment No. 1, Article XII, Sections (a) through (i) and the end of Article XII

3.7.2 2010 and Later Plan Years. Internal Revenue Code Section 436 provisions for 2010 and later.

Origin: Internal Revenue Service Notice 2011-96 including model Internal Revenue Code Section 436 provisions

- 5.1.3 Distribution of Vested Account Upon Termination of Employment
  - (a) & (d) Cash Out Distribution. Corresponding changes to those described above for paragraph "P" of the Adoption Agreement.
  - (b) Lump sum distribution permitted.

Origin: Amendment No. 2, Article IV, Section A

(f) Participant Distribution Notification.

Origin: Amendment No. 1, Article VI

5.2 Amendment to Comply with Section 415. Substitute "separation from service" with "severance from employment" for certain purposes of the Internal Revenue Code.

Origin: Amendment No. 1, Article XI

- 6.5.1 Same explanation as 5.1.3(a) & (d) above.
- 7.1.1 Distribution of Matured Account. Distribution within 180 days (vs. prior requirement of 90 days).

Origin: Amendment No. 1, Section 6.1

- 7.2 Distribution Requirements
  - 7.2.1 Delete superseded provisions before 2003.
  - 7.2.1.20 2009 Required Minimum Distributions. Suspend 2009 required minimum distributions.

Origin: Amendment No. 1, Article XV

7.2.2 Special Rule for Governmental Plans. Special rule for required minimum distributions.

Origin: 2012 IRS Cumulative List; 2009 final Treasury Regulations 401(a)(9)

7.6 Optional Forms of Distribution. Annuity contract distributed from plan must be nontransferable.

Origin: IRS Listing of Required Modifications (2013) #50

(h) Qualified optional survivor annuity. Optional survivor annuities upon election of surviving spouse.

Origin: Amendment No. 1, Article IX

#### 7.12 Rollovers

#### 7.12.1 Definitions

(a) Eligible rollover distributions. After-tax rollovers allowed.

Origin: Amendment No. 1, Section 5.1

IRS Listing of Required Modifications (2013) #55, Section 2.1

(b) Eligible retirement plan. Expand plans to which direct rollover may be made, including Roth IRA.

Origin: Amendment No. 1, Sections 10.1

IRS Listing of Required Modifications (2013) #55, Section 2.2

(c) Distributee. Allow nonspouse beneficiaries to directly roll over qualifying distributions to IRAs.

Origin: Amendment No. 1, Sections 2.1(b), 2.3 & 4.1 – 4.4

IRS Listing of Required Modifications (2013) #55, Section 2.3

8. Inalienability of Benefits. Qualified domestic relations order does not fail solely due to certain timing issues.

Origin: Amendment No. 1, Article VII

#### 12.3 Retirement Committee

- (i) Police and Fire Plans.
- (4) & (5) Annual and Quadrennial Reports. Moved provisions to Section 12.13 as applicable to all municipal plans for which reporting is required by state law.

- 14.1 Accelerated Distributions.
  - (a) Hardship distribution events. Not an issue for police and fire plans requiring retirement or other termination of employment for distributions.

Origin: IRS Listing of Required Modifications (2011)

15.1 Heroes Earnings Assistance and Relief Tax Act of 2008. Required changes for participants who die or are disabled while performing qualified military duty.

Origin: Amendment No. 1, Sections 2.1(e), 2.6 & Article XIII IRS Listing of Required Modifications (10/2011) #86

Appendix B Actuarial Equivalencies and Actuarial and Mortality Assumptions

A. January 1, 2008 effective date of applicable interest rate and mortality table under Internal Revenue Code Section 417(e), as amended by the Pension Protection Act of 2006 ("PPA 2006"). Special effective date for mortality table for Internal Revenue Code Section 415 purposes – "plan years beginning after December 31, 2008."

Origin: Amendment No. 1, Section 14.1 for general 1/1/08 effective date

Amendment No. 1, Sections 2.1(a) & 3.3.3(c) for special mortality table – "plan years beginning after December 31, 2008."

1. **Applicable Interest Rate**. Applicable interest rate under Internal Revenue Code Section 417(e), as amended by PPA 2006.

Origin: Amendment No. 1, Sections 2.1(f), 2.7 & 14.2

2. **Applicable Mortality Table**. Applicable mortality table under Internal Revenue Code Section 417(e), as amended by PPA 2006.

Origin: Amendment No. 1, Sections 2.1(a), 2.2, 3.3.3(c) & 14.3

B. Applicable interest rate and mortality table in effect before January 1, 2008 (January 1, 2009 for mortality table for Internal Revenue Code Section 415 purposes).

Origin: Internal Revenue Code Section 417(e) and applicable IRS guidance before being amended

# BASIC MUNICIPAL EMPLOYEES PLAN AND TRUST AGREEMENT

# **FOR**

# CITY OF GRAND ISLAND, NEBRASKA POLICE OFFICERS' RETIREMENT SYSTEM PLAN AND TRUST

(201308)

<del>328780</del>334458

#### BASIC MUNICIPAL EMPLOYEES

#### PLAN AND TRUST AGREEMENT

# [REVISE TABLE OF CONTENTS WHEN FINALIZING DOCUMENT]

SECTION 1 - Introduction				
1.1	Definiti	ions	1	
	1.1.1	Accounts  (a) Total Account  (b) Employer Account  (c) Employee Account  (d) Matured Account	1 1	
	1.1.2	Actuarial Equivalent	1	
	1.1.3	Agreement	1	
	1.1.4	Annual Valuation Date	2	
	1.1.5	Annuity Contract	2	
	1.1.6	Beneficiary	2	
	1.1.7	City	2	
	1.1.8	Effective Date	2	
	1.1.9	Eligibility Service	2 2	
	1.1.10	Employee	3	
	1.1.11	Employer	3	
	1.1.12	Event of Maturity	3	
	1.1.13	Fire Plan	3	
	1.1.14	Forfeiture	3	
	1.1.15	Fund	3	
	1.1.16	Funding Agent		
	1.1.17	Highly Compensated Employee		
	1.1.18	Hours of Service	4	

	(c) Back Pay	
	(e) Special Rules	5
1.1.19		
	Leased Employee	
1.1.20	Limitation Year	6
1.1.21	Net Earnings	6
1.1.22	Normal Retirement Date	6
1.1.23	One Year Break in Service	6
1.1.24	Participant	6
1.1.25	Plan	6
1.1.26	Plan Year	6
1.1.27	Police Plan	6
1.1.28	Prior Plan Statement	6
1.1.29	Recognized Compensation	7
1.1.30	Recognized Employment	8
1.1.31.	Regular Interest	8
1.1.32	Regular Pay	
	(a) Police Plans	
1 1 22		
1.1.33	Retirement Committee or Committees	9
1.1.34	Retirement Date	9
1.1.35	Retirement Value	9
1.1.36	Salary	
	(a) Police Plans	
1 1 27		
1.1.37	Straight Life Annuity	, <b>.</b> 10
1.1.38	Supplemental Effective Date	10
1.1.39	Taxable Wage Base ("TWB")	10
1.1.40	Tax Rate	10
1.1.41	Trustee	10

		1.1.42 Valuation Date	11
		1.1.43 Vested	11
		1.1.44 Vesting Service	11 11 11
	1.2	Special Rules for Employer's Computing Service on the Basis of Elapsed Time	
	1.3	Gender, Number and Headings	12
	1.4	Return of Employer Contribution	12
	1.5	Amendment of Prior Plan Statement	13
	1.6	Supplemental Elections, Information and Plan Provisions	13
SEC	CTION 2	- Eligibility and Participation	13
	2.1	Initial Entry into Plan	13
	2.2	Special Rules for Former Participants and Eligible Classes	13
	2.3	Annual Certification	13
SEC	CTION 3	- Contributions and Allocation Thereof	14
	3.1	Employee Contributions	14
	3.2	Employer Contributions  (a) Police and Fire Plans  (b) Other Municipal Plans  (i) Nonintegrated  (ii) Integrated  (c) Mistake  (d) Return of Contribution	15 15 15 15
	3.3	Eligible Participants  (a) Police and Fire Plans  (b) Other Municipal Plans	17
	3.4	Forfeitures	18
	3.5	Rollover Contributions  (a) Police and Fire Plans  (i) Police  (ii) Fire  (b) Other Municipal Plans	18 18 18
	3.6	(c) Merger of Services  Limitation on Allocations and Benefits	
	5.0	Lithianon on Anocanons and Delicitly	1 9

3.6.1	Sing	le Plan	19
3.6.2	Dete	ermination of Maximum Permissible Amount	19
3.6.3	Disp (a) (b)	Position of Excess Amount	20
3.6.4	Mast	ter or Prototype Plans	21
3.6.5	Amo	ount Attributable to this Plan	21
3.6.6	Corr	ection of Excess Amount or Benefit	22
3.6.7	Othe	er Defined Contribution Plans	22
3.6.8	Othe	er Defined Benefit Plans	22
3.6.9	Defi	nitions and Special Rules	22
	(a)	Annual Additions	
	(b)	Compensation	25 25 26
	(c)	Employer	
	(d)	Excess Amount	27
	(e)	Master or Prototype Plan	27
	(f)	Maximum Permissible Amount	27
	(g)	Aggregation and Disaggregation of Plans	28
	(h)	Predecessor Employer	29
	(i)	Annual Benefit	29
	(j)	RPA '94 Old Law Benefit	31
	(k)	Employee Contributions	31
	(1)	Benefit Increases	31
	(m)	Limitation Years	31
	(n)	Maximum Permissible Benefit	31

SECTION 4 -	- Valuation, Adjustment and Investment of Accounts	35
4.1	Annual Valuation and Adjustment of Accounts	35
4.2	Intermediate Valuations and Adjustments	35
4.3	Management and Investment of Fund	35
	(a) Police Plan	
	(b) Fire Plan	35
4.4	All Plans	36
4.5	Individual Subfunds	36
SECTION 5 -	- Vesting	36
5.1	Employer Accounts	36
	5.1.1 Progressive Vesting	36
	5.1.2 Full Vesting	36
	5.1.3 Distribution of Vested Account Upon Termination of Employment	37
	(a) Cash Out Distributions	
	(b) Non Cash-Out Distributions	37
5.2.	Amendment to Comply with Section 415	39
5.3	Other Accounts	39
SECTION 6	Maturity and Distributable Events	39
6.1	Events of Maturity	39
6.2	Determination of Matured Benefit	39
6.3	Effect of Maturity Upon Further Participation in the Plan	39
6.4	Disposition of Nonvested Portion of Account	40
	6.4.1. Police and Fire Plans	40
	6.4.2 All Other Plans - Reallocation to Other Participants	40
	6.4.3 Reallocation Date	40
	6.4.4 Restoration	40
6.5	Other Distributable Events	40
	6.5.1 Termination of Employment - Police and Fire Plans	41
	6.5.2 Early Retirement Swing Employment - Fire Plans	41

SECTION 7 –	Distributi	on		41
7.1	Time o	f Distribution	on	41
	7.1.1	Distributi	on from Matured Account	41
7.2	Distrib	ution Requi	rements	42
	7.2.1		ons in calendar years beginning after December 31, 1984 and ending nuary 1, 2003	42
		7.2.1.1	Supersession	42
		7.2.1.2	Requirements	42
		7.2.1.3	Required Beginning Date	42
		7.2.1.4	Limits on Distribution Periods	42
		7.2.1.5	Determination of Amount to be Distributed Each Year	42
		7.2.1.6	Distribution Beginning before Death	43
		7.2.1.7	Distribution Beginning after Death	43
		7.2.1.8	Applicable Life Expectancy	43
		7.2.1.9	Designated Beneficiary	44
		7.2.1.10	Distribution Calendar Year	44
		7.2.1.11	Life Expectancy	44
		7.2.1.12	Participant's Benefit	44
		7.2.1.13.	Required Beginning Date	44
		7.2.1.14	Transitional Rule	44
	7.2.2	Distrib	utions in calendar years beginning after December 31, 2002	45
		7.2.2.1	Supersession	45
		7.2.2.2	Requirements	45
		7.2.2.3	Required Beginning Date	46
		7.2.2.4	Participant's Death Before Distribution Begins	46
		7.2.2.5	Forms of Distribution	46
		7.2.2.6	Amount of Required Minimum Distribution for Each Distribution Calendar Year	46
		7.2.2.7	Lifetime Required Minimum Distributions Continue Through	

			Year of Participant's Death	47
		7.2.2.8	Death On or After Date Distributions Begin	47
		7.2.2.9	Death Before Date Distributions Begin	47
		7.2.2.10	Designated Beneficiary	48
		7.2.2.11	Distribution Calendar Year	48
		7.2.2.12	Life Expectancy	48
		7.2.2.13	Participant's Account Balance	48
		7.2.2.14	Required Beginning Date	48
		7.2.2.15	Participants or Beneficiaries Permitted to Elect 5-Year Rule	48
		7.2.2.16	Election to Apply 5-Year Rule to Distributions to Designated Beneficiarie	es.48
		7.2.2.17		49
			7.2.2.17.1 Determination of Amount to be Distributed Each Year	49
			7.2.2.17.2 Requirements For Annuity Distributions That Commence During Participant's Lifetime	50
		7.2.2.17.3	Requirements for Minimum Distributions After Participant Dies	51
		7.2.2.17.4	Changes to Annuity Payment Period	53
		7.2.2.17.5	Payments to Surviving Child	53
		7.2.2.17.6	Additional Definitions	53
	7.2.2.1	8 TEFRA se	ection 242(b)(2) Elections	53
			Rules	
7.3	Minim	um Benefits	Upon Retirement, Death and Disability of Certain Police and Fire	
	7.3.1		Retirement Benefits	
		` /	ice Plan	
		` '	e Plan irement Benefits for Certain Firefighters Employed on August 7, 1965	
	7.3.2	Minimum	Death Benefits	57
		(a) Dea	ath in the Line of Duty – Police and Fire Plans	57
		(i	<b>,</b>	
		,	ii) Fire Plan	
		* *	ath Not in the Line of Duty – Police and Fire Plans	
		(i	<b>,</b>	
		(1	ii) Fire Plan	58

		7.3.3 Minimum Disability Benefits	58	
	7.4	Effect of Reemployment After Distribution Has Been Made or Commenced	59	
	7.5	Designation of Beneficiaries		
	7.6	Optional Forms of Distribution		
	7.7	Distribution in Kind	61	
	7.8	Facility of Payment	61	
	7.9	Withdrawal from Voluntary Account	62	
	7.10	When a Participant or a Beneficiary Cannot Be Found	62	
	7.11	Determination of Required Distributions	62	
	7.12	Rollovers	62	
		7.12.1 Definitions	62	
SECT	ION 8	Inalienability of Benefits	63	
SECT	ION 9	Amendment and Termination	64	
	9.1	Amendment	64	
	9.2	Discontinuance of Contributions and Termination of Plan	64	
	9.3	Limitations	64	
	9.4	Merger, Etc., with Another Plan	64	
SECT	ION 10 -	- Concerning the Trustee	65	
	10.1	Dealings with the Trustee	65	
	10.2	Fees and Expenses from Fund.	65	
	10.3	Fire Plans	65	
	10.4	Resignation and Removal of Trustee	66	
	10.5	Accountings by Trustee	66	
	10.6	Trustee's Power to Protect Itself on Account of Taxes	66	
	10.7	Fiduciary Duties	67	
	10.8	Prohibited Transactions	67	
	10.9	Indemnity	67	
	10.10	Investment in Insurance	68	

		(a) Ordinary Life	
		(b) Term and Universal Life	
		(c) Combination	00
	10.11	Directed Investments	69
		10.11.1 Employer Direction	69
		10.11.2 Participant Direction of Investments	69
	10.12	Investment Managers	60
	10.12	(a) Police Plan	
		(b) Fire Plan	
		(c) All Other Plans	70
		(d) All Plans	70
	10.13	Participant Loans	70
	10.14	Other Powers	71
SECT	ION 11 -	Certifications - Rules and Regulations	74
	11.1	Certificates of Fact by Employer	74
	11.2	Rules and Regulations	74
	11.3	Method of Executing Instruments	74
	11.4	Claims Procedure	75
		(a) Original Claim	75
		(b) Claim Review Procedure	75
		(c) General Rules	75
	11.5	Information Furnished by Participants	75
SECT	TON 12 -	Plan Administration	76
	12.1	Trustee	76
	12.2	Delegation of Duties	76
	12.3	Retirement Committee	76
	12.4.	Benefits Errors	79
	12.5	Employer Action	79
	12.6	Limitation on Authority	79
	12.7	Dual Capacity	79
	12.8	Administrator	79
	12.9	Named Fiduciaries	79

	12.10	Service of Process	79
	12.11	Conflict of Interest	79
	12.12	Residual Authority	80
SECTI	ON 13	In General	80
	13.1	Disclaimers	80
	13.2	Duration of Fund	80
	13.3	Continuity	81
	13.4	State Law	81
	13.5	Execution in Counterparts	81
SECTI	ON 14	Accelerated Distributions	81
	14.1	Accelerated Distributions	81
	14.2	Qualified Participant	81
	14.3	Amount of Accelerated Distribution	81
	14.4	Distributions After Normal Retirement Date	81
	14.5	No Forfeitures	81
SECTI	ON 15	Uniformed Services Employment and Reemployment Rights Act ("USERRA")	82

#### BASIC MUNICIPAL EMPLOYEES

#### PLAN AND TRUST AGREEMENT

This Basic Municipal Employees Plan and Trust Agreement consisting of a Defined Contribution Plan and Trust Agreement and the related Adoption Agreement(s) as executed by the City and Trustee constitute and are integral parts of a retirement plan and trust established by the City and are designed to comply with the provisions of the Internal Revenue Code of the United States (the "Code") dealing with plans and trusts of such character and, in particular, Sections 401(a) and 501(a) thereof and including without limitation sections 401(a)(9), 401(a)(9), 401(a)(16), 401(a)(31), and 401(a)(37).

#### SECTION 1

#### Introduction

- 1.1 <u>Definitions</u>. As used herein, the following terms shall have the following meanings:
- 1.1.1 <u>Accounts</u> As indicated in the Adoption Agreement, the following Accounts or Account classifications may be maintained under this Plan, in addition to any other account specified elsewhere herein:
  - (a) <u>Total Account</u> A Participant's entire interest in the Plan, including his Employer Account and his Employee Account.
  - (b) <u>Employer Account</u> The Account maintained for each Participant to which is credited his allocable share of the Employer's contributions and forfeitures under the Plan, together with any increase or decrease thereon.
  - (c) <u>Employee Account</u> The Account maintained for each Participant to which is credited his Employee Contributions under the Plan, together with any increase or decrease thereon.
  - (d) <u>Matured Account</u> The portion of a Participant's Total Account which is Vested in him upon the occurrence of an Event of Maturity, together with an increase or decrease thereon.

The Trustee shall not be required to maintain separate investments for any Account; provided, however, no investments in Insurance Contracts shall be made from Voluntary Contributions Accounts. The portion of the Account balance derived from nondeductible employee contributions is the Employee's total Account balance multiplied by a fraction, the numerator of which is the total amount of nondeductible employee contributions less withdrawals and the denominator of which is the sum of the numerator and the total contributions made by the Employer on behalf of the Employee less withdrawals. For this purpose, contributions include contributed amounts used to provide ancillary benefits and withdrawals include only amounts distributed to the Employee and do not reflect the cost of any death benefits.

1.1.2 <u>Actuarial Equivalent</u> - Equality in value of the aggregate amount of benefit expected to be received under different forms of benefit or at different times determined as of a given date as adopted by the City or the Retirement Committee for use by the Plan. Actuarial Equivalencies shall be specified in or determined in accordance with Appendix B. If benefits under the Plan are obtained through the purchase of an annuity contract, the Actuarial Equivalent of any such form of benefit shall be the amount of the benefit which can be purchased or otherwise provided by the Participant's Retirement Value. All actuarial and mortality assumptions adopted by the City or the Retirement Committee shall be on a sex-neutral basis and such assumptions shall be stated in Appendix B.

- 1.1.3 <u>Agreement</u> The "Basic Municipal Employees Plan and Trust Agreement" and the related "Adoption Agreement" as approved and adopted by the Employer as the official statement of its qualified retirement plan to be effective from and after the Effective Date (if the Employer's adoption of the Agreement is the establishment of a new qualified retirement plan) or the Supplemental Effective Date (if the Employer's adoption is the amendment and restatement of a preexisting qualified retirement plan), as the same may be from time to time thereafter amended.
- 1.1.4 <u>Annual Valuation Date</u> Unless indicated otherwise herein or in the Adoption Agreement, December 31.
- 1.1.5 <u>Annuity Contract</u> The contract or contracts issued by one or more life insurance companies or designated trusts and purchased by the Plan in order to provide any of the benefits described in this Agreement. Annuity conversion rates in any such contract shall be specified on a sex-neutral basis.
- 1.1.6 <u>Beneficiary</u> The person or persons designated by a Participant in a written instrument designated by and filed with the Retirement Committee prior to the Participant's death to receive death benefits which may be payable under the Plan.
- 1.1.7 <u>City</u> The municipality chartered in the state of Nebraska which adopts this Agreement by executing the Adoption Agreement.
- 1.1.8 <u>Effective Date</u> The date (set forth in the Adoption Agreement) as of which the Employer first established this Plan.
- 1.1.9 <u>Eligibility Service</u> A measure of an Employee's service with the Employer (stated as a number of years) which is equal to the number of computation periods in which the Employee is credited with one thousand (1,000) or more Hours of Service; subject, however, to such of the following rules as are applicable under the Adoption Agreement:
  - (a) <u>Computation Periods</u> The computation periods for determining the Employee's Eligibility Service (and One-Year Breaks in Service as applied to his Eligibility Service) shall be (as indicated in the Adoption Agreement) either:
    - (i) The twelve (12) consecutive month period beginning with the date the Employee first performs an Hour of Service plus all Plan Years beginning after the date the employee first performs an Hour of Service (irrespective of any termination of employment and subsequent reemployment), or
    - (ii) The twelve (12) consecutive month period beginning with the date the Employee first performs an Hour of Service plus all twelve (12) consecutive month periods commencing on annual anniversaries of such date (irrespective of any termination of employment and subsequent reemployment).

If, in the Adoption Agreement, the computation periods selected for the computation of years of Eligibility Service are determined under Section 1.1.9(a)(i) above, an Employee who is credited with one thousand (1,000) Hours of Service in the computation period beginning on the date the Employee first performs an Hour of Service and also is credited one thousand (1,000) Hours of Service in the computation period which is the first Plan Year beginning after the date the Employee first performs an Hour of Service, he shall be credited with two (2) years of Eligibility Service notwithstanding that the same Hours of Service are counted in both (overlapping computation periods).

(b) <u>Completion</u> - A year of Eligibility Service shall be deemed completed only as of the last day of the computation period irrespective of the date in such period that the Employee completes one thousand (1,000) Hours of Service. (Fractional years of Eligibility Service shall not be credited.)

Grand Island

(c) <u>Eligibility Rule of Parity</u> - In the case of a Participant who does not have any nonforfeitable right to the account balance derived from Employer Contributions, Years of Service before a period of Consecutive one-year breaks in service will not be taken into account in computing eligibility service if the number of consecutive one-year breaks in service in such period equals or exceeds the greater of five or the aggregate number of years of service. Such aggregate number of years of service will not include any years of service disregarded under the preceding sentence by reason of prior breaks in service.

If a participant's years of service are disregarded pursuant to the preceding paragraph, such participant will be treated as a new employee for eligibility purposes. If a participant's years of service may not be disregarded pursuant to the preceding paragraph, such participant shall continue to participate in the plan, or, if terminated, shall participate immediately upon reemployment.

1.1.10 <u>Employee</u> - Any individual who is a paid employee of the Employer maintaining the Plan as defined in the Adoption Agreement.

The term Employee shall also include any Leased Employee deemed to be an employee of any employer described in the previous paragraph as provided in Code section 414.

1.1.11 <u>Employer</u> - The City which adopts this Agreement by executing the Adoption Agreement.

The Employer is the "named fiduciary", as that term is defined in ERISA, for all purposes of this Plan, to the extent that the provisions of ERISA regarding "named fiduciary" are applicable to this Plan.

- 1.1.12 <u>Event of Maturity</u> Any of the occurrences described in Section 6 by reason of which a Participant or Beneficiary may become entitled to a distribution from the Plan.
- 1.1.13 Fire Plan A plan created and maintained pursuant to Neb. Rev. Stat. Sections 16-1020 to 16-1042 for the purpose of investing payroll deductions and contributions by a city of the first class to a retirement system established for firefighters of such city, the provisions of which statutes, as enacted or amended from time to time, are incorporated herein by this reference and shall control over any inconsistent terms hereof so long as not contrary to the law applicable to plans intended to be qualified under Internal Revenue Code Section 401(a).
- 1.1.14 <u>Forfeiture</u> The portion of a Participant's Employer Account, if any, which is not Vested in him upon an Event of Maturity.
- 1.1.15 <u>Fund</u> The assets of the Plan held by the Trustee from time to time, including all assets initially transferred to and held by the Trustee, together with all subsequent contributions of the Employer and the Participants and the investments and reinvestments, earnings, gains and losses thereon.
- 1.1.16 <u>Funding Agent</u> Any bank, trust company, life insurance company, thrift institution, credit union, or investment management firm selected by the City or Retirement Committee (subject to the approval of the City for Fire Plans) to hold or invest the funds of the Plan.
- 1.1.17 <u>Highly Compensated Employee</u> The term "Highly Compensated Employee" includes highly compensated active Employees and highly compensated former Employees.

Effective for years beginning on or after January 1, 1997, aA highly compensated active employee includes any Employee who had compensation from the Employer in the preceding year in excess of \$80,000, as adjusted from time to time in the same manner as under Code section 415(d), except that the base period is the calendar quarter ending September 30, 1996. Compensation for purposes of this section shall be as defined in Code section 415(c)(3) and determined by the Employer.

The determination of whether a former employee is aA highly compensated former Employee is based on the rules applicable to determining Highly Compensated Employee status in effect for the determination year, in accordance with Treasury Regulation section 1.414(q)-IT, A-4 and Notice 97-45 as updated from time to time.

The determination of who is a Highly Compensated Employee will be made in accordance with Section 414(q) of the Code and the regulations thereunder.

- 1.1.18 <u>Hours of Service</u> A measure of an Employee's service with the Employer (in Recognized Employment with respect to Police and Fire Plans) determined for a particular computation period, equal to the sum of the number of hours credited to the Employee according to the election made in the Adoption Agreement and the following rules:
  - (a) Paid Duty An Hour of Service shall be credited for each hour for which the Employee is paid, or entitled to payment, for the performance of duties for the Employer. These Hours of Service shall be credited to the Employee for the computation period or periods in which the duties are performed.
  - (b) <u>Paid Nonduty</u> An Hour of Service shall be credited for each hour for which the Employee is paid, or entitled to payment, by the Employer on account of a period of time during which no duties are performed (irrespective of whether the employment relationship has terminated) due to vacation, holiday, illness, incapacity (including Disability), layoff, jury duty, military duty or leave-of-absence; provided, however, that:
    - (i) No more than five hundred one (501) Hours of Service shall be credited on account of a single continuous period during which the Employee performs no duties (whether or not such period occurs in a single computation period);
    - (ii) No Hours of Service shall be credited on account of payments made under a plan maintained solely for the purpose of complying with the applicable worker's compensation, employment compensation or disability insurance laws;
    - (iii) No Hours of Service shall be credited on account of payments which solely reimburse the Employee for medical or medically related expenses incurred by the Employee;
    - (iv) Payments shall be deemed made by or due from the Employer, whether made directly or indirectly from a trust fund or an insurer to which the Employer contributes or pays premiums.

These Hours of Service shall be credited to the Employee for the computation period for which payment is made or, if the payment is not computed by reference to units of time, the hours shall be credited to the first computation period in which the event for which any part of the payment is made occurred.

- (c) <u>Back Pay</u> An Hour of Service shall be credited for each hour of which back pay, irrespective of mitigation of damages, has been either awarded or agreed to by the Employer. The same Hours of Service credited under paragraphs (a) or (b) shall not be credited under this paragraph (c). The crediting of Hours of Service under this paragraph (c) for periods and payments described in paragraph (b) shall be subject to all the limitations of that paragraph. These hours shall be credited to the Employee for the computation period or periods to which the award or agreement pertains, rather than the computation period in which the award, agreement or payment is made.
  - (d) <u>Unpaid Absences</u> For all purposes of this Plan:

- (i) <u>Leaves-of-Absence</u> An assumed eight (8) hour day and forty (40) hour week shall be credited during each unpaid leave-of-absence authorized by the Employer under uniform rules of non-discriminatory application; provided, however, that, if the Employee does not return to employment for any reason other than death, Disability or attainment of his Normal Retirement Date at the expiration of the leave-of-absence, such Hours of Service shall not be credited.
- (ii) <u>Military Leaves</u> An assumed eight (8) hour day and forty (40) hour week shall be credited during service in the Armed Forces of the United States if the Employee both entered such service and returned to employment with the Employer from such service under circumstances entitling him to reemployment rights granted veterans under federal law; provided, however, that, if the Employee does not return to employment for any reason other than death, Disability or attainment of Normal Retirement Date within the time prescribed by law for the retention of veteran's reemployment rights, such Hours of Service shall not be credited.
- Special Rules Department of Labor regulations at 29 C.F.R. Sec. 2530.200b-2 are hereby, by reference, incorporated herein for the purpose of calculating and crediting Hours of Service; though neither ERISA nor said regulations otherwise are applicable if this is a governmental plan as defined in Code section 414(d). For periods prior to the ERISA date, Hours of Service may be determined using whatever records are reasonably accessible and by making whatever calculations are necessary to determine the approximate number of Hours of Service completed during such prior period. If this Plan is maintained by the Employer as the plan of a predecessor employer, Hours of Service will be credited for employment with that predecessor employer as if it were the Employer. Hours of Service shall be credited for employment with other members of an affiliated service group (under Code section 414(m)), a controlled group of corporations (under Code section 414(b)), or a group of trades or businesses under common control (under Code section 414(c)), of which the adopting employer is a member, and any other entity required to be aggregated with the Employer pursuant to Code section 414(o) and the regulations thereunder. Hours of Service will also be credited for any individual considered an Employee for purposes of this Plan under Code section 414(n) or Code section 414(o) and regulations thereunder.
- (f) <u>Maternity and Paternity Absences</u> Solely for purposes of determining whether a Break in Service, as defined in Section 1.1.23, for participation and vesting purposes has occurred in a computation period, an individual who is absent from work for maternity or paternity reasons shall receive credit for the Hours of Service which would otherwise have been credited to such individual but for such absence, or in any case in which such hours cannot be determined, eight hours of service per day of such absence. For purposes of this paragraph, an absence from work for maternity or paternity reasons means an absence (1) by reason of the pregnancy of the individual, (2) by reason of a birth of a child of the individual, (3) by reason of the placement of a child with the individual in connection with the adoption of such child by such individual, or (4) for purposes of caring for such child for a period beginning immediately following such birth or placement. The Hours of Service credited under this paragraph shall be credited (1) in the computation period in which the absence begins if the crediting is necessary to prevent a Break in Service for that period, or (2) in all other cases, in the following computation period.
- 1.1.19 <u>Leased Employee</u> <u>Effective for years beginning on or after January 1, 1997, tThe term "Leased Employee" means any person (other than an Employee of the recipient) who pursuant to an agreement between the recipient and any other person ("leasing organization") has performed services for the recipient (or for the recipient and related persons determined in accordance with Code section 414(n)(6)) on a substantially full time basis for a period of at least one year, and such services are performed under primary direction or control by the recipient. Contributions or benefits provided a Leased</u>

Employee by the leasing organization which are attributable to services performed for the recipient Employer shall be treated as provided by the recipient Employer.

A Leased Employee shall not be considered an Employee of the recipient if: (i) such Employee is covered by a money purchase pension plan providing: (1) a nonintegrated Employer contribution rate of at least 10 percent of compensation, as defined in Code section 415(c)(3), but including amounts contributed pursuant to a salary reduction agreement which are excludable from the Employee's gross income under Code section 125, section 402(a)(8), section 402(h) section 403(b), or, effective January 1, 2001, Code section 132(f)(4), (2) immediate participation, and (3) full and immediate vesting; and (ii) Leased Employees do not constitute more than 20 percent of the recipient's nonhighly compensated work force.

- 1.1.20 <u>Limitation Year</u> A calendar year, or the twelve (12) consecutive month period selected by the Employer in the Adoption Agreement. All qualified plans maintained by the Employer must use the same Limitation Year. If the Limitation Year is amended to a different twelve (12) consecutive month period, the new Limitation Year must begin on a date within the Limitation Year in which the amendment is made. Effective for Limitation Years beginning on or after July 1, 2007, the Limitation Year may only be changed by a Plan amendment. Furthermore, if the Plan is terminated effective as of a date other than the last day of the Plan's Limitation Year, then the Plan is treated as if the Plan had been amended to change its Limitation Year.
- 1.1.21 <u>Net Earnings</u> The amount by which income or gain realized from the investments of the Fund exceeds the amount of any realized losses from such investments during the calendar year.
- 1.1.22 <u>Normal Retirement Date</u> As indicated in the Adoption Agreement: (i) the date upon which a Participant attains his sixty-fifth (65th) birthday or, (ii) the later of the date a Participant attains age sixty-five (65) or the fifth (5th) anniversary of the date he first commenced Participation in the Plan, or (iii) such earlier time as the Employer may designate (though not less than 55 unless substantially all participants in the Plan are qualified public safety employees as defined in applicable regulations and other guidance). The Normal Retirement Date may not be later than the age at which the Employer requires Employees eligible to participate in the Plan to retire on account of attained age. The normal form of benefit is a single life annuity commencing on the normal retirement date.
- 1.1.23 One Year Break in Service A computation period during which the Participant has not completed more than five hundred (500) Hours of Service. (A One-Year Break in Service shall be deemed to occur only on the last day of such computation period. Unless otherwise specified, the computation period shall be the Plan Year.)
- 1.1.24 <u>Participant</u> An Employee of the Employer who becomes a Participant in the Plan pursuant to Section 2 hereof. An Employee who has become a Participant shall be considered to continue as a Participant in the Plan until the date of his death, or, if earlier, the earliest date as of which the Participant no longer has any Account under the Plan.
- 1.1.25 <u>Plan</u> The qualified retirement plan and trust of the Employer maintained pursuant to this Agreement, as amended from time to time (and which may have been previously in existence pursuant to a Prior Plan Statement).
- 1.1.26 <u>Plan Year</u> The twelve (12) consecutive month period ending on any Annual Valuation Date or as otherwise indicated in the Adoption Agreement.
- 1.1.27 <u>Police Plan</u> A plan created and maintained pursuant to Neb. Rev. Stat. Sections 16-1001 to 16-1019 for the purpose of investing payroll deductions and contributions by a city of the first class to a retirement system established for police officers of such city, the provisions of which statutes, as enacted or amended from time to time, are incorporated herein by this reference and shall control over any inconsistent terms hereof so long as not contrary to the law applicable to plans intended to be qualified under Internal Revenue Code Section 401(a).

- 1.1.28 Prior Plan Statement The written instrument or instruments or the series of written instruments under which this Plan was established and maintained from time to time prior to the Supplemental Effective Date. (If this Plan was first established by the Employer's approval and adoption of this Agreement, there will have been no Prior Plan Statement and all references thereto shall be disregarded.) To the extent a Prior Plan Statement exists, benefits provided to Participants may be affected by the provisions of such Prior Plan Statement.
- 1.1.29 Recognized Compensation As elected by the Employer in the Adoption Agreement, Recognized Compensation will mean all of each Participant's (a) W-2 earnings or (b) wages as defined under Code Section 3401(a), (c) compensation (as that term is defined in Code section 415(c)(3) and Treasury Regulations, section 1.415(c)-2(d)(2)), or (d) such other compensation as specified in the For any self-employed individual covered under the Plan, Recognized Adoption Agreement. Compensation will mean earned income. Except as elected elsewhere in this Plan, Recognized Compensation shall include only that compensation which is actually paid to the Participant during the applicable period, and the applicable period shall be the period elected by the Employer in the Adoption Agreement. If the Employer makes no election, the applicable period shall be the Plan Year.

Notwithstanding the above, if elected by the Employer in the Adoption Agreement, Recognized Compensation shall include any amount which is contributed by the Employer pursuant to a salary reduction agreement and which is not includable in the gross income of the Employee under Code sections 125, 402(e)(3), 402(h)(1)(B), 403(b), or, effective January 1, 2001, Code section 132(f)(4). Effective for Plan and Limitation Years beginning on or after January 1, 1998, reference to section 125 (unless otherwise provided in the Adoption Agreement) shall include deemed section 125 compensation. Deemed section 125 compensation is an amount that is excludable under Code section 106 that is not available to a participant in cash in lieu of group health coverage under a section 125 arrangement solely because the participant is unable to certify that he or she has other health coverage. Amounts are deemed section 125 compensation only if the Employer does not request or collect information regarding the Employee's other health coverage as part of the enrollment process for the health plan.

For Plan Years commencing after December 31, 1988 and before January 1, 1996, the Annual Recognized Compensation of each Participant taken into account under the Plan for any year shall not exceed \$200,000, as adjusted by the Secretary at the same time and in the same manner as under Code section 415(d), except that the dollar increase in effect on January 1 of any calendar year is effective for plan years beginning with or within such calendar year and the first adjustment to the \$200,000 limitation is effective on January 1, 1990. For Plan Years beginning on or after January 1, 1996 and before January 1, 2002, the Annual Recognized Compensation of each Employee taken into account under the Plan shall not exceed \$150,000, as adjusted by the Commissioner for increases in the cost-of-living in accordance with Code section 401(a)(17)(B).

For any Plan Year beginning on or after January 1, 2002, the annual compensation of each Employee taken into account under this Plan shall not exceed \$200,000, as adjusted for cost-of-living increases in accordance with Section 401(a)(17)(B) of the Code. Annual compensation means compensation during the Plan Year or such other consecutive 12-month period over which compensation is otherwise determined under the Plan (the determination periods). The cost-of-living adjustment in effect for a calendar year applies to annual compensation for the determination period that begins with or within such calendar year.

If a determination period consists of fewer than 12 months the annual compensation limit is an amount equal to the otherwise applicable annual compensation limit multiplied by a fraction, the numerator of which is the number of months in the short determination period, and the denominator of which is 12.

If compensation for any prior determination period is taken into account in determining an Employee's allocations or benefits for the current Plan Year, the compensation for such prior determination period is subject to the applicable annual compensation limit in effect for that prior period. For this purpose, in determining allocations or benefits in Plan Years beginning on or after January 1, 1989 and before January 1, 1996, the annual compensation limit in effect for determination periods beginning before January 1, 1989 is \$200,000. In determining benefits in

Plan Years beginning on or after January 1, 1996 and before January 1, 2002, the annual compensation limit in effect for determination periods beginning before January 1, 2002 is \$150,000, as adjusted. In determining benefits in Plan Years beginning on or after January 1, 2002, the annual compensation limit in effect for determination periods beginning before that date is \$200,000.

Unless otherwise provided under the Plan, each Code section 401(a)(17) Employee's Accrued Benefit under this Plan will be the greater of the Accrued Benefit determined for the Employee under (a) or (b) below:

(a) the Employee's Accrued Benefit determined with respect to the benefit formula applicable for the Plan Year beginning on or after January 1, 1996, as applied to the Employee's total Years of Service taken into account under the Plan for the purposes of benefit accruals, or

#### (b) the sum of:

- (i) the Employee's Accrued Benefit as of the last day of the last Plan Year beginning before January 1, 1996, frozen in accordance with Treasury Regulation section 1.401(a)(4)-13, and
- (ii) the Employee's Accrued Benefit determined under the benefit formula applicable for the Plan Year beginning on or after January 1, 1996, as applied to the Employee's Years of Service credited to the Employee for Plan Years beginning on or after January 1, 1996, for purposes of benefit accruals.

A Code section 401(a)(17) Employee means a Participant whose current accrued benefit as of a date on or after the first day of the first Plan Year beginning on or after January 1, 1996, is based on Recognized Compensation for a year beginning prior to the first day of the first Plan Year beginning on or after January 1, 1996, that exceeded \$150,000.00.

- 1.1.30 <u>Recognized Employment</u> for all years, that employment, as set forth in the Adoption Agreement, recognized for the purpose of participation in the benefits of this Plan; excluding, however, all
  - (a) Employees included in a unit of Employees covered by an agreement found by the Secretary of Labor to be a collective bargaining agreement between Employee representatives and the Employer, if there is evidence that retirement benefits were the subject of good faith bargaining between such Employee representatives and the Employer and such bargaining agreement or applicable state law does not provide for participation by such Employees in this Plan. For this purpose, the term "Employee representatives" does not include any organization more than half of whose numbers are Employees who are owners, officers or executives of the Employer; and,
  - (b) Employees of the Employer who are nonresident aliens receiving no earned income from the Employer which constitutes income from sources within the United States unless the Adoption Agreement specifically provides that they shall participate in this Plan.

In the event an Employee who is not a member of the eligible class of Employees becomes a member of the eligible class, such Employee will participate immediately if such Employee has satisfied the minimum age and service requirements and would have otherwise previously become a Participant.

1.1.31. <u>Regular Interest</u> - The rate of interest earned each calendar year beginning January 1, 1984 which is equal to the rate of net earnings realized for the calendar year from the investments of the Fund. For Fire Plans, the Retirement Committee shall annually report the amount of regular interest earned for each year.

#### 1.1.32 Regular Pay -

- (a) Police Plans Effective April 16, 2012, tThe average Salary of a Participant for the 5 years preceding the date such Participant elects to retire, the five years preceding his or her death, or the 5 years preceding the date of Disability, whichever is earliest, except that for any Participant who retires, dies or becomes disabled after July 15, 1992, Regular Pay shall mean the average Salary of the Participant for the period of 5 consecutive years preceding such elective retirement, death or date of Disability which produces the highest average.
- (b) <u>Fire Plans</u> The Salary of a Participant at the date such Participant elects to retire or terminate employment with the City.
- (c) Regular Pay shall be limited by Code section 401(a)(17) in the same manner as described in 1.1.29 above.
- 1.1.33 <u>Retirement Committee or Committees</u> The Retirement Committee created under Section 12.3 for general supervision of the Plan and to perform the duties stated in this Agreement and those other duties as are properly delegated in this Agreement, by law or otherwise by the City to the Retirement Committee.

#### 1.1.34 Retirement Date -

- 1) The first of the month immediately following the last day of work in the following cases:
  - a. Normal Retirement Date (defined above)
  - b. Early Retirement Date:
    - i. Police: Attainment of age 55 and completion of 21 years of service.
    - ii. Fire: Attainment of age 50 and completion of 21 years of service with the City, in which case the firefighter will receive the actuarial equivalence of the pension benefit he or she would otherwise receive upon attainment of age 55.
  - c. Disability: Retire or be retired as a result of a Permanent Disability (defined in Section 7.3.3(a))in the line of duty, in which the police officer or firefighter, as the case may be, will receive the applicable pension benefit described herein.
- 2) Fire Plan Special Rule: The first of the month immediately following the attainment of age 55 for a Fire Plan Participant who terminates employment with the City after 21 years of service. Upon attainment of age 55, the firefighter will receive the applicable retirement pension benefit described herein.
- 3) Other. The first of the month immediately following the last day of work after attaining such other age specified in the Adoption Agreement.
- 1.1.35 <u>Retirement Value</u> Subject to any benefit provided in any Prior Plan Statement(s), if any for a Participant, the accumulated value of the Participant's Employee Account and Employer Account. The Retirement Value consists of the sum of the contributions made or transferred to such accounts by the Participant and by the City on the Participant's behalf and the Regular Interest credited to the accounts as of the date of computation, reduced by any realized losses which were not taken into account in determining Regular Interest in any year, and further adjusted each year to reflect the pro rata share for the accounts of the appreciation or depreciation of the fair market value of the assets of the Plan as determined by the Retirement Committee. The Retirement Value shall be reduced by the amount of all distributions made to or on the behalf of the Participant from the Plan, and shall be further reduced each year by the appropriate share of the investment costs of the Fund. Such valuation shall be computed annually as of December 31 and at such other times as the Retirement Committee shall direct on a nondiscriminatory basis. If separate investment accounts are established, a Participant's Retirement Value with respect to such accounts shall be equal to the value of his or her separate investment accounts as determined pursuant to provisions of Nebraska statute as applicable to the Plan.

#### 1.1.36 Salary -

- (a) <u>Police Plans</u> All amounts paid to a Participant by the City for personal services as reported on the Participant's federal income tax withholding statement, including the Participant's contributions picked up by the City as provided in this Agreement and any salary reduction contributions which are excludable from income for federal income tax purposes pursuant to Code sections 125 or 457.
- (b) <u>Fire Plans</u> The base rate of pay, excluding overtime, callback pay, clothing allowances, and other such benefits as reported on the participant's federal income tax withholding statement including the Participants' contributions picked up by the City as provided in this Agreement and any salary reduction contributions which are excludable from income for federal income tax purposes pursuant to Code sections 125 or 457.
- (c) Salary shall be limited by Code section 401(a)(17) in the same manner as described in 1.1.29 above. Effective for Plan and Limitation Years beginning on or after January 1, 1998, any reference in (a) or (b) above shall include deemed section 125 compensation. Deemed section 125 compensation is an amount that is excludable under Code section 106 that is not available to a participant in cash in lieu of group health coverage under a section 125 arrangement solely because the participant is unable to certify that he or she has other health coverage. Amounts are deemed section 125 compensation only if the Employer does not request or collect information regarding the Employee's other health coverage as part of the enrollment process for the health plan.
- 1.1.37 <u>Straight Life Annuity</u> An annuity payable in equal installments in the form of a single life annuity for the life of the Participant (with no ancillary benefits and terminating upon the Participant's death) and to which Employees do not contribute and under which no rollover contributions (as defined in Code sections 402(c), 403(a)(4), 403(b)(8), 408(d)(3), and 457(e)(16)) are taken into account, but taking into account social security supplements and benefits transferred from another defined benefit plan (other than transfers of distributable benefits).
- 1.1.38 <u>Supplemental Effective Date</u> Except as otherwise specified in this Plan or required by applicable law or regulations or other guidance thereunder: (i) the <u>Supplemental Effective Date shall be</u> the date of an amendment by the Employer of this Plan and Trust or a Prior Plan Statement as specified in the Adoption Agreement and (ii) the following general <u>Supplemental Effective Dates shall apply with respect to amended or restated Plan documents for changes in applicable laws, regulations or other guidance —</u>

The Supplemental Effective Date of the restatement of the Plan intended to comply with the Tax Reform Act of 1986 is the first day of the Plan Year commencing after December 31, 1986, except that the effective date of the eligibility conditions in Section 2, the contribution and/or allocation formula in Section 3, and the vesting provisions of Section 5 is the first day of the Plan Year commencing after December 31, 1988; or, as otherwise specifically designated in this Basic Plan. The general Supplemental Effective Date of the Plan intended to comply with the Economic Growth and Tax Relief Reconciliation Act of 2001 ("EGTRRA") shall be the first Plan Year and Limitation Year beginning after December 31, 2001, unless otherwise specified in the this Plan or required by applicable law or regulations or other guidance thereunder. The Supplemental Effective Date of Plan provisions intended to comply with the Pension Protection Act of 2006, Heroes, Earnings Assistance and Relief Tax Act of 2008, and Worker, Retiree, and Employer Recovery Act of 2008 (together the 2006 through 2008 acts referred to herein as "PPA") shall be the first Plan Year and Limitation Year beginning after December 31, 2003, unless otherwise specified in the Plan or required by applicable law, regulations or other guidance. The Supplemental Effective Date of applicable provisions of the Plan for Nebraska Legislative Bill 1082 (2012), if any, shall be April 16, 2012, unless otherwise provided in the Plan or required by applicable law or guidance.

- 1.1.39 <u>Taxable Wage Base ("TWB")</u> The Taxable Wage Base ("TWB") is the maximum annual amount of earnings which may be considered wages for a year under Code section 3121(a)(1), as amended, in effect at the beginning of the Plan Year.
- 1.1.40 <u>Tax Rate</u> The Tax Rate shall be the percentage equal to the portion of rate of tax attributable to Old-Age Insurance under Code section 3111(a), as amended, in effect as of the beginning of the Plan Year.
- 1.1.41 <u>Trustee</u> The Trustee shall be designated in the Adoption Agreement. For Police and Fire Plans, the Trustee of any trust fund shall be a designated funding agent which is qualified to act as a fiduciary in the state of Nebraska, the City treasurer, an appropriate City officer authorized to administer funds of the City, or any combination thereof. The terms of any trust agreement adopted and executed by the City and Trustee separate and apart from this document shall be supplemental and additional to, and incorporated by reference into, this document to the extent not contrary to terms contained herein. If any terms of any such separate trust document conflict with the terms of this document, the terms of this document shall control.
- 1.1.42 <u>Valuation Date</u> The Annual Valuation Date and any other valuation date selected under Section 1.1.35 or Section 4 hereof.
- 1.1.43 <u>Vested</u> Nonforfeitable, i.e., a claim obtained by a Participant or his Beneficiary to that part of an immediate or deferred benefit hereunder which arises from the Participant's service, which is unconditional, and which is legally enforceable against the Plan.
- 1.1.44 <u>Vesting Service</u> A measure of an Employee's service with the Employer (stated as a number of years) which is equal to the number of computation periods in which the Employee is credited with one thousand (1,000) or more Hours of Service; subject, however, to such of the following rules as are applicable under the Adoption Agreement:
  - (a) <u>Computation Periods</u> The computation periods for determining the Employee's Vesting Service shall be Plan Years. Hours of Service will also be credited for any individual considered an Employee for the purposes of this Plan under Code section 414(n).
  - (b) <u>Completion</u> A year of Vesting Service shall be deemed completed as of the date in the computation period that the Employee completes one thousand (1,000) Hours of Service. (Fractional years of Vesting Service shall not be credited.)
  - (c) <u>Pre-Effective Date Service</u> Vesting Service before the Effective Date shall be credited according to whichever of the following rules the Employer indicates in the Adoption Agreement:
    - (i) Vesting Service shall not be credited with respect to periods of employment before the Effective Date of this Plan or a predecessor plan and shall be credited for the period from the Effective Date to the Supplemental Effective Date as if the rules of this Agreement had then been in effect; provided, however, that periods of employment before the ERISA date, if such provisions of ERISA are applicable to this Plan, shall be disregarded in determining Vesting Service if such periods would have been disregarded under the rules of the Plan as in effect immediately before the ERISA date.
    - (ii) Vesting Service shall be credited with respect to all periods before the Effective Date as if the rules of this Agreement had then been in effect; provided, however, that periods of employment before the ERISA Date, if such provisions of ERISA are applicable to this Plan, shall be disregarded in determining Vesting Service if such periods would have been disregarded under the rules of the Plan as in effect

immediately before the ERISA Date. The provisions of this subsection (ii) shall be the default rule and control absent an affirmative election in the Adoption Agreement.

The ERISA Date is the date as of which this Plan was first required to comply with the vesting and eligibility requirements of the Employee Retirement Income Security Act of 1974, if applicable to the Plan, as a condition of tax qualification.

- (d) <u>Police and Fire Plans</u> Participants shall be credited with all years of service in Recognized Employment completed after 1965 (after August 7, 1965 for Fire Plan) for the purpose of determining years of Vesting Service.
- (e) Age 18 If so indicated in the Adoption Agreement, computation periods completed prior to the beginning of the computation period in which the Employee attained the age designated in the Adoption Agreement shall be disregarded in determining the Employee's Vesting Service.
- 1.2 <u>Special Rules for Employer's Computing Service on the Basis of Elapsed Time</u>. For Employers using the elapsed time method of crediting service, the following definitions replace the definitions of Year of Service, Break in Service and Hour of Service definitions.

For purposes of determining an Employee's initial or continued eligibility to participate in the Plan or the nonforfeitable interest in the Participant's Account Balance derived from Employer contributions, (except for periods of service which may be disregarded on account of the "rule of parity") an Employee will receive credit for the aggregate of all time period(s) commencing with the Employee's first day of employment or reemployment (in Recognized Employment with respect to Police and Fire Plans) with the Employer and ending on the date a Break in Service begins. The first day of employment or reemployment is the first day the Employee performs an Hour of Service. An Employee will also receive credit for any period of severance of less than 12 consecutive months. Fractional periods of a year will be expressed in terms of days.

For purposes of this Section, Hour of Service shall mean each hour for which an Employee is paid or entitled to payment for the performance of duties for the Employer.

A Break in Service is a period of severance of at least 12 consecutive months.

Period of severance is a continuous period of time during which the Employee is not employed by the Employer. Such period begins on the date the Employee retires, quits or is discharged, or if earlier, the 12 month anniversary of the date on which the Employee was otherwise first absent from service.

In the case of an individual who is absent from work for maternity or paternity reasons, the 12-consecutive month period beginning on the first anniversary of the first date of such absence shall not constitute a termination of employment. For purposes of this paragraph, an absence from work for maternity or paternity reasons means an absence (1) by reason of the pregnancy of the individual, (2) by reason of the birth of a child of the individual, (3) by reason of the placement of a child with the individual in connection with the adoption of such child by such individual, or (4) for purposes of caring for such child for a period beginning immediately following such birth or placement.

Each Employee will share in Employer contributions for the period beginning on the date the Employee commences participation under the plan and ending on the date on which such Employee severs employment with the Employer or is no longer a member of an eligible class of Employees.

1.3 <u>Gender, Number and Headings.</u> In this Agreement, the pronouns "he," "him" and "his," referring to a Participant or Beneficiary, shall also refer to and include females as well as males, and the singular, when used in this agreement, shall include the plural, and the plural the singular, except when the context otherwise requires. Section, subsection and paragraph headings used herein are not part of this Agreement and shall not be used or considered in determining the proper meaning, purpose or intention hereof.

1.4 <u>Return of Employer Contribution</u>. In the event that the Commissioner of Internal Revenue determines that the Plan is not initially qualified under the Internal Revenue Code, any contribution made incident to that initial qualification by the Employer must be returned to the Employer within one year after the date the initial qualification is denied, but only if the application for the qualification is made by the time prescribed by law for filing the Employer's return for the taxable year in which the Plan is adopted, or such other date as the Secretary of the Treasury may prescribe.

The Trustee, upon written request from the Employer, must return to the Employer the amount of the Employer's contribution made by the Employer by mistake of fact. The Trustee will not return any portion of the Employer's contribution under the provisions of this paragraph more than one year after the Employer made the contribution by mistake of fact.

The Trustee will not increase the amount of the Employer contribution returnable under this Section 1.4 for any earnings attributable to the contribution, but the Trustee will decrease the Employer contribution returnable for any losses attributable to it. The Trustee may require the Employer to furnish it whatever evidence the Trustee deems necessary to enable the Trustee to confirm the amount the Employer has requested be returned is properly returnable under ERISA, if such provisions of ERISA are applicable to the Plan.

- 1.5 <u>Amendment of Prior Plan Statement.</u> If this Plan is adopted as an amendment of an earlier Plan Statement of which the Trustee was not the Trustee, the Employer has caused, or will forthwith cause, the transfer of the existing fund to the Trustee to be held in trust for the purposes contained and set forth in this Agreement.
- 1.6 <u>Supplemental Elections, Information and Plan Provisions</u>. The elections, information and provisions made, provided and specified in the section of the Adoption Agreement corresponding to this Section of the Basic Plan Document shall be deemed a part of an incorporated into this Plan notwithstanding anything in this Basic Plan document or the Adoption Agreement to the contrary.

#### **SECTION 2**

#### Eligibility and Participation

2.1 <u>Initial Entry into Plan</u>. If this Agreement is approved and adopted as an amendment of a Prior Plan Statement, each Employee who immediately before the Supplemental Effective Date was a Participant in the Plan and who on the Supplemental Effective Date continues in Recognized Employment shall continue as a Participant in this Plan.

Each other Employee of the Employer <u>as specified in the Adoption Agreement</u> shall become a Participant on the first date:

- (a) that such person is in Recognized Employment, and as designated by the Employer in the Adoption Agreement,
- (b) that is the first day of the specified period not more than six months after the Employee satisfies age and service requirements specified in the Adoption Agreement,
- (c) that is also not earlier than the first (1st) day of the first (1st) month or the first (1st) day of the seventh (7th) month of the Plan Year, whichever first occurs after the Employee has both satisfied the age and service requirements set forth in the Adoption Agreement,
- (c)—that is the first day of the Plan Year in which the Employee has satisfied the age and service requirements specified in the Adoption Agreement, or
- (d) that is the first (1st) day of the first (1st) month or the first (1st) day of the seventh (7th) month of the Plan Year, whichever first occurs after the Employee has both satisfied the age and service requirements set forth in the Adoption Agreement, or

- (e) as otherwise specified in the Adoption Agreement that is the first day of the Plan Year following the Plan Year in which the Employee has satisfied the age and service requirements specified in the Adoption Agreement.
- 2.2 Special Rules for Former Participants and Eligible Classes. A Participant whose employment with the Employer terminates and who subsequently is reemployed by the Employer shall immediately reenter the Plan as a Participant upon his return to Recognized Employment. In the event a Participant is no longer a member of an eligible class of Employees and becomes ineligible to participate but has not incurred a Break in Service, such Employee will participate immediately upon returning to an eligible class of Employees. Except for Police and Fire Plans, iH such a Participant incurs a Break in Service, eligibility will be determined under the Break in Service rules of the Plan. For Police and Fire Plans, a Participant immediately will participate upon returning to Recognized Employment of the Police or Fire Plan.

In the event an Employee who is not a member of an eligible class of Employees becomes a member of an eligible class, such Employee will participate immediately if such Employee has satisfied the minimum age and service requirements and would have otherwise previously become a Participant.

2.3 <u>Annual Certification</u>. As of each Annual Valuation Date during the continuance of the Plan, the Employer shall certify in writing the names of all Participants who are entitled to participate in the Employer contribution for the Plan Year ending on that date and all other facts that may be required to enable the Trustee to perform its duties hereunder.

#### **SECTION 3**

#### Contributions and Allocation Thereof

#### 3.1 Employee Contributions.

- (a) Mandatory Employee Contributions. Each Participant shall, as a condition of employment with the City, contribute to the Plan the amount stated in the Adoption Agreement. Such payment shall be made by regular payroll deduction from periodic salary and shall be credited to Employee Accounts at least monthly. Employee Accounts shall also be credited with Regular Interest as provided in Section 4.1.
  - (i) If so elected by the City in the Adoption Agreement, the City shall pick up Mandatory Employee contributions, paying such contributions in lieu of contributions by Participants, and the contributions so picked up shall be treated as Employer contributions under Code section 414(h)(2) for federal income tax purposes. Participants shall not have the option to receive the contributed amounts directly instead of having them paid by the City to the Plan. The City shall withhold federal income taxes on picked up Employee contributions until if required by the Internal Revenue Service Code or other applicable laws, regulations or guidance or the federal courts determine that picked up contributions shall not be included as gross income of the Employee prior to being distributed from the Plan.
  - (ii) Picked up contributions shall be paid from the same source of funds as used to pay earnings to Employees. The City shall pick up Employee contributions by salary deduction either by reducing cash salary of an Employee or by reducing cash salary and offsetting future salary increases.
- (b) Voluntary Employee Contributions. If elected in the Adoption Agreement, each Employee shall be entitled to make voluntary cash contributions to the Plan to the extent and maximum amount permissible under, and subject to applicable requirements and limitations of, the Internal Revenue Code. Voluntary Contributions shall be credited to and separately accounted for under Employee Accounts and shall thereafter be credited with Regular Interest as provided in Section 4.1. Not in limitation of the foregoing, Voluntary Employee Contributions shall satisfy any special nondiscrimination or other

applicable requirements of the Code, rules, regulations and guidance, including, but not limited to, Code sections 401(a)(4) and 401(m). Any applicable requirements of Code section 401(m) shall be satisfied in accordance with Code section 401(m)-2 and Treasury Regulations section 1.401(m)-2 (which provisions are incorporated herein by this reference), using the prior year testing method and deeming the average contribution percentage for eligible nonhighly compensated employees to be 3% for the first Plan Year, unless otherwise elected by the Employer in a written addendum to the Adoption Agreement or Plan amendment, or otherwise required by applicable rules, regulations or other guidance. Provided, however, that certain provisions of the Code and corresponding rules, regulations and other guidance shall not apply if this Plan is a governmental plan as defined in Code section 414(d), including provisions that may otherwise be applicable to Voluntary Employee Contributions. (Inapplicable provisions include Code sections 401(a)(3), (4), (7), (11) through (15), (19) and (20), 401(m), 410(b), 411, 412 and 417 pursuant to-For express exclusion of governmental plans from these requirements, see operative provisions of these Code sections, as well as Code section 401(a)(5)(G) and the flush language following Code section 401(a)(376).)

- (c) Pre-1984 Contributions. <u>Effective April 16, 2012 w</u>With respect to <u>Police and Fire plans</u> only, an Employee's contributions to a qualified plan of deferred compensation maintained by the City prior to January 1, 1984 shall be transferred to his or her Employee Account without interest unless the City, at the time of the Transfer, credited interest on such contributions.
- (d) All contributions in this Section 3.1 shall be credited to the Employee Accounts of Participants.

# 3.2 <u>Employer Contributions</u>.

- (a) <u>Police and Fire Plans</u>. Employer contributions shall be made pursuant to (i), (ii) and (iii) below for Police and Fire Plans:
  - (i) The City shall contribute to the Trustee during the continuance of the Plan the amount stated in the Adoption Agreement. Employer contributions shall be credited at least monthly to Employer Accounts.
  - (ii) The City also shall contribute to the Employer Account of each Employee who was employed by the City on January 1, 1984 an amount equal to the Employee's contributions prior to January 1, 1984 to a qualified plan of deferred compensation maintained by the City, without interest unless the City elects to pay interest thereon. The contribution shall be made at the time the Employee retires or terminates his or her employment with the City. The City may make this contribution prior to an Employee's retirement or termination of employment; however, the City must make such contributions in a manner which does not impermissibly discriminate in favor of any Highly Compensated Employee.
  - (iii) The City shall make additional contributions to the Plan in amounts and at such times as are necessary to provide the minimum benefits stated in this Agreement. For Fire Plans, such additional Employer contributions shall be paid to an Unallocated Employer Account, and in all cases shall not exceed the applicable limitations of Code section 415.

#### (b) Other Municipal Plans.

(i) <u>Nonintegrated</u>. The City shall make contributions from year to year during the continuance of the Plan to the Trustee in the amount specified in the Adoption Agreement. The appropriate contribution of the Employer to the Plan, determined as herein provided, may be paid by the Employer either in cash or in kind and shall be allocated, as of the Annual Valuation Date in the Plan Year in question, to the Employer Account of Participants entitled to share therein in the same proportion that each Participant's Recognized Compensation bears to the aggregate Recognized Compensation of all Participants for that Fiscal Year.

- (ii) <u>Integrated</u>. For Plan Years beginning after December 31, 1988 and before August 5, 1997, if the Plan is integrated and is not Top-Heavy for the Plan Year:
  - (1) Employer contributions and forfeitures will be allocated to each Participant's Employer Account in the ratio that the sum of each Participant's Recognized Compensation and Recognized Compensation in excess of the integration level bears to the sum of all Participants' Recognized Compensation and Recognized Compensation in excess of the integration level, but not in excess of the profit-sharing maximum disparity rate.
  - (2) Any remaining Employer contributions and forfeitures will be allocated to each Participant's Employer Account in the ratio that each Participant's Recognized Compensation for the Plan Year bears to all Participants' Recognized Compensation for that Plan Year.

If the Plan is not Top-Heavy, the maximum profit sharing disparity rate is equal to the lesser of:

- (A) the greater of 5.7% or the Tax Rate; or,
- (B) the applicable percentage determined in accordance with the following table:

## If the integration level

is more than	but not more than	the applicable percentage is:
\$0	X*	5.7%
X* of TWB	80% of TWB	4.3%
80% of TWB	Y**	5.4%

<sup>\*</sup>X =the greater of \$10,000 or twenty percent (20%) of the TWB

If the integration level is equal to the taxable wage base, the applicable percentage is the greater of 5.7% or the Tax Rate.

#### (3) Overall permitted disparity limit:

- (A) Annual overall permitted disparity limit: Notwithstanding the preceding paragraph, for any Plan Year this Plan benefits any Participant who benefits under another qualified plan or simplified employee pension, as defined in Code section 408(k), maintained by the employer that provides for permitted disparity (or imputes disparity), the Employer will contribute for each Participant who either completes more than 500 Hours of Service during the Plan Year or is employed on the last day of the Plan Year an amount equal to the excess contribution percentage multiplied by the Participant's total compensation.
- (B) Cumulative permitted disparity limit: Effective for Plan Years beginning on or after January 1, 1995, the cumulative permitted disparity limit for a

<sup>\*\*</sup>Y = any amount more than eighty percent (80%) of the TWB but less than 100% of the TWB.

Participant is thirty-five (35) total cumulative permitted disparity years. Total cumulative permitted years means the number of years credited to the Participant for allocation or accrual purposes under this Plan, any other qualified plan or simplified employee pension plan (whether or not terminated) ever maintained by the Employer. For purposes of determining the Participant's cumulative permitted disparity limit, all years ending in the same calendar year are treated as the same year. If the Participant has not benefited under a defined benefit or target benefit plan for any year beginning on or after January 1, 1994, the Participant has no cumulative disparity limit.

- (c) <u>Mistake</u>. If, after the Employer's contribution has been made and allocated, it should appear that, through oversight or a mistake of fact or law, a Participant who should have been entitled to share in such contribution received no allocation or received an allocation which was less than he should have received, the Employer may, at its election, and in lieu of reallocating such contribution, make a special make-up contribution to provide for him the same percentage of his Recognized Compensation for such Plan Year as was allocated to the Employer Accounts of other Participants for such Plan Year. A special make-up contribution shall not be permitted, and the Employer shall reallocate such contribution, if a special make-up contribution would result in a Participant receiving a greater allocation than is permitted under any law.
- (d) <u>Return of Contribution</u>. The corpus or income of the Plan may not be diverted to or used for any other purpose than the exclusive benefit of the Participants or their Beneficiaries. Provided, however, that any contribution made by the Employer because of a mistake of fact may be returned to the Employer within one (1) year of the contribution.

If the Retirement Value of a Participant's Account is not sufficient to purchase or provide a Required Minimum Benefit required by Section 7.3, the City shall utilize such Funds as may be necessary from the Unallocated Employer Account (or from the Employer Account of the Participant in the case of a Police Plan) to purchase or provide the benefits required by such Section.

#### 3.3 Eligible Participants.

(a) <u>Police and Fire Plans</u>. The Employer contributions (with the exception of contributions to provide a Participant a Required Minimum Benefit) shall be allocated to the accounts of all Participant Employees of the City paid compensation during the Plan Year.

#### (b) Other Municipal Plans.

- (i) Unless otherwise specified in the Adoption Agreement, the Employer contribution and forfeitures available for allocation for the Plan Year shall be allocated to the accounts of all Participants who had at least one thousand (1,000) Hours of Service during such Plan Year and were Employees of the Employer on the last day of such Plan Year, and to the accounts of Participants who terminated their employment with the Employer within such Plan Year by reason of (i) their death, or (ii) their retirement at or after their Normal Retirement Date, or (iii) their Disability.
- (ii) Except if the Plan provides for "picked up" contributions: iIf the Employer has so indicated in the Adoption Agreement, an Employee eligible to participate, or any present Participant, may elect not to participate in the Plan. For an election to be effective for a particular Plan Year, the Employee or Participant must file the election in writing with the Retirement Committee or City not later than 60 days prior to the last day of that Plan Year. The Employer may not make a contribution under the Plan for the Employee or for the Participant for the Plan Year for which the election is effective, nor for any succeeding Plan Year unless the Employee or Participant re-elects to participate in the Plan. After an Employee's or Participant may re-elect to participate in the Plan for any Plan Year and subsequent Plan Years. An Employee or

Participant may re-elect to participate in the Plan by filing his election in writing with the Retirement Committee not later than 60 days prior to the last day of the Plan Year for which his election is to be effective. An Employee or Participant who re-elects to participate may not again elect not to participate. An election timely filed is effective for the entire Plan Year.

A Participant who elects not to participate may not receive a distribution of his Total Account (except his Prior Voluntary Non-Deductible Account) attributable either to Employer or to Participant contributions except as provided under Section 6 or 7. However, for each Plan Year for which a Participant's election not to participate is effective, the Participant's Account, if any, continues to share in Fund earnings under Section 4. Furthermore, the Employee or the Participant receives vesting credit under Section 1.1.44 for each included Year of Service during the period the election not to participate is effective.

- (c) For any Plan Year, if application of the limitations or requirements of subsection (b) would cause the Plan to fail to satisfy the requirements of Sections 401(a)(26) and/or 410(b) of the Code, if applicable to the Plan (Section 410(b) being inapplicable if this is a government plan as defined in Code section 414(d)), such subsections shall not apply and subsection (d) shall apply. Solely for the Plan Year commencing in 1989, a Participant shall be treated as benefiting under the Plan if the sole reason for the Participant's failure to receive an allocation to his account is due to the requirement that a Participant complete at least 1,000 Hours of Service during the Plan Year commencing in 1989.
- (d) For a plan with respect to which paragraph (c) applies, the Employer contribution and Forfeitures available for allocation for the Plan Year shall be allocated to the accounts of all Participants; provided, however, in the event that the Employer has so indicated in the Adoption Agreement, if the Employee terminated employment during the Plan Year with not more than 500 Hours of Service, such terminated Employee shall not be allocated any portion of the Employer contribution or Forfeitures allocated for such Plan Year. If this subsection (d) applies, subsection (b) shall not apply.
- 3.4 <u>Forfeitures</u>. Except as otherwise provided in Section 6.4, <u>and</u> to the extent elected or otherwise provided for in the Adoption Agreement, Forfeitures shall be added to or be used to reduce the Employer's contribution. Forfeitures available for allocation in addition to or as a part of the Employer's contribution for the Fiscal Year shall be allocated as Employer contributions as provided in Section 3.2.

# 3.5 Rollover Contributions.

## (a) <u>Police and Fire Plans</u>.

- (i) <u>Police</u>. If a police officer terminates his or her employment with another city of the first class in Nebraska for the purpose of becoming a police officer employed by the City, and commences such employment with the City within one hundred and twenty days of such termination, such police officer shall be entitled to transfer to the Trustee of the Plan the full amount of his or her Employee Account and the vested portion of his or her Employer Account in the Police Officers Retirement System Fund at the time of termination so long as the transfer is made in accordance with the requirements of Section 3.5(b) below. The funds shall be transferred directly to the Employee Account of the Employee and administered thereafter by the Retirement Committee. The Employee shall otherwise be deemed a new Employee for purposes of the Plan after he or she commences employment with the City.
- (ii) <u>Fire</u>. If a firefighter terminates his or her employment after four (4) or more years as a firefighter with another city of the first class in Nebraska, for purposes of becoming a firefighter with the City, and commences employment with the City as a firefighter within ninety (90) days after such termination, the firefighter shall be entitled to transfer to the Trustee of the Plan the full amount of his or her Employee Account and the vested portion of his or her Employer Account in the Firefighters Retirement System Fund at the time of termination so long as the transfer is made in accordance with the requirements of Section 3.5(b) below. The transferred funds shall be administered by the Retirement Committee of the Plan. The Employee

shall otherwise be deemed a new Employee for purposes of the Plan after he or she commences employment with the City.

- (b) Other Municipal Plans. Notwithstanding the limitations of Section 3.6, a Participant may, if elected by the Employer in the Adoption Agreement, make a rollover contribution or direct rollover (in addition to any Employer contribution described in Section 3.2, and any Participant contribution described in Section 3.1):
  - (i) from a qualified plan described in Code section 401(a) or 403(a), which includes a trust exempt from federal income tax, and excludes after-tax Employee contributions; or
  - (ii) from a conduit individual retirement account or a conduit individual retirement annuity from a qualified plan; or
  - (iii) such other types of plans, effective on such dates, as specified in the Adoption Agreement.

Furthermore, an eligible Employee, prior to satisfying the Plan's eligibility condition(s), may make a rollover contribution or direct rollover to the same extent and in the same manner as a Participant. If an Employee makes a rollover contribution or direct rollover to the Trust prior to satisfying the Plan's eligibility condition(s), the Retirement Committee and Trustee shall treat the Employee as a Participant for all purposes of the Plan except that the Employee shall not share in Employee or Employer contributions or Participant forfeitures under Section 3 until he actually becomes a Participant in the Plan. If the Employee terminates employment prior to becoming a Participant, the Trustee shall distribute his rollover contribution Account to him as if it were an Employer contribution Account.

All rollover contributions and direct rollovers to the Plan shall be <u>fully vested and made and</u> allocated to the Rollover Account of such Participant or Employee <u>in accordance with applicable law, including separate accounting for amounts includible and not includible in gross income.</u>

- (c) Merger of Services. To the extent Neb. Rev. Stat. Section 13-2401 is applicable to this Plan: If the Employer transfers employees who are Participants to a separate political subdivision (as defined by Neb. Rev. Stat. Section 13-2401(a)(1)), or receives employees from another political subdivision, in a merger of services, such employees who became employees of the new political subdivision, or the Employer, shall receive credit for their years of participation in the retirement system of the transferring entity for purposes of membership in the retirement system of the receiving entity. Rollovers, contributions and transfers from and between plans in such cases shall be carried out in accordance with Neb. Rev. Stat. Section 13-2401, to the extent consistent with applicable provisions of the Code.
- 36 Limitation on Allocations and Benefits. Notwithstanding anything in this Plan to the contrary, contributions and benefits under this Plan shall not exceed applicable limitations of Code Section 415, as set forth therein and the Treasury Regulations and other applicable guidance thereunder, the provisions of which are incorporated herein by this reference. More specifically, annual additions to this Plan shall not exceed the Maximum Permissible Amount, and any annual benefit from this Plan that is treated as a distribution from a defined benefit plan shall not exceed the Maximum Permissible Benefit. All such annual additions and annual benefits hereunder shall be automatically limited to a level necessary to prevent the limitations of Code section 415, as adjusted from time to time pursuant to Code section 415(d), from being exceeded with respect to any Participant. If a benefit or contribution during a Limitation Year would produce an annual benefit or annual addition in excess of the Maximum Permissible Benefit or Amount, the benefit or amount will be limited to the extent necessary so that the a-benefit or amount that-does not exceed the Maximum Permissible Benefit or Amount. Provided, however, that, for purposes of the Maximum Permissible Benefit, the compensation limitation of subsection 415(b)(1)(B) shall not apply if this Plan constitutes a "governmental plan" within the meaning of Code section 414(d), pursuant to Code section 415(b)(11). Furthermore, special rules cross referenced in Treasury Regulations section 1.415(a)-1(b)(3) shall apply to the extent this Plan is a governmental plan described in said rules.

- 3.6.1 Single Plan. If the Participant does not participate in, and has never participated in another qualified plan maintained by the Employer or a welfare benefit fund, as defined in Code section 419(e) maintained by the Employer; or an individual medical account, as defined in Code section 415(1)(2), maintained by the Employer; or a simplified employee pension as defined by Code section 408(k), maintained by the Employer, which provides an annual addition as defined in Section 3.6.9, the amount of annual additions which may be credited to the Participant's Total Account for any Limitation Year will not exceed the lesser of the maximum permissible amount or any other limitation contained in this Plan. If the Employer contribution that would otherwise be contributed or allocated to the Participant's Employer Account would cause the annual additions for the Limitation Year to exceed the maximum permissible amount, the amount contributed or allocated will be reduced so that the annual additions for the Limitation Year will equal the maximum permissible amount. In addition to, and not in limitation of, the foregoing, distributions from the Plan (or from an annuity contract making distributions on behalf of the Plan or from an annuity contract that has been distributed under the Plan) during a limitation year that are or are treated as payments from a defined benefit plan shall not exceed the limitations of Code Section 415 and Regulations thereunder (as such limitations shall be adjusted from time to time pursuant to Code section 415(d)), the terms of which are incorporated herein by this reference. Benefits payable under this Plan shall be automatically limited as necessary to prevent the limitations of Code Section 415 from being exceeded with respect to any Participant. The foregoing shall operate automatically without discretion of the Employer.
- 3.6.2 <u>Determination of Maximum Permissible Amount</u>. Prior to determining the Participant's actual compensation for the Limitation Year, the Employer may determine the maximum permissible amount for a Participant on the basis of a reasonable estimation of the Participant's compensation for the Limitation Year, uniformly determined for all Participants similarly situated. As soon as is administratively feasible after the end of the Limitation Year, the maximum permissible amount for the Limitation Year will be determined on the basis of the Participant's actual compensation for the Limitation Year.

In the case of an individual who was a participant in one or more defined benefit plans of the Employer as of the first day of the first Limitation Year beginning after December 31, 1994, the application of the limitations of this Article shall not cause the Maximum Permissible Amount for such individual under all such defined benefit plans to be less than the individual's Retirement Protection Act of 1994 (RPA '94) Old Law Benefit. The preceding sentence applies only if such defined benefit plans met the requirement of Code section 415 on December 7, 1994.

3.6.3 <u>Disposition Correction of Excess Amount</u>. If, pursuant to Section 3.6.1 or as a result of the allocation of Forfeitures, there is an excess amount, the excess will be disposed of in the time and manner permitted under IRC Section 415 or other applicable laws, rules, regulations or guidance, as adopted or amended from time to time, including, but not limited to, Preambles to Final Regulations under Code Section 415, 72 FR 65, 16888 (April 5, 2007); Rev. Proc. 2008-50-50 (Revenue Procedure 2013-12 effective April 1, 2013)2006-27, Section 2.02(2) & Appendices B and C. To the extent permitted, the following correction methodology and order will be used:

#### (a) <u>Police and Fire Plan</u>.

- (i) Any Nondeductible Voluntary Employee contributions (plus attributable earnings), to the extent that they would reduce the excess amount, will be returned to the Participant;
- (ii) If after the application of Subparagraph (i) an excess amount still exists, and the Participant is covered by the Plan at the end of a Limitation Year, the excess amount in the Participant's Employer Account will be used to pay Plan administration expenses incurred by the City;
- $(iii) \qquad \text{If after the application of Subparagraphs (i) and (ii) an excess amount still exists, and the Participant is covered by the Plan at the end of a Limitation Year, the}\\$

excess amount in the Participant's Employer Account will be used to reduce Employer Contributions (including any allocation of Forfeitures) for such Participant in the next Limitation year, and each succeeding Limitation Year if necessary.

(iv) If after the application of Subparagraphs (i) and (ii), an excess amount still exists and the Participant is not covered by the Plan at the end of a Limitation Year, the excess amount will be held unallocated in a suspense account. The suspense account will be applied to reduce further Employer Contributions (including allocation of any Forfeitures) for all remaining Participants in the next Limitation Year, and each succeeding Limitation Year if necessary.

#### (b) Other Municipal Plans.

- (i) Any Nondeductible Voluntary Employee Contributions (plus attributable earnings), to the extent that they would reduce the excess amount, will be returned to the Participant;
- (ii) If after the application of Subparagraph (i) an excess amount still exists, and the Participant is covered by the Plan at the end of a Limitation Year, the excess amount in the Participant's Employer Account will be used to reduce Employer Contributions (including any allocation of Forfeitures) for such Participant in the next Limitation Year, and each succeeding Limitation Year if necessary.
- (iii) If after the application of Subparagraph (i) an excess amount still exists, and the Participant is not covered by the Plan at the end of a Limitation Year, the excess amount will be held unallocated in a suspense account. The suspense account will be applied to reduce further Employer Contributions (including allocation of any Forfeitures) for all remaining Participants in the next Limitation Year, and each succeeding Limitation Year if necessary;
- (c) If a suspense account is in existence at any time during a Limitation Year pursuant to this Section, it will not participate in the allocation of the Fund's investment gains and losses. If a suspense account is in existence at any time during a particular Limitation Year, all amounts in the suspense account must be allocated and reallocated to Participants' accounts before any Employer Contributions or any Employee Contributions may be made to the Plan for that Limitation Year. Excess amounts may not be distributed to Participants or former Participants.
- (d) Notwithstanding any provision of the Plan to the contrary, effective for Limitation Years beginning on or after July 1, 2007, if the annual additions or benefits (within the meaning of Code § 415) are exceeded for any Participant, then the Plan may only correct such excess in accordance with the Employee Plans Compliance Resolution System (EPCRS) as set forth in Revenue Procedure 2006-272008-50-50 (Revenue Procedure 2013-12 effective April 1, 2013) or any amended or superseding guidance, including, but not limited to, the preamble of the final Code §415 regulations.
- 3.6.4 <u>Master or Prototype Plans</u>. This Section applies if the Participant <u>also</u> is covered under a qualified master or prototype defined contribution plan maintained by the Employer, or a welfare benefit fund, as defined in Code section 419(e), maintained by the Employer; an individual medical account, as defined in Code section 415(1)(2), maintained by the Employer; or a simplified employer pension, maintained by the Employer, which provides an annual addition as defined in Section 3.6.9, during any Limitation Year. The annual additions which may be credited to a Participant's Total Account under this Plan for any such Limitation Year will not exceed the maximum permissible amount reduced by the annual additions credited to a Participant's Total Account under the other defined contribution plans, welfare benefit funds, individual medical accounts, and simplified employer pensions for the same Limitation Year. If the annual additions with respect to the Participant under other defined contribution plans, welfare benefit funds, individual medical accounts, and simplified employer pensions maintained by the Employer

are less than the maximum permissible amount and the Employer contribution that would otherwise be contributed or allocated to the Participant's Total Account under this Plan would cause the annual additions for the Limitation Year to exceed this limitation, the amount contributed or allocated will be reduced so that the annual additions under all such plans for the Limitation Year will equal the maximum permissible amount. If the annual additions with respect to the Participant under such other defined contribution plans, welfare benefit funds, individual medical accounts, and simplified employer pensions in the aggregate are equal to or greater than the maximum permissible amount, no amount will be contributed or allocated to the Participant's Total Account under this Plan for the Limitation Year.

Prior to determining the Participant's actual compensation for the Limitation Year, the Employer may determine the maximum permissible amount for a Participant in the manner described in Section 3.6.2. As soon as is administratively feasible after the end of the Limitation Year, the maximum permissible amount for the Limitation Year will be determined on the basis of the Participant's actual compensation for the Limitation Year.

If, pursuant to the preceding paragraph or as a result of the allocation of forfeitures, a Participant's annual additions under this Plan and such other plans would result in an excess amount for a Limitation Year, the excess amount will be deemed to consist of the annual additions last allocated, except that annual additions attributable to a welfare benefit fund or individual medical account will be deemed to have been allocated first regardless of the actual allocation date.

- 3.6.5 <u>Amount Attributable to this Plan</u>. If an excess amount was allocated to a Participant on an Annual Valuation Date of this Plan which coincides with an Annual Valuation Date of another plan, the excess amount attributed to this Plan will be the product of,
  - (a) the total excess amount allocated as of such date, multiplied by
  - (b) the ratio of (i) the annual additions allocated to the Participant for the Limitation Year as of such date under this Plan to (ii) the total annual additions allocated to the Participant for the Limitation Year as of such date under this and all the other qualified master or prototype defined contribution plans.
- 3.6.6 <u>Correction of Excess Amount or Benefit</u>. Any excess amount or benefit attributed to this Plan will be disposed or otherwise corrected in the manner described in Section 3.6.3. <u>Provided, however, effective for Limitation Years beginning on or after July 1, 2007 and notwithstanding any provision of the Plan to the contrary, if the annual additions or benefits (within the meaning of Code § 415) are exceeded for any Participant, then the Plan may only correct such excess in accordance with the Employee Plans Compliance Resolution System (EPCRS) as set forth in Revenue Procedure 2006 27 or any superseding guidance, including, but not limited to, the preamble of the final Code §415 regulations.</u>
- 3.6.7 Other Defined Contribution Plans. If the Participant is covered under another qualified defined contribution plan maintained by the Employer which is not a master or prototype plan, annual additions which may be credited to the Participant's Total Account under this Plan for any Limitation Year will be limited in accordance with Sections 3.6.4 through 3.6.6 as though the other plan were a master or prototype plan unless the Employer provides other limitations in Section K of the Adoption Agreement.
- 3.6.8 Other Defined Benefit Plans. If distributions hereunder are treated as payments from a defined benefit plan and the Participant is or has ever been a participant in another qualified defined benefit plan (regardless whether or not terminated) maintained by the Employer or a predecessor employer, the sum of the payments from all such plans shall not exceed applicable limitations of Code section 415(b), as adjusted from time to time pursuant to Code section 415(d). Where the Participant's employer-provided benefits under all such defined benefit plans would exceed said applicable limitations, payments under this Plan shall be limited so that the limitations under Code section 415(b) are not exceeded, unless otherwise provided in Section K of the Adoption Agreement.
  - 3.6.9 <u>Definitions and Special Rules</u>.

- (a) <u>Annual Additions</u> The sum of the following amounts credited to a Participant's Total Account for the Limitation Year:
  - (i) Employer contributions;
  - (ii) Employee contributions;
  - (iii) Forfeitures;
  - (iv) Amounts allocated, after March 31, 1984, to an individual medical account, as defined in Code section 415(1)(2), which is part of a pension or annuity plan maintained by the Employer are treated as annual additions to a defined contribution plan. Also amounts derived from contributions paid or accrued after December 31, 1985, in taxable years ending after such date, which are attributable to post-retirement medical benefits, allocated to the separate account of a key employee, as defined in Code section 419A(d)(3), under a welfare benefit fund, as defined in Code section 419(e), maintained by the Employer are treated as annual additions to a defined contribution plan; and.
    - (v) allocations under a simplified employer pension.

For this purpose, any excess amount applied under Sections 3.6.3 or 3.6.6 in the Limitation Year to reduce Employer Contributions will be considered annual additions for such Limitation Year.

- 3.6.9(a)-1 <u>Final Section 415 Regulations Annual Additions.</u> The Plan's definition of "annual additions" is modified as follows for Limitation Years beginning on or after July 1, 2007:
  - Restorative payments. Annual additions for purposes of Code § 415 shall not include restorative payments. A restorative payment is a payment made to restore losses to a Plan resulting from actions by a fiduciary for which there is reasonable risk of liability for breach of a fiduciary duty under ERISA or under other applicable federal or state law, where participants who are similarly situated are treated similarly with respect to the payments. Generally, payments are restorative payments only if the payments are made in order to restore some or all of the plan's losses due to an action (or a failure to act) that creates a reasonable risk of liability for such a breach of fiduciary duty (other than a breach of fiduciary duty arising from failure to remit contributions to the Plan). This includes payments to a plan made pursuant to a Department of Labor order, the Department of Labor's Voluntary Fiduciary Correction Program, or a court-approved settlement, to restore losses to a qualified defined contribution plan on account of the breach of fiduciary duty (other than a breach of fiduciary duty arising from failure to remit contributions to the Plan). Payments made to the Plan to make up for losses due merely to market fluctuations and other payments that are not made on account of a reasonable risk of liability for breach of a fiduciary duty under ERISA or other applicable federal or state law are not restorative payments and generally constitute contributions that are considered annual additions.
  - b. Other Amounts. Annual additions for purposes of Code  $\S$  415 shall not include: (1) The direct transfer of a benefit or employee contributions from a qualified plan to this Plan; (2) Rollover contributions (as described in Code  $\S$  401(a)(31), 402(c)(1), 403(a)(4), 403(b)(8), 408(d)(3), and 457(e)(16)); (3) Repayments of loans made to a participant from the Plan; and (4) Repayments of amounts described in Code  $\S$  411(a)(7)(B) (in accordance with Code  $\S$  411(a)(7)(C)) and Code  $\S$  411(a)(3)(D) or repayment of contributions to a governmental plan (as defined in Code  $\S$  414(d)) as described in Code  $\S$  415(k)(3), as well as Employer restorations of benefits that are required pursuant to such repayments.

c. Date of tax-exempt Employer contributions. Notwithstanding anything in the Plan to the contrary, in the case of an Employer that is exempt from Federal income tax (including a governmental employer), Employer contributions are treated as credited to a participant's account for a particular limitation year only if the contributions are actually made to the plan no later than the 15th day of the tenth calendar month following the end of the calendar year or fiscal year (as applicable, depending on the basis on which the employer keeps its books) with or within which the particular limitation year ends.

#### (b) Compensation -

- (i) 415 Safe Harbor Compensation. For the purpose of this Section 3.6, a Participant's earned income, wages, <u>differential wage payments under Code section 3401(h)</u> (for years beginning after December 31, 2008), salaries, and fees for professional services and other amounts received (without regard to whether or not an amount is paid in cash) for personal services actually rendered in the course of employment with the Employer who maintains the Plan to the extent that the amounts are includable in gross income (including, but not limited to, commissions paid salespersons, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits, and reimbursements or other expense allowances under a nonaccountable plan (as described in Treasury Regulations, Section 1.62-2(c)) and excluding the following:
- a. Employer contributions (other than elective contributions described in section 402(e)(3), 408(k)(6), 408(p)(2)(A)(i) or 457(b)) made by the Employer to a plan of deferred compensation (including a simplified employee pension or simple retirement account, described in Code section 408(k) or (p), respectively, and whether or not qualified) to the extent not includable in the Employee's gross income for the taxable year in which contributed., and any distribution from a plan of deferred compensation (whether or not qualified and regardless whether such amounts are includible in the gross income of the Employee when distributed). However, if the Employer so provides elsewhere in this Plan, any amounts received by an Employee pursuant to a nonqualified unfunded deferred compensation plan are permitted to be considered as compensation for section 415 purposes in the year the amounts are actually received, but only to the extent such amounts are includible in the Employee's gross income;
- b. Amounts realized from the exercise of a nonstatutory option, or when restricted stock (or other property) held by the Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture (see Code section 83 and related regulations);
- c. Amounts realized from the sale, exchange or other disposition of stock acquired under a statutory stock option;
- d. Other amounts which received special tax benefits, such as premiums for group-term life insurance (but only to the extent that the premiums are not includible in the gross income of the Employee and are not salary reduction amounts that are described in section 125); and
- e. Other items of remuneration that are similar to any of the items listed in (a) through (d) above.
- 415 Safe Harbor Compensation as contained in this paragraph 3.6.9(b)-i shall be the definition of compensation for purposes of satisfying the requirements of Code

section 415 unless an alternative definition of compensation described below is elected in the Adoption Agreement.

- (ii) Alternative Definitions of Compensation. The following alternative definition of compensation shall apply for Code Section 415 purposes if elected by the Employer in the Adoption Agreement:
  - a. W-2 Wages. Compensation is defined as wages within the meaning of Code section 3401(a) and all other payments of compensation to an Employee by the Employer (in the course of the Employer's trade or business) for which the Employer is required to furnish the Employee a written statement under Code sections 6041(d), 6051(a)(3), and 6052. Compensation shall be determined without regard to any rules under Code section 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in section 3401(a)(2)); or
  - b. Section 3401(a) Wages. Compensation is defined as wages within the meaning of Code section 3401(a) for the purposes of income tax withholding at the source but determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in section 3401(a)(2)).
  - c. Such other definition of compensation for Code section 415 purposes as specified in the Adoption Agreement.
- (iii) For purposes of applying the limitations of this Section, compensation for a Limitation Year is the compensation actually paid or made available during such year.
- (iv) For limitation years beginning before July 1, 2007, compensation for a Participant in a defined contribution plan who is permanently and totally disabled (as defined in Code section 22(e)(3)) is the compensation such Participant would have received for the Limitation Year if the Participant had been paid at the rate of compensation paid immediately before becoming permanently and totally disabled; for Limitation Years beginning before January 1, 1997, but not for Limitation Years beginning after December 31, 1996, such imputed compensation for the disabled Participant may be taken into account only if the Participant is not a Highly Compensated Employee (as defined in Code section 414(q)), and contributions made on behalf of such Participant are nonforfeitable when made. See section 3.6.9(b)-3(d) below for limitation years beginning on or after July 1, 2007.
- (v) Notwithstanding anything in this section to the contrary, effective for Limitation Years beginning on or after January 1, 1998, compensation for purposes of this Section 3.6 shall include any amount that would be included in compensation but for an election under Code section 125(a), 402(e)(3), 402(h)(1)(B), 402(k) or 457(b); or, effective January 1, 2001, Code section 132(f)(4).
- (vi) Unless elected by the Employer in the Adoption Agreement, amounts under Code Section 125 will not include any amounts that are not available to an Employee in cash in lieu of group health coverage because the Employee is unable to certify that he or she has other health coverage (deemed Section 125 compensation). If an election is made in the Adoption Agreement, for limitation years beginning after December 31, 2001 or such earlier date specified by the Employer elsewhere in this Plan, compensation shall include deemed section 125 compensation. Deemed section 125 compensation is an amount that is excludable under Code section 106 that is not available to a participant in cash in lieu of group health coverage under a section 125 arrangement solely because the participant is unable to certify that he or she has other health coverage. Amounts are deemed section 125 compensation only if the Employer does not

request or collect information regarding the Employee's other health coverage as part of the enrollment process for the health plan.

- (vii) Compensation for Code section 415 purposes shall be limited under Code section 401(a)(17)(A) in the same manner as described in section 1.1.29 above.
- 3.6.9(b)-1 <u>Final Section 415 Regulations Compensation</u>. The following shall apply to the definition of compensation for Code section 415 purposes effective for Limitation Years beginning on or after July 1, 2007:
- 3.6.9(b)-2 <u>Default Provisions</u>. Unless the Employer elects otherwise in the Adoption Agreement, the following defaults will apply:
  - a. The provisions of the Plan setting forth the definition of compensation for purposes of Code § 415 (hereinafter referred to as "415 Compensation"), as well as compensation for purposes of determining highly compensated employees pursuant to Code § 414(q) and for top-heavy purposes under Code § 416 (including the determination of key employees) which top-heavy provisions shall not apply if this Plan is a governmental plan as defined in Code section 414(d) shall be modified by (1) including payments for unused sick, vacation or other leave and payments from nonqualified unfunded deferred compensation plans (Section 3.6.9(b)-3(b) below, (2) excluding salary continuation payments for Employees on military service (Section 3.6.9(b)-3(c) below), and (3) excluding salary continuation payments for disabled Employees (Section 3.6.9(b)-3(d) below).
  - b. The "first few weeks rule" does not apply for purposes of 415 Compensation (Section 3.6.9(b)-4 below).
  - c. 415 Compensation shall include any amount described in section 3.6.9(b)-3, 3.6.9(b)-4 or 3.6.9(b)-5 that is included as compensation for purposes of making or allocating contributions under the Plan, notwithstanding any election in Section "K" of the Adoption Agreement to the contrary.
- 3.6.9(b)-3 415 Compensation Paid After Severance From Employment. 415 Compensation shall be adjusted, as set forth herein, subject to 3.6.9(b)-2 above and as otherwise elected in Adoption Agreement Section "K", for the following types of compensation paid after an Employee's severance from employment with the Employer maintaining the Plan (or any other entity that is treated as the Employer pursuant to Code § 414(b), (c), (m) or (o)). However, amounts described in subsections (a) and (b) below may only be included in 415 Compensation to the extent such amounts are paid by the later of 2 1/2 months after severance from employment or by the end of the Limitation Year that includes the date of such severance from employment, if later. Any other payment of compensation paid after severance of employment that is not described in the following types of compensation is not considered 415 Compensation within the meaning of Code § 415(c)(3), even if payment is made within the time period specified above.
  - a. Regular pay. 415 Compensation shall include regular pay after severance of employment if:
    - (1) The payment is regular compensation for services during the participant's regular working hours, or compensation for services outside the participant's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments; and
    - (2) The payment would have been paid to the participant prior to a severance from employment if the participant had continued in employment with the Employer.

- b. Leave cashouts and deferred compensation. Leave cashouts shall be included in 415 Compensation, unless otherwise elected in Section "K" of the Adoption Agreement, if those amounts would have been included in the definition of 415 Compensation if they were paid prior to the participant's severance from employment, and the amounts are payment for unused accrued bona fide sick, vacation, or other leave, but only if the participant would have been able to use the leave if employment had continued. In addition, deferred compensation shall be included in 415 Compensation, unless otherwise elected in Section "K" of the Adoption Agreement, if the compensation would have been included in the definition of 415 Compensation if it had been paid prior to the participant's severance from employment, and the compensation is received pursuant to a nonqualified unfunded deferred compensation plan, but only if the payment would have been paid at the same time if the participant had continued in employment with the Employer and only to the extent that the payment is includible in the participant's gross income.
- c. Salary continuation payments for military service participants. 415 Compensation does not include, unless otherwise elected in Section "K" of the Adoption Agreement, payments to an individual who does not currently perform services for the Employer by reason of qualified military service (as that term is used in Code § 414(u)(1)). If an election is made in the Adoption Agreement to include such payments, they shall be included to the extent those payments do not exceed the amounts the individual would have received if the individual had continued to perform services for the Employer rather than entering qualified military service.
- d. Salary continuation payments for disabled Participants. Unless otherwise elected in Section "K" of the Adoption Agreement, 415 Compensation does not include compensation paid to a participant who is permanently and totally disabled (as defined in Code § 22(e)(3)). If elected, this provision shall apply to either just non-highly compensated participants, immediately before becoming disabled, or to all participants, for the period specified in Section "K" of the Adoption Agreement.

Back pay, within the meaning of Treasury Regulations section 1.415(c)-2(g)(8) shall be treated as compensation for the limitation year to which the back pay relates to the extent the back pay represents wages and compensation that would otherwise be included under this definition.

- 3.6.9(b)-4 <u>Administrative Delay ("The First Few Weeks") Rule</u>. 415 Compensation for a Limitation Year shall not include, unless otherwise elected in Section "K" of the Adoption Agreement, amounts earned but not paid during the Limitation Year solely because of the timing of pay periods and pay dates. However, if elected in Section "K" of the Adoption Agreement, 415 Compensation for a Limitation Year shall include amounts earned but not paid during the Limitation Year solely because of the timing of pay periods and pay dates, provided the amounts are paid during the first few weeks of the next Limitation Year, the amounts are included on a uniform and consistent basis with respect to all similarly situated participants, and no compensation is included in more than one Limitation Year.
- 3.6.9(b)-5 <u>Inclusion of Certain Nonqualified Deferred Compensation Amounts</u>. If the Plan's definition of compensation for purposes of Code § 415 is the definition in Regulation Section 1.415(c)-2(b) (Regulation Section 1.415-2(d)(2) under the Regulations in effect for Limitation Years beginning prior to July 1, 2007) and the simplified compensation definition of Regulation 1.415(c)-2(d)(2) (Regulation Section 1.415-2(d)(10) under the Regulations in effect for Limitation Years prior to July 1, 2007) is not used, then 415 Compensation shall include amounts that are includible in the gross income of a Participant under the rules of Code § 409A or Code § 457(f)(1)(A) or because the amounts are constructively received by the Participant. [Note if the

Plan's definition of compensation is W-2 wages or wages for withholding purposes, then these amounts are already included in compensation.]

- (c) Employer For purposes of this Section 3.6, Employer shall mean the Employer that adopts this Plan, and all members of a controlled group of corporations (as defined in Code section 414(b) as modified by Code section 415(h)), all commonly controlled trades or businesses (as defined in Code section 414(c) as modified, except in the case of a brother-sister group of trades or businesses under common control, by section 415(h)), or affiliated service groups (as defined in Code section 414(m)) of which the adopting Employer is a part , and any other entity required to be aggregated with the Employer pursuant to Code section 414(o).
- (d) <u>Excess Amount</u> The excess of the Participant's annual additions for the Limitation Year over the maximum permissible amount. Effective for Limitation Years beginning on or after July 1, 2007 and notwithstanding any provision of the Plan to the contrary, if the annual additions (within the meaning of Code § 415) are exceeded for any Participant, then the Plan may only correct such excess in accordance with the Employee Plans Compliance Resolution System (EPCRS) as set forth in Revenue Procedure 2006-272008-50-50 (Revenue Procedure 2013-12 effective April 1, 2013) or any amended or superseding guidance, including, but not limited to, the preamble of the final §415 regulations.
- (e) <u>Master or Prototype Plan</u> a plan the form of which is the subject of a favorable opinion letter from the Internal Revenue Service.
- (f) <u>Maximum Permissible Amount</u> For Limitation Years beginning on or after January 1, 2002, the maximum annual addition that may be contributed or allocated to a Participant's account under the Plan for any Limitation Year shall not exceed the lesser of:
  - (i) \$40,000, as adjusted for increases in the cost-of-living under Section 415(d) of the Code, or
    - (ii) 100 percent of the Participant's compensation for the Limitation Year.

The compensation limitation referred to in (ii) shall not apply to any contribution for medical benefits after separation from service (within the meaning of Code section 401(h) or Code section 419A(f)(2)) which is otherwise treated as an annual addition.

If a short Limitation Year is created because of an amendment changing the Limitation Year to a different 12-consecutive month period, the maximum permissible amount will not exceed the defined contribution dollar limitation multiplied by the following fraction:

# Number of months in the short Limitation Year

12

If the Plan is terminated as of a date other than the last day of the Limitation Year, the Plan is deemed to have been amended to change its Limitation Year and the Maximum Permissible Amount shall be prorated for the short Limitation Year.

- (g) <u>Aggregation and Disaggregation of Plans</u>. Effective for Limitation Years beginning on or after July 1, 2007 and notwithstanding any provision of the Plan to the contrary,
  - a. For purposes of applying the limitations of Code § 415, all defined contribution plans (without regard to whether a plan has been terminated) ever maintained by the Employer (or a "predecessor employer") under which the participant receives annual additions are treated as one defined contribution plan. Likewise all defined benefit plans ever maintained by the Employer are treated as one defined benefit plan. The "Employer" means the Employer that adopts this Plan and all members of a

controlled group or an affiliated service group that includes the Employer (within the meaning of Code §§ 414(b), (c), (m) or (o)), except that for purposes of this Section, the determination shall be made by applying Code § 415(h), and shall take into account tax-exempt organizations under Regulation Section 1.414(c)-5, as modified by Regulation Section 1.415(a)-1(f)(1). For purposes of this Section:

- (1) A former Employer is a "predecessor employer" with respect to a participant in a plan maintained by an Employer if the Employer maintains a plan under which the participant had accrued a benefit while performing services for the former Employer, but only if that benefit is provided under the plan maintained by the Employer. For this purpose, the formerly affiliated plan rules in Regulation Section 1.415(f)-1(b)(2) apply as if the Employer and predecessor Employer constituted a single employer under the rules described in Regulation Section 1.415(a)-1(f)(1) and (2) immediately prior to the cessation of affiliation (and as if they constituted two, unrelated employers under the rules described in Regulation Section 1.415(a)-1(f)(1) and (2) immediately after the cessation of affiliation) and cessation of affiliation was the event that gives rise to the predecessor employer relationship, such as a transfer of benefits or plan sponsorship.
- (2) With respect to an Employer of a participant, a former entity that antedates the Employer is a "predecessor employer" with respect to the participant if, under the facts and circumstances, the Employer constitutes a continuation of all or a portion of the trade or business of the former entity.
- Break-up of an affiliate employer or an affiliated service group. For b. purposes of aggregating plans for Code § 415, a "formerly affiliated plan" of an employer is taken into account for purposes of applying the Code § 415 limitations to the employer, but the formerly affiliated plan is treated as if it had terminated immediately prior to the "cessation of affiliation." For purposes of this paragraph, a "formerly affiliated plan" of an employer is a plan that, immediately prior to the cessation of affiliation, was actually maintained by one or more of the entities that constitute the employer (as determined under the employer affiliation rules described in Regulation Section 1.415(a)-1(f)(1) and (2)), and immediately after the cessation of affiliation, is not actually maintained by any of the entities that constitute the employer (as determined under the employer affiliation rules described in Regulation Section 1.415(a)-1(f)(1) and (2)). For purposes of this paragraph, a "cessation of affiliation" means the event that causes an entity to no longer be aggregated with one or more other entities as a single employer under the employer affiliation rules described in Regulation Section 1.415(a)-1(f)(1) and (2) (such as the sale of a subsidiary outside a controlled group), or that causes a plan to not actually be maintained by any of the entities that constitute the employer under the employer affiliation rules of Regulation Section 1.415(a)- 1(f)(1) and (2) (such as a transfer of plan sponsorship outside of a controlled group).
- c. Midyear Aggregation. Two or more defined contribution plans that are not required to be aggregated pursuant to Code § 415(f) and the Regulations thereunder as of the first day of a limitation year do not fail to satisfy the requirements of Code § 415 with respect to a participant for the limitation year merely because they are aggregated later in that limitation year, provided that no annual additions are credited to the participant's account after the date on which the plans are required to be aggregated.
- (h) <u>Predecessor Employer</u> If the Employer maintains a plan that provides a benefit which the Participant accrued while performing services for a former employer, the former employer is a predecessor employer with respect to the participant in the plan. A former entity that antedates the Employer is also a predecessor employer with respect to a Participant if, under

the facts and circumstances, the Employer constitutes a continuation of all or a portion of the trade or business of the former entity.

(i) <u>Annual Benefit</u> - A benefit that is payable annually in the form of a straight life annuity. Except as provided below or in the regulations under section 415, where a benefit is payable in a form other than a straight life annuity, the benefit shall be adjusted to an actuarially equivalent straight life annuity that begins at the same time as such other form of benefit and is payable on the first day of each month, before applying the limitations of this section 3.6.

The following provisions incorporate certain provisions of the Pension Funding Equity Act of 2004 ("PFEA"), as modified by the Pension Protection Act of 2006 and Worker, Retiree and Employer Recovery Act of 2008. Except as otherwise provided herein, expective for distributions in Plan Years beginning after December 31, 2003, the required determination of actuarial equivalence of forms of benefit other than a straight life annuity shall be made in accordance with "I" or "II" below, as applicable. However, these provisions shall not supersede any prior election to apply the transition rule of section 101(d)(3) of the Pension Funding Equity Act of 2004 as described in IRS Notice 2004-78.

- I. Benefit Forms Not Subject to section 417(e)(3). The straight life annuity that is actuarially equivalent to the Participant's form of benefit shall be determined under this section "I" if the form of the Participant's benefit is either (i) a nondecreasing annuity (other than a straight life annuity) payable for a period of not less than the life of the participant (or, in the case of a qualified pre-retirement survivor annuity, the life of the surviving spouse), or (ii) an annuity that decreases during the life of the participant merely because of (a) the death of the survivor annuitant (but only if the reduction is not below 50% of the benefit payable before the death of the survivor annuitant), or (b) the cessation or reduction of Social Security supplements or qualified disability payments (as defined in Code section 401(a)(11)).
  - a. Limitation Years Beginning Before July 1, 2007. For limitation years beginning before July 1, 2007, the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the participant's form of benefit computed using whichever of the following produces the greater annual amount: (i) the interest rate and mortality table (or other tabular factor) specified in Appendix B for adjusting benefits in the same form; and (ii) a 5% interest rate assumption and the applicable mortality table under Code section 417 as set forth in Rev. Rul. 2001 62defined in Appendix B for that annuity starting date.
  - b. Limitation Years Beginning On or After July 1, 2007. For limitation years beginning on or after July 1, 2007, the actuarially equivalent straight life annuity is equal to the greater of (i) the annual amount of the straight life annuity (if any) payable to the participant under the plan commencing at the same annuity starting date as the participant's form of benefit; and (ii) the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the participant's form of benefit computed using a 5% interest rate assumption and the applicable mortality table under Code section 417 as set forth in Rev. Rul. 2001 62 defined in Appendix B for that annuity starting date.
- II. Benefit Forms Subject to Code Section 417(e)(3): The straight life annuity that is actuarially equivalent to the participant's form of benefit shall be determined under this paragraph—subsection II if the form of the participant's benefit is other than a benefit form described in subsection "I" above. In this case, the actuarially equivalent straight life annuity shall be determined as follows:

a. Annuity Starting Date in Small Plans for Plan Years Beginning in 2009 and Later. Notwithstanding anything in this Plan to the contrary, if the annuity starting date of the Participant's form of benefit is in a Plan Year beginning in or after 2009, and if the Plan is maintained by an eligible employer as defined in Code Section 408(p)(2)(C)(i), the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using whichever of the following produces the greater annual amount:

(i) The interest rate and the mortality table (or other tabular factor) specified in Appendix B for adjusting benefits in the same form; and

(ii) 5.5 percent interest rate assumption and the applicable mortality table described in Appendix B

Annuity Starting Date in Plan Years Beginning After 2005. Except as provided in immediately preceding subsection "a", Iif the annuity starting date of the participant's form of benefit is inoccurs during a plan year beginning after December 31, 2005, the actuarially equivalent straight life annuity is equal to the greatest of (i) the annual amount of the straight life annuity commencing at the same annuity stating date that has the same actuarial present value as the participant's form of benefit, computed using the interest rate and mortality table (or other tabular factor) specified in Appendix B of the Plan for adjusting benefits in the same form; (ii) the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the participant's form of benefit, computed using a 5.5% interest rate assumption and the applicable mortality table for the distribution under Treasury Regulations Section 1.417(e)-1(d)(2), (determined in accordance with Code section 417Appendix B-as set forth in Rev. Rul. 2001-62; and (iii) the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the participant's form of benefit, computed using the applicable interest rate for the distribution under Treasury Regulations Section 1.417(e)-1(d)(3) (determined in accordance with Appendix B) on 30 year Treasury securities as specified by the Commissioner for the lookback month for the stability period, and the applicable mortality table for the distribution under Treasury Regulations Section 1.417(e)-1(d)(2) (determined in accordance with defined in Appendix B), divided by 1.05.

Unless otherwise elected by the Employer in a written addendum to this Plan, (i) the stability period is the successive period of one calendar month which contains the annuity starting date for the distribution and for which the applicable interest rate remains constant, and (ii) the lookback month is the first calendar month preceding the first day of the stability period. Notwithstanding anything in this paragraph or election of the Employer regarding stability period or lookback month, a plan amendment that changes the date for determining the applicable interest rate (including an indirect change as a result of a change in plan year), shall not be given effect with respect to any distribution during the period commencing one year after the later of the amendment's effective date or adoption date, if, during such period and as a result of such amendment, the participant's distribution would be reduced.

c. Annuity Starting Date in Plan Years Beginning in 2004 or 2005. If the annuity starting date of the participant's form of benefit is in a plan year beginning in 2004 or 2005, the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at

the same annuity starting date that has the same actuarial present value as the participant's form of benefit, computed using whichever of the following produces the greater annual amount: (i) the interest rate and mortality table (or other tabular factor) specified in Appendix B of the plan for adjusting benefits in the same form; and (ii) a 5.5% interest rate assumption and the applicable mortality table for the distribution under Code section 417Treasury Regulations Section 1.417(e)-1(d)(2) as set forth in Rev. Rul. 2001 62.

- (b) However, this subsection does not supersede any prior election to apply the transition rule of section 101(d)(3) as described in Notice 2004-78.
- (j) RPA '94 Old Law Benefit. The Participant's Accrued Benefit under the terms of the Plan as of January 1, 2000 (the RPA '94 Freeze Date), for the Annuity Starting Date and optional form and taking into account the limitations of Code section 415, as in effect on December 7, 1994, including the participation requirements under Code section 415(b)(5). In determining the amount of a Participant's RPA '94 Old Law Benefit, the following shall be disregarded:
  - I. any plan amendment increasing benefits adopted after the RPA '94 Freeze Date; and
    - II. any cost of living adjustments that become effective after such date.

A Participant's RPA '94 Old Law Benefit is not increased after the RPA '94 Freeze Date, but if the limitations of Code section 415, as in effect on December 7, 1994, are less than the limitations that were applied to determine the Participant's RPA '94 Old Law Benefit on the RPA '94 Freeze Date, then the Participant's RPA '94 Old Law Benefit will be reduced in accordance with such reduced limitation. If, at any date after the RPA '94 Freeze Date, the Participant's total plan benefit, before the application of Code section 415, is less than the Participant's RPA '94 Old Law Benefit, the RPA '94 Old Law Benefit will be reduced to the Participant's total plan benefit.

- (k) <u>Employee Contributions</u>. Employee contributions, rollovers and transfers, and benefits or limitations attributable or applicable thereto, shall be determined in accordance with Treasury Regulations, section 1.415(b)-1(b)(2) and (3).
- (l) <u>Benefit Increases</u>. Benefit increases as a result of an increase in a limitation of Code section 415, if any, shall be provided to all Employees participating in the Plan having at least one hour of service on or after the first day of the limitation year ending on or after the date that the change becomes effective
- (m) <u>Limitation Years</u>. Provisions of this section 3.6 adopted to comply with final Treasury Regulations under Code section 415 shall be effective for limitation years beginning on or after July 1, 2007, and application of such provisions shall not cause applicable Code section 415 limitations for any Participant to be less than the Participant's benefits under all qualified plans of the Employer as of the end of the limitation year beginning before July 1, 2007 under provisions of the plans that were both adopted and in effect before April 5, 2007; provided the provisions of such plans both adopted and in effect before April 5, 2007 satisfied all applicable laws, rules and regulations in effect as of the end of the last limitation year beginning before July 1, 2007, as described in Treasury Regulations, section 1.415(a)-1(g)(4)
- (n) <u>Maximum Permissible Benefit</u>. Pursuant to Code section 415(b)(11), the maximum permissible benefit for a governmental plan described in Code section 414(d) means the defined benefit dollar limitation set forth in Code section 415(b)(1)(A), subject to adjustment pursuant to Code section 415(d) periodically and as follows.

- I. Adjustment for Less Than 10 Years of Participation or Service: If the Participant has less than 10 years of participation in the Plan, the defined benefit dollar limitation shall be multiplied by a fraction (i) the numerator of which is the number of years (or part thereof, but not less than one year) of participation in the Plan, and (ii) the denominator of which is 10, pursuant to the rules set forth in Treasury Regulations, Section 1.415(b)-1(g)((1).
- II. Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement Before Age 62 or after Age 65: Effective for benefits commencing in limitation years ending after December 31, 2001. the defined benefit dollar limitation shall be adjusted if the annuity starting date of the participant's benefit is before age 62 or after age 65, as follows.
  - a. Adjustment of defined benefit dollar limitation for benefit commencement before age 62:
    - I. Limitation Years Beginning Before July 1, 2007. If the annuity starting date for the participant's benefit is prior to age 62 and occurs in a limitation year beginning before July 1, 2007, the defined benefit dollar limitation for the participant's annuity starting date is the annual amount of a benefit payable in a straight life annuity commencing at the participant's annuity starting date that is the actuarial equivalent of the defined benefit dollar limitation (adjusted above as necessary if less than 10 years of participation) with actuarial equivalence computed using whichever of the following produces the smaller annual amount (i) the interest rate specified in Appendix B and the mortality table (or other tabular factor) specified in Appendix B; or (ii) a 5-percent interest rate assumption and the applicable mortality table under Code section 417 as set forth in Rev. Rul. 2001-62.

II. Limitation Years Beginning on or After July 1 2007.

A. Plan Does Not Have Immediately Commencing Straight Life Annuity Payable at Both Age 62 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a limitation year beginning on or after July 1, 2007, and the Plan does not have an immediately commencing straight life annuity payable at both age 62 and the age of benefit commencement, the defined benefit dollar limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the defined benefit dollar limitation (adjusted above for years of participation less than 10, if required) with actuarial equivalence computed using a 5 percent interest rate assumption and the applicable mortality table for the annuity starting date as under Code section 417 as set forth in Rev. Rul. 2001-62 (and

expressing the Participant's age based on completed calendar months as of the annuity starting date).

- B. Plan Has Immediately Commencing Straight Life Annuity Payable at Both Age 62 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a limitation year beginning on or after July 1, 2007, and the Plan has an immediately commencing straight life annuity payable at both age 62 and the age of benefit commencement, the defined benefit dollar limitation for the Participant's annuity starting date is the lesser of the limitation determined under subsection A above and the defined benefit dollar limitation (adjusted above for years of participation less than 10, if required) multiplied by the ratio of the annual amount of the immediately commencing straight life annuity under the Plan at the Participant's annuity starting date to the annual amount of the immediately commencing straight life annuity under the Plan at age 62, both determined without applying the limitations of this Section 3.6.
- C. Notwithstanding any other provisions of this subsection 3.6.9(n)(II)(a), the age-adjusted dollar limit applicable to a Participant shall not decrease on account of an increase in age or the performance of additional services.
- b. Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement After Age 65.
  - I. Limitation Years Beginning Before July 1, 2007. If the annuity starting date for the participant's benefit is after age 65 and occurs in a limitation year beginning before July 1, 2007, the defined benefit dollar limitation for the participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the participant's annuity starting date that is the actuarial equivalent of the defined benefit dollar limitation with actuarial equivalence computed using whichever of the following produces the smaller amount: (1) the interest rate and mortality table specified in Appendix B or (2) a 5% interest rate assumption and the applicable mortality table as defined under Code section 417 as set forth in Rev. Rul. 2001-62.
  - II. Limitation Years Beginning on or after July 1, 2007.
    - A. <u>Plan Does Not Have Immediately Commencing Straight Life Annuity Payable at Both Age 65 and the Age of Benefit Commencement.</u> If the annuity starting date for the Participant's benefit is after age 65 and occurs in a limitation year beginning on or after July 1, 2007, and the Plan does

not have an immediately commencing straight life annuity payable at both age 65 and the age of benefit commencement, the defined benefit dollar limitation at the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the defined benefit dollar limitation (adjusted above for years of participation less than 10, if required), with actuarial equivalence computed using a 5 percent interest rate assumption and the applicable mortality table for that annuity starting date as defined under Code section 417 as set forth in Rev. Rul. 2001-62 (and expressing the Participant's age based on completed calendar months as of the annuity starting date).

B. Plan Has Immediately Commencing Straight Life Annuity Payable at Both Age 65 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a limitation year beginning on or after July 1, 2007, and the Plan has an immediately commencing straight life annuity payable at both age 65 and the age of benefit commencement, the defined benefit dollar limitation at the Participant's annuity starting date is the lesser of the limitation determined under preceding subsection A and the defined benefit dollar limitation (adjusted above for years of participation less than 10, if required) multiplied by the ratio of the annual amount of the adjusted immediately commencing straight life annuity under the Plan at the Participant's annuity starting date to the annual amount of the adjusted immediately commencing straight life annuity under the Plan at age 65, both determined without applying the limitations of this Section 3.6. For this purpose the adjusted immediately commencing straight life annuity under the Plan at the Participant's annuity starting date is the annual amount of such annuity payable to the Participant, computed disregarding the Participant's accruals after age 65 but including actuarial adjustments even if those actuarial adjustments are used to offset accruals; and the adjusted immediately commencing straight life annuity under the Plan at age 65 is the annual amount of such annuity that would be payable under the Plan to a hypothetical Participant who is age 65 and has the same accrued benefit as the Participant.

c. Notwithstanding the other requirements of this subsection "3.6.9(n)II" no adjustment shall be made to in adjusting the defined benefit dollar limitation for the Participant's annuity starting date under section 3.6.9(n)(II)(a)(I), 3.6.9(n)(II)(b)(I), 3.6.9(n)(II)(b)(II)(A), no adjustment shall be made to reflect the probability of a Participant's death between the annuity

starting date and age 62, or between age 65 and the annuity starting date, as applicable, if benefits are not forfeited upon the death of the Participant prior to the annuity starting date. To the extent benefits are forfeited upon death before the annuity starting date, such an adjustment shall be made. For this purpose no forfeiture shall be treated as occurring upon the Participant's death if the Plan does not charge Participants for providing a qualified preretirement survivor annuity, if any, as defined in Section 417(c) of the Internal Revenue Code, upon the Participant's death.

Adjustments provided under the Code for benefit commencement before age 62 shall not apply to a governmental plan with respect to any "qualified participant" pursuant to Treasury Regulations section 1.415(b)-1(d)(3).

- (o) The application of this Section 3.6 revised for final regulations under Code section 415 adopted April 5, 2007 shall not cause the Maximum Permissible Benefit or Amount of any Participant to be less than the Participant's benefit or allocation under this and all other defined benefit or defined contribution plans of the Employer or predecessor employer as of the end of the limitation year beginning before July 1, 2007 under provisions of the plans that were both adopted and in effect before April 5, 2007. The preceding sentence applies only if the provisions of such plans that were both adopted and in effect before April 5, 2007 satisfied the applicable requirements of statutory provisions, regulations, and other published guidance relating to Code section 415 in effect as of the end of the last limitation year beginning before July 1, 2007, as described in Treasury Regulations section 1.415(a)-1(g)(4).
- 3.7 Code Section 436 Benefit Restrictions. Except as otherwise provided herein or by applicable law, regulations or other guidance, benefits restrictions of Code Section 436 apply to single employer plans in Plan Years beginning after December 31, 2007.
  - (a) **Governmental Plan Exception.** Notwithstanding anything in this Section 3.7 to the contrary, Code section 436, the benefit restrictions thereunder and the provisions of Section 3.7 below shall not apply to this Plan to the extent it is a governmental plan within the meaning of Code section 414(d) and exempt from the requirements of Code sections 401(a)(29) and 436 by reason of being exempt from the funding requirements of Code section 412.
  - 3.7.1 **2008 and 2009 Plan Years**. The provisions of this Section 3.7.1 shall apply to Plan Years beginning after December 31, 2007 and before January 1, 2010.

#### (a) **Application of Section 3.7.1**.

- (1) This Section 3.7.1 only applies to single employer plans (a plan that is not a multiemployer plan within the meaning of Code Section 414(f)) and does not apply to a plan maintained pursuant to one or more collective bargaining agreements between employee representatives and one or more employers.
- (A) Multiple Employer Plans. In the case of a multiple employer plan to which Code Section 413(c)(4)(A) applies, Code Section 436 applies separately with respect to each employer under the plan, as if each employer maintained a separate plan. Thus, the benefit limitations under Code Section 436 could apply differently to participants who are employees of different employers under such a multiple employer plan. In the case of a multiple employer plan to which Code Section 413(c)(4)(A) does not apply (that is, a plan described in Code Section 413(c)(4)(B) that has not made the election for Code Section 413(c)(4)(A) to apply), Code Section 436 applies as if all participants in the plan were employed by a single employer.
- (B) Governmental Plans. Code Section 436 benefit restrictions and other provisions described in this Section 3.7.1 shall not apply to this Plan to the extent it is a governmental plan

within the meaning of Code Section 414(d) and exempt from the requirements of Code Sections 401(a)(29) and 436 by reason of being exempt from the funding requirements of Code Section 412.

- (2) The limitations described in Subsections (b), (c) and (e) do not apply to the Plan for the first five (5) Plan Years of the Plan. Except as otherwise provided by the Commissioner in guidance of general applicability, the Plan Years taken into account for this purpose include the following (in addition to Plan Years during which the Plan was maintained by the Employer):
  - (A) Plan Years when the Plan was maintained by a predecessor employer within the meaning of Regulations Section 1.415(f)-1(c)(1);
  - (B) Plan years of another defined benefit plan maintained by a predecessor employer within the meaning of Regulations Section 1.415(f)-1(c)(2) within the preceding five years if any Participants in the Plan participated in that other defined benefit plan (even if the Plan maintained by the Employer is not the plan that was maintained by the predecessor employer); and
  - (C) Plan years of another defined benefit plan maintained by the Employer within the preceding five years if any Participants in the Plan participated in that other defined benefit plan.
- (3) Notwithstanding anything in this Section 3.7.1 to the contrary, the provision of Code Section 436 and the Regulations thereunder are incorporated herein by reference.
- (4) For Plans that have a valuation date other than the first day of the Plan Year, the provisions of Code Section 436 and this Section 3.7.1 will be applied in accordance with Regulations.

# (b) Funding-Based Limitation on Shutdown Benefits and Other Unpredictable Contingent Event Benefits

- (1) In general. If a Participant is entitled to an "unpredictable contingent event benefit" payable with respect to any event occurring during any Plan Year, then such benefit may not be provided if the "adjusted funding target attainment percentage" for such Plan Year (A) is less than sixty percent (60%) or, (B) sixty percent (60%) or more, but would be less than sixty percent (60%) percent if the "adjusted funding target attainment percentage" were redetermined applying an actuarial assumption that the likelihood of occurrence of the "unpredictable contingent event" during the Plan Year is one hundred percent (100%).
- (2) Exemption. Paragraph (1) shall cease to apply with respect to any Plan Year, effective as of the first day of the Plan Year, upon payment by the Employer of the contribution described in Regulations Section 1.436-1(f)(2)(iii).

#### (c) Limitations on Plan Amendments Increasing Liability for Benefits

- (1) In general. No amendment which has the effect of increasing liabilities of the Plan by reason of increases in benefits, establishment of new benefits, changing the rate of benefit accrual, or changing the rate at which benefits become nonforfeitable may take effect during any Plan Year if the "adjusted funding target attainment percentage" for such Plan Year is:
  - (A) less than eighty percent (80%), or
  - (B) eighty percent (80%) or more, but would be less than eighty percent (80%) if the benefits attributable to the amendment were taken into account in determining the "adjusted funding target attainment percentage."

- (2) Exemption. Paragraph (c)(1) above shall cease to apply with respect to a Plan amendment upon payment by the Employer of the contribution described in Regulations Section 1.436-1(f)(2)(iv).
- (3) Exception for certain benefit increases. Paragraph (1) shall not apply to any amendment as otherwise provided in Regulations Section 1.436-1(c).

## (d) Limitations on Prohibited Payments

- (1) Funding percentage less than sixty percent (60%). If the Plan's "adjusted funding target attainment percentage" for a Plan Year is less than sixty percent (60%), then a Participant or Beneficiary shall not be permitted to elect, and the Plan may not pay, any "prohibited payment" with an "annuity starting date" on or after the applicable "Section 436 measurement date."
- Bankruptcy. A Participant or Beneficiary shall not be permitted to elect, and the Plan may not pay, any "prohibited payment" with an "annuity starting date" that occurs during any period in which the Employer is a debtor in a case under Title 11, United States Code, or similar Federal or State law. The preceding sentence shall not apply to payments made within a Plan Year with an "annuity starting date" that occurs on or after the date on which the enrolled actuary of the Plan certifies that the "adjusted funding target attainment percentage" of the Plan is not less than one hundred percent (100%).
- (3) Limited payment if percentage at least sixty percent (60%) but less than eighty percent (80%) percent.
  - (A) In general. If the Plan's "adjusted funding target attainment percentage" for a Plan Year is sixty percent (60%) or greater but less than eighty percent (80%), then a Participant or Beneficiary shall not be permitted to elect, and the Plan may not pay, any "prohibited payment" with an "annuity starting date" on or after the applicable "Section 436 measurement date," unless the present value (determined in accordance with Code Section 417(e)(3)) of the portion of the benefit that is being paid in a "prohibited payment" (which portion is determined under paragraph (C)(i) below) does not exceed the lesser of:
    - (i) fifty (50) percent of the amount of the present value (determined in accordance with Code Section 417(e)(3)) of the benefit payable in the optional form of benefit that includes the prohibited payment; or
    - (ii) 100% of the "PBGC maximum benefit guarantee amount."
  - (B) Bifurcation if optional form unavailable.
    - (i) Requirement to offer bifurcation. If an optional form of benefit that is otherwise available under the terms of the plan is not available as of the "annuity starting date" because of the application of Regulations Section 1.436-1(d)(3)(i), then the Participant or Beneficiary may elect to:
      - (1) Receive the unrestricted portion of that optional form of benefit (determined under the rules of Regulations Section 1.436-1(d)(3)(iii)(D)) at that "annuity starting date," determined by treating the unrestricted portion of the benefit as if it were the Participant's or Beneficiary's entire benefit under the plan:
      - (2) Commence benefits with respect to the Participant's or Beneficiary's entire benefit under the Plan in any other optional form of

benefit available under the Plan at the same "annuity starting date" that satisfies Regulations Section 1.436-1(d)(3)(i); or

- (3) Defer commencement of the payments to the extent described in Regulations Section 1.436-1(d)(5).
- (ii) Rules relating to bifurcation. If the Participant or Beneficiary elects payment of the unrestricted portion of the benefit as described in Regulations Section 1.436-1(d)(3)(ii)(A)(1), then the Participant or Beneficiary may elect payment of the remainder of the Participant's or Beneficiary's benefits under the Plan in any optional form of benefit at that "annuity starting date" otherwise available under the Plan that would not have included a "prohibited payment" if that optional form applied to the entire benefit of the Participant or Beneficiary. The rules of Regulations Section 1.417(e)-1 are applied separately to the separate optional forms for the "unrestricted portion of the benefit" and the remainder of the benefit (the restricted portion).
- (iii) Plan alternative that anticipates election of payment that includes a "prohibited payment." With respect to every optional form of benefit that includes a "prohibited payment" and that is not permitted to be paid under Regulations Section 1.436-1 (d)(3)(i), for which no additional information from the Participant or Beneficiary (such as information regarding a Social Security leveling optional form of benefit) is needed to make that determination, rather than wait for the Participant or Beneficiary to elect such optional form of benefit, the Plan will provide for separate elections with respect to the restricted and unrestricted portions of that optional form of benefit.
- (C) Definitions applicable to limited payment option. The following definitions apply for purposes of this subsection (d)(3).
  - (i) Portion of benefit being paid in a prohibited payment. If a benefit is being paid in an optional form for which any of the payments is greater than the amount payable under a straight life annuity to the Participant or Beneficiary (plus any Social Security supplements described in the last sentence of Code Section 411(a)(9) payable to the Participant or Beneficiary) with the same "annuity starting date," then the portion of the benefit that is being paid in a "prohibited payment" is the excess of each payment over the smallest payment during the Participant's lifetime under the optional form of benefit (treating a period after the "annuity starting date" and during the Participant's lifetime in which no payments are made as a payment of zero).
  - (ii) PBGC maximum benefit guarantee amount. The "PBGC maximum benefit guarantee amount" is the present value (determined under guidance prescribed by the Pension Benefit Guaranty Corporation, using the interest and mortality assumptions under Code Section 417(e)) of the maximum benefit guarantee with respect to a Participant (based on the Participant's age or the Beneficiary's age at the "annuity starting date") under ERISA Section 4022 for the year in which the "annuity starting date" occurs.
  - (iii) Unrestricted portion of the benefit:
    - (1) General rule. Except as otherwise provided in this paragraph (iii), the unrestricted portion of the benefit with respect to any optional form of benefit is fifty percent (50%) of the amount payable under the optional form of benefit.

- (2) Special rule for forms which include Social Security leveling or a refund of employee contributions. For an optional form of benefit that is a prohibited payment on account of a Social Security leveling feature (as defined in Regulations Section 1.411(d)-3(g)(16)) or a refund of employee contributions feature (as defined in Regulations Section 1.411(d)-3(g)(11)), the unrestricted portion of the benefit is the optional form of benefit that would apply if the Participant's or Beneficiary's Accrued Benefit were fifty percent (50%) smaller.
- (3) Limited to PBGC maximum benefit guarantee amount. After the application of the preceding rules of this paragraph (iii), the unrestricted portion of the benefit with respect to the optional form of benefit is reduced, to the extent necessary, so that the present value (determined in accordance with Code Section 417(e)) of the unrestricted portion of that optional form of benefit does not exceed the "PBGC maximum benefit guarantee amount."

## (D) Other Rules.

- (i) One time application. If a Participant with respect to whom a prohibited payment (or a series of prohibited payments under a single optional form of benefit) is made pursuant to paragraph (d)(3)(A) or (B) above, no additional prohibited payment may be made with respect to that Participant during any consecutive Plan Years for which prohibited payments are limited under this subsection (d).
- (ii) Treatment of beneficiaries. For purposes of this subparagraph (d)(3), benefits provided with respect to a Participant and any Beneficiary of the Participant (including an alternate payee, as defined in Code Section 414(p)(8)) are aggregated. If the only benefits paid under the plan with respect to the Participant are death benefits payable to the Beneficiary, then paragraph (d)(3)(C)(i) of this section is applied by substituting the lifetime of the Beneficiary for the lifetime of the Participant. If the Accrued Benefit of a Participant is allocated to such an alternate payee and one or more other persons, then the "unrestricted amount" of (d)(3)(C)(iii) is allocated among such persons in the same manner as the accrued benefit is allocated, unless a qualified domestic relations order (as defined in Code Section 414(p)(1)(A)) with respect to the Participant or the alternate payee provides otherwise.
- (iii) Treatment of annuity purchases and plan transfers. This paragraph (d)(3)(D)(iii) applies for purposes of applying subsections (d)(3)(A) and (d)(3)(C)(iii). In the case of a prohibited payment described in Regulations Section 1.436-1(j)(6)(i)(B) (relating to purchase from an insurer), the present value of the portion of the benefit that is being paid in a prohibited payment is the cost to the plan of the irrevocable commitment and, in the case of a prohibited payment described in Regulations Section 1.436-1(j)(6)(i)(C) (relating to certain plan transfers), the present value of the portion of the benefit that is being paid in a prohibited payment is the present value of the liabilities transferred (determined in accordance with Code Section 414(I)). In addition, the present value of the accrued benefit is substituted for the present value of the benefit payable in the optional form of benefit that includes the prohibited payment in Regulations Section 1.436-1(d)(3)(i)(A).
- (4) Exception. This subsection (d) shall not apply for any Plan Year if the terms of the Plan (as in effect for the period beginning on September 1, 2005, and ending with such Plan Year) provide for no benefit accruals with respect to any Participant during such period.

- Right to delay commencement. If a Participant or Beneficiary requests a distribution in an optional form of benefit that includes a "prohibited payment" that is not permitted to be paid under paragraph (d)(1), (d)(2), or (d)(3) of this Section 3.7.1, then the Participant retains the right to delay commencement of benefits in accordance with the terms of the plan and applicable qualification requirements (such as Code Sections 411(a)(11) and 401(a)(9)).
- (6) "Prohibited payment." For purposes of this subsection (d), the term "prohibited payment" means:
  - (A) Any payment for a month that is in excess of the monthly amount paid under a single life annuity (plus any Social Security supplements described in the last sentence of Code Section 411(a)(9)), to a Participant or Beneficiary whose "annuity starting date" occurs during any period a limitation under paragraph (d) is in effect;
  - (B) Any payment for the purchase of an irrevocable commitment from an insurer to pay benefits; and
  - (C) Any transfer of assets and liabilities to another plan maintained by the same Employer (or by any member of the Employer's controlled group) that is made in order to avoid or terminate the application of Code Section 436 benefit limitations; and
  - (D) Any other amount that is identified as a prohibited payment by the Commissioner in revenue rulings and procedures, notices, and other guidance published in the Internal Revenue Bulletin.

Such term shall not include the payment of a benefit which under Code Section 411(a)(11) may be immediately distributed without the consent of the Participant. Furthermore, in the case of a Beneficiary that is not an individual, the amount that is a prohibited payment is determined by substituting the monthly amount payable in installments over 240 months that is actuarially equivalent to the benefit payable to the Beneficiary, as provided in Regulations Section 1.436-1(j)(6)(ii).

# (e) Limitation on Benefit Accruals for Plans with Severe Funding Shortfalls

- (1) In general. If the Plan's "adjusted funding target attainment percentage" for a Plan Year is less than sixty percent (60%), benefit accruals under the Plan shall cease as of the "section 436 measurement date." If the Plan is required to cease benefit accruals under this subsection (e), then the Plan is not permitted to be amended in a manner that would increase the liabilities of the Plan by reason of an increase in benefits or establishment of new benefits. The preceding sentence applies regardless of whether an amendment would otherwise be permissible under subsections (c)(2) or (c)(3) of this Section 3.7.1.
- (2) Exemption. Paragraph (1) shall cease to apply with respect to any Plan Year, effective as of the first day of the Plan Year, upon payment by the Employer of the contribution described in Regulations Section 1.436-1(f)(2)(v).
- (3) Temporary modification of limitation. In the case of the first Plan Year beginning during the period beginning on October 1, 2008, and ending on September 30, 2009, the provisions of (e)(1) above shall be applied by substituting the Plan's "adjusted funding target attainment percentage" for the preceding Plan Year for such percentage for such Plan Year, but only if the "adjusted funding target attainment percentage" for the preceding year is greater.

#### (f) Rules Relating to Contributions Required to Avoid or Terminate Benefit Limitations

The application of the Code Section 436 benefit limitations may be avoided or terminated in

#### (g) **Presumed Underfunding for Purposes of Benefit Limitations**

## (1) Presumption of continued underfunding.

(A) In general. This paragraph (g)(1) applies to a Plan for a Plan Year if a limitation under subsection (b), (c), (d), or (e) applied to the Plan on the last day of the preceding Plan Year. If this paragraph (g)(1) applies to a Plan, then the first day of the Plan Year is a "Section 436 measurement date" and the presumed "adjusted funding target attainment percentage" for the Plan is the percentage under paragraph (g)(1)(B) or (C) of this subsection, whichever applies to the Plan, beginning on that first day of the Plan Year and ending on the date specified in subparagraph (g)(1)(D) of this section.

#### (B) Rule where preceding year certification issued during preceding year.

- (i) General rule. In any case in which the Plan's enrolled actuary has issued a certification under Regulations Section 1.436-1(h)(4) of the "adjusted funding target attainment percentage" for the Plan Year preceding the current Plan Year before the first day of the current Plan Year, the presumed "adjusted funding target attainment percentage" of the Plan for the current Plan Year is equal to the prior Plan Year "adjusted funding target attainment percentage" until it is changed under Regulations Section 1.436-1(h)(1)(iv).
- (ii) Special rule for late certifications. If the certification of the adjusted funding target attainment percentage for the prior Plan Year occurred after the first day of the 10th month of that prior Plan Year, the Plan is treated as if no such certification was made, unless the certification took into account the effect of any unpredictable contingent event benefits that are permitted to be paid based on unpredictable contingent events that occurred, and any Plan amendments that became effective, during the prior Plan Year but before the certification (and any associated Code Section 436 contributions).

# (C) No certification for preceding year issued during preceding year.

- (i) Deemed percentage continues. In any case in which the Plan's enrolled actuary has not issued a certification under Regulations Section 1.436-1(h)(4) of the "adjusted funding target attainment percentage" of the Plan for the Plan Year preceding the current Plan Year during that prior Plan Year, the presumed "adjusted funding target attainment percentage" of the Plan for the current Plan Year is equal to the presumed "adjusted funding target attainment percentage" that applied on the last day of the preceding Plan Year until the presumed "adjusted funding target attainment percentage" is changed under Regulations Section 1.436-1(h)(1)(iii)(B) or (h)(1)(iv).
- (ii) Enrolled actuary's certification in following year. In any case in which the Plan's enrolled actuary has issued the certification under Regulations Section 1.436-1(h)(4) of the adjusted funding target attainment percentage of the Plan for the Plan Year preceding the current Plan Year on or after the first day of the current Plan Year, the date of that prior Plan Year certification is a new "Section 436 measurement date" for the current Plan Year. In such a case, the presumed adjusted funding target attainment percentage for the current Plan Year is equal to the prior Plan Year adjusted funding target attainment percentage (reduced by 10 percentage points if Regulations Section 1.436-1(h)(2)(iv) applies to the Plan) until it is changed under Regulations Section 1.436-1(h)(1)(iv). The rules of Regulations Section 1.436-1(h)(1)(ii)(B) apply for purposes of determining

- whether the enrolled actuary has issued a certification of the adjusted funding target attainment percentage for the prior Plan Year during the current Plan Year.
- (D) Duration of use of presumed "adjusted funding target attainment percentage." If this paragraph (g)(1) applies to a Plan for a Plan Year, then the presumed "adjusted funding target attainment percentage" determined under this paragraph (g)(1) applies until the earliest of:
  - (i) The first day of the 4th month of the Plan Year if paragraph (g)(2) of this section applies:
  - (ii) The first day of the 10th month of the Plan Year if paragraph (g)(3) of this section applies;
  - (iii) The date of a change in the presumed adjusted funding target attainment percentage under Regulations Section 1.436-1(g)(4); or
  - (iv) The date the enrolled actuary issues a certification under Regulations Section 1.436-1(h)(4) of the "adjusted funding target attainment percentage" for the Plan Year.
- Presumption of underfunding beginning on first day of 4th month for certain underfunded plans. This paragraph (2) applies to a Plan for a Plan Year if the enrolled actuary for the Plan has not issued a certification of the "adjusted funding target attainment percentage" for the Plan Year before the first day of the 4th month of the Plan Year, and the Plan's "adjusted funding target attainment percentage" for the preceding Plan Year was either (1) at least sixty percent (60%) but less than seventy percent (70%); or (2) at least eighty percent (80%) but less than ninety percent (90%). This paragraph (2) also applies to a Plan for the first effective Plan Year if the enrolled actuary for the Plan has not issued a certification of the "adjusted funding target attainment percentage" for the Plan Year before the first day of the 4th month of the Plan Year, and the prior Plan Year "adjusted funding target attainment percentage" is at least seventy percent (70%) but less than eighty percent (80%).
  - (A) Presumed adjusted funding target attainment percentage. Application of this paragraph. If this paragraph (2) applies to a Plan for a Plan Year and the date of the enrolled actuary's certification of the "adjusted funding target attainment percentage" under Regulations Section 1.436-1(h)(4) for the prior Plan Year (taking into account the special rules for late certifications under Regulations Section 1.436-1(h)(1)(ii)(B)) occurred before the first day of the 4th month of the current Plan Year, then, commencing on the first day of the 4th month of the current Plan Year:
    - (i) The presumed "adjusted funding target attainment percentage" of the Plan for the Plan Year is reduced by 10 percentage points; and
    - (ii) The first day of the 4th month of the Plan Year is a "Section 436 measurement date."
  - (B) Certification for prior Plan Year. If this paragraph (2) applies to a Plan and the date of the enrolled actuary's certification of the "adjusted funding target attainment percentage" under Regulations Section 1.436-1(h)(4) for the prior Plan Year (taking into account the rules for late certifications under Regulations Section 1.436-1(h)(1)(ii)(B)) occurs on or after the first day of the 4th month of the current Plan Year, then, commencing on the date of that prior Plan Year certification:

- (i) The presumed "adjusted funding target attainment percentage" of the Plan for the current Plan Year is equal to 10 percentage points less than the prior Plan Year "adjusted funding target attainment percentage"; and
- (ii) The date of the prior Plan Year certification is a "Section 436 measurement date."
- (C) Duration of use of presumed "adjusted funding target attainment percentage."

  If this paragraph (2) applies to a Plan for a Plan Year, the presumed adjusted funding target attainment percentage determined under this paragraph (2) applies until the earliest of:
  - (i) The first day of the 10th month of the Plan Year if paragraph (3) of this section applies;
  - (ii) The date of a change in the presumed "adjusted funding target attainment percentage" under Regulations Section 1.436-1(g)(4); or
  - (iii) The date the enrolled actuary issues a certification under Regulations Section 1.436-1(h)(4) of the "adjusted funding target attainment percentage" for the Plan Year.
- Presumption of underfunding beginning on first day of 10th month. In any case in which no certification of the specific adjusted funding target attainment percentage for the current Plan Year under Regulations Section 1.436-1(h)(4) is made with respect to the Plan before the first day of the 10th month of the Plan Year, then, commencing on the first day of the 10th month of the current Plan Year:
  - (A) The presumed "adjusted funding target attainment percentage" of the Plan for the Plan Year is presumed to be less than sixty percent (60%); and
  - (B) The first day of the 10th month of the Plan Year is a "Section 436 measurement date."

# (h) Treatment of Plan as of Close of Prohibited or Cessation Period.

- (1) Application to prohibited payments and accruals.
  - (A) Resumption of prohibited payments. If a limitation on prohibited payments under section (d) of this Section 3.7.1 applied to a Plan as of a "Section 436 measurement date," but that limit no longer applies to the Plan as of a later "Section 436 measurement date," then the limitation on prohibited payments under the Plan does not apply to benefits with "annuity starting dates" that are on or after that later "Section 436 measurement date." Any amendment to eliminate an optional form of benefit that contains a prohibited payment with respect to an "annuity starting date" during a period in which the limitations of Code Section 436(d) and Regulations Section 1.436-1(d) do not apply to the Plan is subject to the rules of Code Section 411(d)(6).
  - (B) Resumption of benefit accruals. If a limitation on benefit accruals under Regulations Section 1.436-1(e) applied to a Plan as of a "Section 436 measurement date," but that limit no longer applies to the Plan as of a later "Section 436 measurement date," then that limitation does not apply to benefit accruals that are based on service on or after that later "Section 436 measurement date," except to the extent that the Plan provides that benefit accruals will not resume when the limitation ceases to apply. The Plan will comply with the rules relating to partial years of participation and the prohibition on double proration under Department of Labor regulation 29 CFR Section 2530.204-2(c) and (d).

Grand Island

- Restoration of options and missed benefit accruals. Participants who had an "annuity starting date" within a period during which a limitation under Regulations Section 1.436-1(d) applied to the Plan will not be provided with the opportunity to have a new "annuity starting date" (which would constitute a new "annuity starting date" under Code Sections 415 and 417) under which the form of benefit previously elected may be modified, subject to applicable qualification requirements, once the limitations of Regulations Section 1.436-1(d) cease to apply. However, subject to the rules of Regulations Section 1.436-1(c)(3), the Plan will automatically restore benefit accruals that had been limited under Code Section 436(e) as of the "Section 436 measurement date" that the limitation ceases to apply.
- Shutdown and other unpredictable contingent event benefits. If unpredictable contingent event benefits with respect to an unpredictable contingent event that occurs during the Plan Year are not permitted to be paid after the occurrence of the event because of the limitations of Code Section 436(b) and Regulations Section 1.436-1(b), but are permitted to be paid later in the Plan Year as a result of additional contributions under Regulations Section 1.436-1(f)(2) or pursuant to the enrolled actuary's certification of the "adjusted funding target attainment percentage" for the Plan Year that meets the requirements of Regulations Section 1.436-1(g)(5)(ii)(B), then those unpredictable contingent event benefits must automatically become payable, retroactive to the period those benefits would have been payable under the terms of the Plan (other than Plan terms implementing the requirements of Code Section 436(b)). If the benefits do not become payable during the Plan Year in accordance with the preceding sentence, then the Plan is treated as if it does not provide for those benefits. However, all or any portion of those benefits can be restored pursuant to a Plan amendment that meets the requirements of Code Section 436(c) and Regulations Section 1.436-1(c) and other applicable qualification requirements.
- Treatment of Plan amendments that do not take effect. If a Plan amendment does not take effect as of the effective date of the amendment because of the limitations of Code Section 436(c) and Regulations Section 1.436-1, but is permitted to take effect later in the Plan Year as a result of additional contributions under paragraph Regulations Section 1.436-1(f)(2) or pursuant to the enrolled actuary's certification of the "adjusted funding target attainment percentage" for the Plan Year that meets the requirements of paragraph Regulations Section 1.436-1(g)(5)(ii)(C), then the Plan amendment must automatically take effect as of the first day of the Plan Year (or, if later, the original effective date of the amendment). If the Plan amendment cannot take effect during the Plan Year, then it must be treated as if it were never adopted, unless the Plan amendment provides otherwise.
- (i) **Definitions.** Defined terms shall have the meaning set forth below and as contained in Regulations Section 1.436-1(j) and shall be interpreted consistent with said Regulations.
  - (1) The term "adjusted funding target attainment percentage" means the "funding target attainment percentage" per paragraph (A) below, and increasing each of the amounts under subparagraphs (A) and (B) of Code Section 430(d)(2) by the aggregate amount of purchases of annuities for employees other than highly compensated employees (as defined in Code Section 414(q)) which were made by the Plan during the preceding two (2) Plan Years.
    - (A) The term "funding target attainment percentage" has the same meaning given such term by Code Section 430(d)(2) and the Regulations thereunder, except as otherwise provided herein. However, in the case of Plan Years beginning in 2008, the "funding target attainment percentage" for the preceding Plan Year may be determined using such methods of estimation as the Secretary may provide.
    - (B) Application to plans which are fully funded without regard to reductions for funding balances.
      - (1) In general. In the case of a Plan for any Plan Year, if the "funding

target attainment percentage" is one hundred percent (100%) or more (determined without regard to the reduction in the value of assets under Code Section 430(f)(4)), the "funding target attainment percentage" for purposes of paragraphs (1) and (1)(A) above shall be determined without regard to such reduction.

(2) Transition rule. Subparagraph (B)(1) shall be applied to Plan Years beginning after 2007 and before 2011 by substituting for "one hundred percent (100%)" the applicable percentage determined in accordance with the following table:

In the case of a Plan Year The applicable percentage is: beginning in calendar year:

2008	92%
2009	94%
2010	96%

- (3) Subparagraph (B)(2) shall not apply with respect to the current Plan Year unless the "funding target attainment percentage" (determined without regard to the reduction in the value of assets under Code Section 430(f)(4)) of the Plan for each preceding Plan Year beginning after 2007 and before the current Plan Year was not less than the applicable percentage with respect to such preceding Plan Year determined under subparagraph (B)(2).
- (2) Section 436 measurement date. A "Section 436 measurement date" is the date that is used to determine when the limitations of Code Sections 436(d) and 436(e) apply or cease to apply, and is also used for calculations with respect to applying the limitations of sections (b) and (c) of this Section 3.7.1.
- (3) Annuity starting date. The term "annuity starting date" means the annuity starting date as defined in Regulations Section 1.436-1(j)(2).
- (4) Unpredictable contingent event benefit. The term "unpredictable contingent event benefit" means an unpredictable contingent event as defined in Regulations Section 1.436-1(j)(9).
- 3.7.2 **2010 and Later Plan Years**. The provisions of this Section 3.7.2 shall apply to Plan Years beginning on or after January 1, 2010.

<u>Part I. Limitations Applicable If the Plan's Adjusted Funding Target Attainment Percentage Is Less Than 80 Percent or If the Plan Sponsor Is In Bankruptcy</u>

Section 1. Limitations Applicable If the Plan's Adjusted Funding Target Attainment Percentage Is Less Than 80 Percent, But Not Less Than 60 Percent.

Notwithstanding any other provisions of the Plan, if the Plan's adjusted funding target attainment percentage for a Plan Year is less than 80 percent (or would be less than 80 percent to the extent described in Section 1 (b) below) but is not less than 60 percent, then the limitations set forth in this Section 1 apply.

(a) 50 Percent Limitation on Single Sum Payments, Other Accelerated Forms of Distribution, and Other Prohibited Payments. A Member or beneficiary is not permitted to elect, and the Plan shall not pay, a single sum payment or other optional form of benefit that includes a prohibited payment with an annuity starting date on or after the applicable section 436 measurement date, and the Plan shall not make any payment for the purchase of an irrevocable commitment from an insurer to pay benefits or any other

payment or transfer that is a prohibited payment, unless the present value of the portion of the benefit that is being paid in a prohibited payment does not exceed the lesser of:

(i) 50 percent of the present value of the benefit payable in the optional form of benefit that includes the prohibited payment; or

(ii) 100 percent of the PBGC maximum benefit guarantee amount (as defined in § 1.436-1 (d)(3)(iii)(C) of the Treasury Regulations).

The limitation set forth in this Section 1(a) does not apply to any payment of a benefit which under § 411 (a)(11) of the Internal Revenue Code may be immediately distributed without the consent of the Member.

If an optional form of benefit that is otherwise available under the terms of the Plan is not available to a Member or beneficiary as of the annuity starting date because of the application of the requirements of this Section 1(a), the Member or beneficiary is permitted to elect to bifurcate the benefit into unrestricted and restricted portions (as described in § 1.436-1 (d)(3)(iii)(D) of the Treasury Regulations). The Member or beneficiary may also elect any other optional form of benefit otherwise available under the Plan at that annuity starting date that would satisfy the 50 percent/PBGC maximum benefit guarantee amount limitation described in this Section 1(a), or may elect to defer the benefit in accordance with any general right to defer commencement of benefits under the Plan. During a period when this Section 1(a) applies to the Plan, Members and beneficiaries are permitted to elect payment in any optional form of benefit otherwise available under the Plan that provides for the current payment of the unrestricted portion of the benefit (as described in § 1.436-1 (d)(3)(iii)(D) of the Treasury Regulations), with a delayed commencement for the restricted portion of the benefit (subject to other applicable qualification requirements, such as §§ 411(a)(11) and 401(a)(9) of the Internal Revenue Code).

(b) Plan Amendments Increasing Liability for Benefits. No amendment to the Plan that has the effect of increasing liabilities of the Plan by reason of increases in benefits, establishment of new benefits, changing the rate of benefit accrual, or changing the rate at which benefits become nonforfeitable shall take effect in a Plan Year if the adjusted funding target attainment percentage for the Plan Year is:

(i) Less than 80 percent; or

(ii) 80 percent or more, but would be less than 80 percent if the benefits attributable to the amendment were taken into account in determining the adjusted funding target attainment percentage.

The limitation set forth in this Section 1(b) does not apply to any amendment to the Plan that provides a benefit increase under a Plan formula that is not based on compensation, provided that the rate of such increase does not exceed the contemporaneous rate of increase in the average wages of Members covered by the amendment.

Section 2. Limitations Applicable If the Plan's Adjusted Funding Target Attainment Percentage Is Less Than 60 Percent.

Notwithstanding any other provisions of the Plan, if the Plan's adjusted funding target attainment percentage for a Plan Year is less than 60 percent (or would be less than 60 percent to the extent described in Section 2(b) below), then the limitations in this Section 2 apply.

(a) Single Sums, Other Accelerated Forms of Distribution, and Other Prohibited
Payments Not Permitted. A Member or beneficiary is not permitted to elect, and the Plan

shall not pay, a single sum payment or other optional form of benefit that includes a prohibited payment with an annuity starting date on or after the applicable section 436 measurement date, and the Plan shall not make any payment for the purchase of an irrevocable commitment from an insurer to pay benefits or any other payment or transfer that is a prohibited payment. The limitation set forth in this Section 2(a) does not apply to any payment of a benefit which under § 411(a)(11) of the Internal Revenue Code may be immediately distributed without the consent of the Member.

(b) Shutdown Benefits and Other Unpredictable Contingent Event Benefits Not Permitted to Be Paid. An unpredictable contingent event benefit with respect to an unpredictable contingent event occurring during a Plan Year shall not be paid if the adjusted funding target attainment percentage for the Plan Year is:

## (i) Less than 60 percent; or

(ii) 60 percent or more, but would be less than 60 percent if the adjusted funding target attainment percentage were redetermined applying an actuarial assumption that the likelihood of occurrence of the unpredictable contingent event during the Plan Year is 100 percent.

(c) Benefit Accruals Frozen. Benefit accruals under the Plan shall cease as of the applicable section 436 measurement date. In addition, if the Plan is required to cease benefit accruals under this Section 2(c), then the Plan is not permitted to be amended in a manner that would increase the liabilities of the Plan by reason of an increase in benefits or establishment of new benefits.

Section 3. Limitations Applicable If the Plan Sponsor Is In Bankruptcy. Notwithstanding any other provisions of the Plan, a Member or beneficiary is not permitted to elect, and the Plan shall not pay, a single sum payment or other optional form of benefit that includes a prohibited payment with an annuity starting date that occurs during any period in which the Plan sponsor is a debtor in a case under title 11, United States Code, or similar Federal or State law, except for payments made within a Plan Year with an annuity starting date that occurs on or after the date on which the Plan's enrolled actuary certifies that the Plan's adjusted funding target attainment percentage for that Plan Year is not less than 100 percent. In addition, during such period in which the Plan sponsor is a debtor, the Plan shall not make any payment for the purchase of an irrevocable commitment from an insurer to pay benefits or any other payment or transfer that is a prohibited payment, except for payments that occur on a date within a Plan Year that is on or after the date on which the Plan's enrolled actuary certifies that the Plan's adjusted funding target attainment percentage for that Plan Year is not less than 100 percent. The limitation set forth in this Section 3 does not apply to any payment of a benefit which under § 411(a)(11) of the Internal Revenue Code may be immediately distributed without the consent of the Member.

Section 4. **Provisions Applicable After Limitations Cease to Apply**. Subject to provisions of the Plan document freezing the Plan:

(a) Resumption of Prohibited Payments. If a limitation on prohibited payments under Section 1(a), Section 2(a), or Section 3 above applied to the Plan as of a section 436 measurement date, but that limit no longer applies to the Plan as of a later section 436 measurement date, then that limitation does not apply to benefits with annuity starting dates that are on or after that later section 436 measurement date.

(b) Resumption of Benefit Accruals. If a limitation on benefit accruals under Section 2(c) applied to the Plan as of a section 436 measurement date, but that limitation no longer applies to the Plan as of a later section 436 measurement date, then benefit accruals shall resume prospectively and that limitation does not apply to benefit accruals that are based on service on or after that later section 436 measurement date, except as otherwise

provided under the Plan. The Plan shall comply with the rules relating to partial years of participation and the prohibition on double proration under Department of Labor regulation 29 CFR §2530.204-2(c) and (d). In addition, benefit accruals that were not permitted to accrue because of the application of Section 2(c) shall be restored when that limitation ceases to apply if the continuous period of the limitation was 12 months or less and the Plan's enrolled actuary certifies that the adjusted funding target attainment percentage for the Plan Year would not be less than 60 percent taking into account any restored benefit accruals for the prior Plan Year.

(c) Shutdown and Other Unpredictable Contingent Event Benefits. If an unpredictable contingent event benefit with respect to an unpredictable contingent event that occurs during the Plan Year is not permitted to be paid after the occurrence of the event because of the limitation of Section 2(b), but is permitted to be paid later in the same Plan Year (as a result of additional contributions or pursuant to the enrolled actuary's certification of the adjusted funding target attainment percentage for the Plan Year that meets the requirements of §1.436-1 (g)(5)(ii)(B) of the Treasury Regulations), then that unpredictable contingent event benefit shall be paid, retroactive to the period that benefit would have been payable under the terms of the Plan (determined without regard to Section 2(b)). If the unpredictable contingent event benefit does not become payable during the Plan Year in accordance with the preceding sentence, then the Plan is treated as if it does not provide for that benefit.

(d) Treatment of Plan Amendments That Do Not Take Effect. If a Plan amendment does not take effect as of the effective date of the amendment because of the limitation of Section 1(b) or Section 2(c), but is permitted to take effect later in the same Plan Year (as a result of additional contributions or pursuant to the enrolled actuary's certification of the adjusted funding target attainment percentage for the Plan Year that meets the requirements of § 1.436-1(g)(5)(ii)(C) of the Treasury Regulations), then the Plan amendment must automatically take effect as of the first day of the Plan Year (or, if later, the original effective date of the amendment). If the Plan amendment cannot take effect during the same Plan Year, then it shall be treated as if it were never adopted, unless the Plan amendment provides otherwise.

Section 5. **Notice Requirement**. See section 101(j) of ERISA and Notice 2012-46 and any related IRS published guidance for rules requiring the plan administrator of a single employer defined benefit pension plan to provide a written notice to Members and beneficiaries within 30 days after certain specified dates if the plan has become subject to a limitation described in Section 1 (a), Section 2, or Section 3.

Section 6. **Methods to Avoid or Terminate Benefit Limitations**. See § 436(b)(2), (c)(2), (e)(2), and (f) of the Internal Revenue Code and §1.436-1(f) of the Treasury Regulations for rules relating to employer contributions and other methods to avoid or terminate the application of the limitations set forth in Sections 1 through 3 for a Plan Year. In general, the methods a Plan sponsor may use to avoid or terminate one or more of the benefit limitations under Sections 1 through 3 for a Plan Year include employer contributions and elections to increase the amount of Plan assets which are taken into account in determining the adjusted funding target attainment percentage, making an employer contribution that is specifically designated as a current year contribution that is made to avoid or terminate application of certain of the benefit limitations, or providing security to the Plan.

## Section 7. Special Rules.

(a) Rules of Operation for Periods Prior to and After Certification of Plan's Adjusted Funding Target Attainment Percentage

(i) In General. Section 436(h) of the Internal Revenue Code and §1.436-1(h) of the Treasury Regulations set forth a series of presumptions that apply

- (1) before the Plan's enrolled actuary issues a certification of the Plan's adjusted funding target attainment percentage for the Plan Year and
- (2) if the Plan's enrolled actuary does not issue a certification of the Plan's adjusted funding target attainment percentage for the Plan Year before the first day of the 10<sup>th</sup> month of the Plan Year (or if the Plan's enrolled actuary issues a range certification for the Plan Year pursuant to §1.436-1 (h)(4)(ii) of the Treasury Regulations but does not issue a certification of the specific adjusted funding target attainment percentage for the Plan by the last day of the Plan Year).

For any period during which a presumption under § 436(h) of the Internal Revenue Code and § 1.436-1 (h) of the Treasury Regulations applies to the Plan, the limitations under Sections 1 through 3 are applied to the Plan as if the adjusted funding target attainment percentage for the Plan Year were the presumed adjusted funding target attainment percentage determined under the rules of § 436(h) of the Internal Revenue Code and § 1.436-1(h)(1), (2), or (3) of the Treasury Regulations. These presumptions are set forth in Sections 7(a)(ii) though (iv).

- (ii) Presumption of Continued Underfunding Beginning First Day of Plan Year. If a limitation under Section 1, 2, or 3 applied to the Plan on the last day of the preceding Plan Year, then, commencing on the first day of the current Plan Year and continuing until the Plan's enrolled actuary issues a certification of the adjusted funding target attainment percentage for the Plan for the current Plan Year, or, if earlier, the date Section 7(a)(iii) or Section 7(a)(iv) applies to the Plan:
  - (1) The adjusted funding target attainment percentage of the Plan for the current Plan Year is presumed to be the adjusted funding target attainment percentage in effect on the last day of the preceding Plan Year: and
  - (2) The first day of the current Plan Year is a section 436 measurement date.
- (iii) Presumption of Underfunding Beginning First Day of 4<sup>th</sup> Month. If the Plan's enrolled actuary has not issued a certification of the adjusted funding target attainment percentage for the Plan Year before the first day of the 4<sup>th</sup> month of the Plan Year and the Plan's adjusted funding target attainment percentage for the preceding Plan Year was either at least 60 percent but less than 70 percent or at least 80 percent but less than 90 percent, or is described in §1.436-1(h)(2)(ii) of the Treasury Regulations, then, commencing on the first day of the 4<sup>th</sup> month of the current Plan Year and continuing until the Plan's enrolled actuary issues a certification of the adjusted funding target attainment percentage for the Plan for the current Plan Year, or, if earlier, the date Section 7(a)(iv) applies to the Plan:
  - (1) The adjusted funding target attainment percentage of the Plan for the current Plan Year is presumed to be the Plan's adjusted funding target attainment percentage for the preceding Plan Year reduced by 10 percentage points; and
  - (2) The first day of the 4<sup>th</sup> month of the current Plan Year is a section

### 436 measurement date.

- (iv) Presumption of Underfunding On and After First Day of 10<sup>th</sup> Month. If the Plan's enrolled actuary has not issued a certification of the adjusted funding target attainment percentage for the Plan Year before the first day of the 10<sup>th</sup> month of the Plan Year (or if the Plan's enrolled actuary has issued a range certification for the Plan Year pursuant to §1.436-1(h)(4)(ii) of the Treasury Regulations but has not issued a certification of the specific adjusted funding target attainment percentage for the Plan by the last day of the Plan Year), then, commencing on the first day of the 10<sup>th</sup> month of the current Plan Year and continuing through the end of the Plan Year:
  - (1) The adjusted funding target attainment percentage of the Plan for the current Plan Year is presumed to be less than 60 percent; and
  - (2) The first day of the 10<sup>th</sup> month of the current Plan Year is a section 436 measurement date.
- (b) New Plans, Plan Termination, Certain Frozen Plans, and Other Special Rules.
  - (i) First 5 Plan Years. The limitations in Section 1 (b), Section 2(b), and Section 2(c) do not apply to a new plan for the first 5 Plan Years of the plan, determined under the rules of §436(i) of the Internal Revenue Code and §1.436-1(a)(3)(i) of the Treasury Regulations.
  - (ii) Plan Termination. The limitations on prohibited payments in Section 1 (a), Section 2(a), and Section 3 do not apply to prohibited payments that are made to carry out the termination of the Plan in accordance with applicable law. Any other limitations under this section of the Plan do not cease to apply as a result of termination of the Plan.
  - (iii) Exception to Limitations on Prohibited Payments Under Certain Frozen Plans. The limitations on prohibited payments set forth in Sections 1(a), 2(a), and 3 do not apply for a Plan Year if the terms of the Plan, as in effect for the period beginning on September 1, 2005, and continuing through the end of the Plan Year, provide for no benefit accruals with respect to any Members. This Section 7(b)(iii) shall cease to apply as of the date any benefits accrue under the Plan or the date on which a Plan amendment that increases benefits takes effect.
  - (iv) Special Rules Relating to Unpredictable Contingent Event Benefits and Plan Amendments Increasing Benefit Liability. During any period in which none of the presumptions under Section 7(a) apply to the Plan and the Plan's enrolled actuary has not yet issued a certification of the Plan's adjusted funding target attainment percentage for the Plan Year, the limitations under Section 1(b) and Section 2(b) shall be based on the inclusive presumed adjusted funding target attainment percentage for the Plan, calculated in accordance with the rules of §1.436-1(g)(2)(iii) of the Treasury Regulations.

# (c) Special Rules Under PRA 2010.

(i) Payments Under Social Security Leveling Options. For purposes of determining whether the limitations under Section 1(a) or 2(a) apply to payments under a social security leveling option, within the meaning of §436(j)(3)(C)(i) of the Internal Revenue Code, the adjusted funding target attainment percentage for a Plan Year shall be determined in accordance with the "Special Rule for Certain Years" under § 436(j)(3) of the Internal Revenue

Code and any Treasury Regulations or other published guidance thereunder issued by the Internal Revenue Service.

(ii) Limitation on Benefit Accruals. For purposes of determining whether the accrual limitation under Section 2(c) applies to the Plan, the adjusted funding target attainment percentage for a Plan Year shall be determined in accordance with the "Special Rule for Certain Years" under § 436(j)(3) of the Internal Revenue Code (except as provided under section 203(b) of the Preservation of Access to Care for Medicare Beneficiaries and Pension Relief Act of 2010, if applicable).

(d) Interpretation of Provisions. The limitations imposed by this section of the Plan shall be interpreted and administered in accordance with §436 of the Internal Revenue Code and §1.436-1 of the Treasury Regulations.

## Section 8. **Definitions**.

The definitions in the following Treasury Regulations apply for purposes of Sections 1 through 7: § 1.436-1(j)(1) defining adjusted funding target attainment percentage; § 1.436-(j)(2) defining annuity starting date; § 1.436-1(j)(6) defining prohibited payment; § 1.436-1 (j)(8) defining section 436 measurement date; and § 1.436-(j)(9) defining an unpredictable contingent event and an unpredictable contingent event benefit.

Part II. Multiple Employer Plan Rules. If the Plan is a multiple employer plan, the following provisions apply depending on whether § 413(c)(4)(A) of the Internal Revenue Code applies to the Plan.

- 1. If this is a multiple employer plan to which § 413(c)(4)(A) of the Internal Revenue Code applies, including if the election described in §413(c)(4)(B) has been made, then the rules in Sections 1 through 8 of Part I above apply separately to each employer under the Plan, as if each such employer maintained a separate plan; and
- 2. If this is a multiple employer plan to which §413(c)(4)(A) of the Internal Revenue Code does not apply, then the rules in Sections 1 through 8 of Part I above apply as if all Members in the Plan are employed by a single employer.

### SECTION 4

### Valuation, Adjustment and Investment of Accounts

4.1 Annual Valuation and Adjustment of Accounts. The Trustee shall value the Fund (or, if applicable, each subfund established pursuant to Section 4.4) as of each Annual Valuation Date, which valuation shall reflect, as nearly as possible, the then fair market value of the assets comprising such fund or subfund (including income accumulations therein). In making such valuations, the Trustee may rely upon information supplied by an Investment Manager having investment responsibility over assets of the Fund or a particular subfund. The Trustee shall then adjust each Account (including undistributed Matured Accounts) or portion thereof which is invested in the Fund or a particular subfund for its proportionate share in any increase or decrease in the value of such Fund or subfund as so determined. Employee contributions, nNondeductible voluntary contributions, deductible contributions, rollover contributions and transfers from qualified retirement plans, if any, made by or on behalf of any Participant for the Plan Year shall then be posted to his Employee proper Account(s) and credited to the Fund (or the appropriate subfund or subfunds or otherwise in accordance with the investment elections then in effect for such Participant); the Employer contributions and Forfeitures, if any, for the Plan Year shall then be posted to the Employer Accounts of Participants entitled thereto and credited to the Fund (or the appropriate subfund or subfunds or otherwise in accordance with the investment elections then in effect for such Participant); and Regular Interest shall then be posted to the Employee and Employeerproper Accounts and credited to the Fund (or the appropriate subfund or subfunds or otherwise in accordance with the investment elections then in effect for such Participant).

4.2 <u>Intermediate Valuations and Adjustments</u>. If a Participant's Matured Account is to be distributed as of a date which is not the Annual Valuation Date, the Employer on a uniform, nondiscriminatory and consistent basis may direct the Trustee to make an intermediate valuation and adjustment for such Account as of any date designated by the Trustee in a nondiscriminatory manner that is coincident with, preceding or following the date as of which such distribution is to occur. For investment purposes, unless elected otherwise by the Employer in the Adoption Agreement, the Trustee may make such intermediate valuation and adjustment of Participant Accounts as designated by the Trustee in a nondiscriminatory manner.

### 4.3 <u>Management and Investment of Fund.</u>

(a) <u>Police Plan</u>. The Fund shall be invested under the general direction of the Retirement Committee. The City, (or the Retirement Committee if delegated such function by the City) shall select and contract with a Funding Agent to hold or invest the assets of the Plan and to provide the benefits hereunder. The City or Committee may select and contract with Investment Managers registered under the Investment Advisers Act of 1940 to invest, reinvest, and otherwise manage such portion of the assets of the Plan as may be assigned by the City or Committee. The Fund shall be invested pursuant to the policies established by the Nebraska Investment Council.

If Participant investment direction is permitted in the Adoption Agreement, the City shall establish separate investment accounts for each Employee for the purpose of allowing each Employee to direct the investment of all or a portion of his or her Employee Account and/or Employer Account, subject to such rules and limitations as imposed by law, the City or the Retirement Committee.

(b) <u>Fire Plan</u>. The Fund shall be invested by the Retirement Committee. The City, subject to the approval of the Retirement Committee, shall contract with a Funding Agent to hold or invest the assets of the Plan and to provide the benefits hereunder. The Retirement Committee, subject to the approval of the City, may select an Investment Manager. The City, subject to the approval of the Retirement Committee, may contract with Investment Managers registered under the Investment Advisers Act of 1940 to invest, reinvest, and otherwise manage such portion of the assets of the Plan as may be assigned by the City or Retirement Committee.

The Retirement Committee shall establish an investment plan which allows each Employee of the Plan to allocate all contributions to his or her Employee Account and, if he or she commenced his or her employment after January 1, 1984, his or her Employer Account, to the various investment options or combinations of investment options described in the investment plan. Each Employee shall have the option of investing his or her Employee Account and, if he or she commenced his or her employment after January 1, 1984, his or her Employer Account, in any proportion, including full allocation, in any investment option offered by the Plan. Upon the direction of the City by election in the Adoption Agreement, Employees employed on January 1, 1984, may have the option to allocate their Employer Account to various investment options or combinations of investment options in any proportion, including full allocation, in any investment option offered by the Plan. Each Employee shall be given a summary of the investment plan and a detailed current description of each investment option prior to making or revising his or her allocation.

4.4 All Plans. Subject to any permitted Participant investment direction or other provisions of the Plan, tThe Fund in the hands of the Trustee, together with all additional contributions made thereto and together with income thereof, which shall be accumulated as hereinafter provided, shall be controlled, managed, invested, reinvested and ultimately paid and distributed to Participants by the Trustee with all the powers, rights and discretions generally possessed by Trustees, and with all the additional powers, rights and discretions hereinafter conferred upon the Trustee. To the extent permitted by the Employer, the Fund may be divided into two or more subfunds for the purpose of investment. In addition, the Trustee may invest in any form of investment authorized by Section 10, provided, however, that no portion of a Participant's Voluntary Deductible Account shall be invested in Insurance Contracts. All amounts shall be credited with Regular Interest after being contributed to the Plan.

4.5 <u>Individual Subfunds</u>. A subfund or subfunds shall be created consisting of the Accounts of Participants who are permitted to direct the investment of their Accounts, as provided in Sections 4.3 and 10.11 hereof.

### **SECTION 5**

### Vesting

# 5.1 <u>Employer Accounts</u>.

- 5.1.1 <u>Progressive Vesting.</u> The Employer Account of each Participant shall become Vested in him in accordance with the Schedule in Section I set forth in the Adoption Agreement; provided, however, that the Vested percentage of a Participant's Employer Account determined as of the Supplemental Effective Date (or the date of actual, formal adoption of this Agreement by the Employer, if later) shall be not less than such Vested percentage computed under the Prior Plan Statement, if any, as of that date.
- 5.1.2 <u>Full Vesting</u>. Notwithstanding any of the foregoing provisions for progressive vesting of Employer Accounts of Participants, the entire Employer Account of each Participant shall be fully vested in him upon the earliest occurrence of any of the following events while in the employment of the Employer:
  - (a) His death,
  - (b) His attainment of his Normal Retirement Date or his attainment of any earlier age specified in the Adoption Agreement,
    - (c) His retirement on account of his Disability,
  - (d) A complete termination of the Plan or a complete discontinuance of Employer contributions hereto,
    - (e) A partial termination of the Plan which is effective as to him, or
    - (f) As provided in relevant state statute governing Police or Fire Plans.
  - 5.1.3 <u>Distribution of Vested Account Upon Termination of Employment.</u>
  - (a) Cash Out Distributions. Notwithstanding anything in this Plan to the contrary:
    - (i) <u>Fire Plan</u>. If an Employee terminates service before his or her Retirement Date, and the value of the Employee's Vested Retirement Value is less than \$3,500, such Employee shall, upon request within one year of such termination, be paid his or her Vested Retirement Value in the form of a single lump sum payment.
    - (ii) All Other Plans (including Police Plans and excluding Fire Plans). Effective for distributions on or after March 28, 2005, For all Plans, effective January 1, 2014, and except as otherwise specified by the Employer in the Adoption Agreement, if an Employee terminates service, and the value of the Employee's Vested Retirement Value derived from Employer and Employee contributions is not greater than \$1,000, the Employee will automatically receive a distribution of the value of the entire Vested portion of such Account balance if the Employee does not elect to transfer or roll over such distribution to an eligible retirement plan or to receive it directly, and the nonvested portion will be treated as a Forfeiture. For purposes of this Section, if the value of an Employee's Vested account balance is zero, the Employee shall be deemed to have received a distribution of such Vested account balance. A Participant's Vested account balance shall not include accumulated deductible Employee contributions within the meaning of Code section 72(o)(5)(B) for Plan

Years beginning prior to January 1, 1989. For distributions before March 28, 2005, "not greater than \$5,000" ("less than \$3,500" for Police Plans) was substituted for "not greater than \$1,000" in this paragraph.

Before January 1, 2014: If an Employee participating in a Fire Plan terminates service before his or her Retirement Date, and the value of the Employee's Vested Retirement Value is less than \$3,500, such Employee shall, upon request within one year of such termination, be paid his or her Vested Retirement Value in the form of a single lump-sum payment. For all other plans (including Police Plans and excluding Fire Plans), effective for distributions on or after March 28, 2005, and except as otherwise specified by the Employer in the Adoption Agreement, if an Employee terminates service, and the value of the Employee's Vested Retirement Value derived from Employer and Employee contributions is not greater than \$1,000, the Employee will automatically receive a distribution of the value of the entire Vested portion of such Account balance if the Employee does not elect to transfer or roll over such distribution to an eligible retirement plan or to receive it directly, and the nonvested portion will be treated as a Forfeiture. For purposes of this Section, if the value of an Employee's Vested account balance is zero, the Employee shall be deemed to have received a distribution of such Vested account balance. A Participant's Vested account balance shall not include accumulated deductible Employee contributions within the meaning of Code section 72(o)(5)(B) for Plan Years beginning prior to January 1, 1989. For distributions before March 28, 2005, "not greater than \$5,000" ("less than \$3,500" for Police Plans) was substituted for "not greater than \$1,000" in this paragraph.

If an Employee terminates service and elects (in accordance with the requirements of Section 7) to receive the value of the Employee's Vested Account balance, the nonvested portion will be treated as a Forfeiture. If the Employee elects to have distributed less than the entire Vested portion of the Account balance derived from Employer contributions, the part of the nonvested portion of such Account that will be treated as a Forfeiture is the total nonvested portion multiplied by a fraction, the numerator of which is the amount of the distribution attributable to Employer contributions and the denominator of which is the total value of the Vested Employer-derived Account balance.

For distributions made under this subsection after December 31, 2001 and before March 28, 2005, the value of an Employee's nonforfeitable Account balance shall be determined without regard to that portion of the Account balance that is attributable to rollover contributions (and earnings allocable thereto) within the meaning of Sections 402(c), 403(a)(4), 403(b)(8), 408(d)(3)(A)(ii), and 457(e)(16) of the Code. For distributions on or after March 28, 2005, the \$1,000 threshold (or other threshold specified in the Adoption Agreement) is determined including any such rollover contributions (and earnings thereon) within the meaning of the specified Code sections. If a greater threshold than \$1,000 is established in the Adoption Agreement, any distribution greater than \$1,000 that is made without the Participant's consent shall be automatically rolled over to an individual retirement plan pursuant to Code section 401(a)(31)(B).

### (b) Non Cash-Out Distributions

(i) Fire Plans. If a Participant in a Fire Plan terminates employment before his or her Retirement Date, the Participant may request and receive, as a lump-sum payment the Retirement Value of his or her Employee Account as determined at the Valuation Date preceding his or her termination of employment. Such Participant, if vested, may in lieu thereof, receive a deferred pension benefit or lump-sum benefit in an amount purchased or provided by the Vested Retirement Value at the date of retirement. The <a href="Vested">Vested</a> Retirement Value at such Retirement Date shall consist of the then accumulated value of the Employee's Account at the date of retirement as reduced by any lump-sum distributions received prior to retirement, together with a vested percentage of the accumulated value of the Participant's Employer Account at the date of retirement.

- (ii) All Other Plans (including Police Plans and excluding Fire Plans). If a Participant terminates service with the City prior to his or her Normal or Early Retirement Date, the Participant may request and receive a lump-sum payment of the Retirement Value of his or her Employee Account as determined as of the Valuation Date preceding his or her termination of employment. The Participant, if Vested, shall also receive a deferred pension benefit in an amount purchased or provided by the <a href="Vested">Vested</a> Retirement Value at the date of retirement. The <a href=Vested</a> Retirement Value at such retirement date shall consist of the accumulated value of the Participant's Employee Account, reduced by any lump-sum distribution prior to retirement, plus the vested portion of the accumulated value of the Participant's Employer Account on the date of Retirement.
- (iii) Effective January 1, 1997 April 16, 2012, a Participant may elect upon his or her termination of employment to receive his or her Vested Retirement Value in the form of a single lump sum payment, notwithstanding any prior Plan provision to the contrary limiting lump sum distributions to Participants who have a Retirement Date on or after January 1, 1997.

Before April 16, 2012, the Plan provided: Effective January 1, 1997, a Participant may elect upon his or her termination of employment to receive his or her Vested Retirement Value in the form of a single lump sum payment. For a Participant whose termination of employment was prior to January 1, 1997, this election shall be available only if the City had adopted a lump-sum distribution option for terminating Participants in the funding medium established for the retirement system.

Upon any lump sum payment of a terminating Participant's Retirement Value, such Participant will not be entitled to any deferred pension benefit and the City and the retirement systemPlan shall have no further obligations to pay such Participant or his or her Beneficiaries any benefits.

- (iv) If the lump-sum payment is not requested upon termination prior to the Retirement Date, the Participant shall receive a deferred vested benefit or lump-sum benefit in an amount purchased or provided by the Vested Retirement Value of the Employee and Employer Accounts on the retirement date; provided, however, if the Participant would have received a distribution under this Section but for the fact that the Participant's Vested Account Balance exceeded \$1,000, when the Participant terminated service and if, at a later time, such account balance is reduced so it does not exceed \$1,000, the Administration Committee may distribute such Account Balance and the non-vested portion will be treated as a forfeiture.
- (c) If an Employee receives a distribution pursuant to this Section 5.1.3 and the Employee resumes employment covered under the Plan, the Employee's Employer-derived Account Balance will be restored, if permitted in the Adoption Agreement, to the amount on the date of the distribution, if the Employee repays to the Plan the full amount of the distribution attributable to Employer contributions before the earlier of:
  - (i) 5 years after the first date on which the Participant is subsequently reemployed by the Employer; or
  - (ii) the date on which the Participant incurs five consecutive one-year Breaks in Service following the date of the distribution. This latter condition also applies if the Participant makes repayment within the Plan Year in which he incurs five consecutive one-year Breaks in Service which would result in a complete Forfeiture of the amount otherwise subject to restoration.

If elected by the Employer in the Adoption Agreement, if an Employee is deemed to receive a distribution pursuant to this Section 5.1.3, and the Employee resumes employment covered under this Plan before the date the Participant incurs five consecutive one-year Breaks in Service, upon the reemployment of such Employee, the Employer-derived Account Balance of the Employee will be restored to the amount on the date of such deemed distribution.

A Participant's Employer-derived Account Balance shall not be restored if:

- (i) the Participant's Employer-derived Account Balance was 100% nonforfeitable at the time of the cash-out distribution; or
  - (ii) the Participant incurred five consecutive One-Year Breaks in Service.

A Participant may not receive a cash-out distribution under this Section 5.1.3 if, prior to the time the Trustee actually makes the cash-out distribution, the Participant returns to employment with the Employer.

- (d) In the case of any plan other than a Fire Plan, iIf the value of the Participant's Account Balance derived from Employer and Employee contributions (other than accumulated Voluntary Deductible Employee contributions) exceeds \$1,000 (before March 28, 2005, "exceeds \$1,000" was "is less than \$3,500" for Police Plans, and "is less than \$5,000" for plans other than a Police Plan), the Participant must consent to any distribution from such Account Balance unless otherwise provided in the Adoption Agreement. Participant consent is required for distributions from Fire Plans regardless of amount.
- (e) Amounts forfeited or treated as Forfeitures shall be reallocated as provided in Section 6.4.
- (f) Participant Distribution Notification. To the extent the particular requirement is applicable to the Plan:
  - (i) 180-Day Notice Period. Effective for notices in Plan Years beginning after December 31, 2006, the 90-day maximum notice period of Code Sections 402(f) (rollover notice), 411(a)(11) (participant's consent to distribution), and 417 (notice regarding the joint and survivor and qualified optional survivor annuity and qualified preretirement annuity rules) shall be increased to 180 days, and any reference to the 90-day maximum notice period shall be deemed changed to 180 days.
  - (ii) Effect of Delayed Distribution. Notices given to Participants pursuant to Code Section 411(a)(11), if applicable, in Plan Years beginning after December 31, 2006 shall include a description of the consequences of failing to defer a distribution, and
  - (iii) Relative Values. Notices to Participants shall include the relative values of the various optional forms of benefit under the Plan satisfying the requirements of Internal Revenue Code Section 417(a)(3), to the extent applicable to the Plan. This provision, if applicable to the Plan, is effective as of the applicable effective date set forth in Treasury Regulations (i.e., to qualified pre-retirement survivor annuity explanations provided on or after July 1, 2004; to qualified joint and survivor annuity explanations with respect to any distribution with an annuity starting date that is on or after February 1, 2006, or on or after October 1, 2004 with respect to any optional form of benefit that is subject to the requirements of Code Section 417(e)(3) if the actuarial present value of that optional form is less than the actuarial present value as determined under Code Section 417(e)(3)).

Provided, however, any requirement of this subsection (f) shall not apply to the extent (i) it is inapplicable to a governmental plan defined in Code section 414(d), pursuant to the flush language of Code Section 401(a) and Code Section 411(e)(1)(B) and other applicable provisions of the Code, regulations and guidance, and (ii) this Plan is a governmental plan.

5.2. <u>Amendment to Comply with Section 415</u>. An adopting Employer may amend the Plan by adding overriding Plan language to the Adoption Agreement where such language is necessary to satisfy Sections 415, and/or 416 if applicable to the Plan, of the Code because of the required aggregation of multiple plans under these Sections. <u>Effective for any Plan Year beginning after December 31, 2001, any provisions of the Plan setting forth top-heavy provisions of Code section 416, to the extent Code section 416 is applicable to the Plan, are modified by</u>

substituting the term "separation from service" with "severance from employment." Provided, however, top-heavy provisions shall not apply if this Plan is a governmental plan within the meaning of Code section 414(d).

5.3 Other Accounts. The Employee and Matured Account of each Participant shall be fully Vested in him at all times.

## SECTION 6

### Maturity and Distributable Events

- 6.1 <u>Events of Maturity</u>. A Participant's Total Account shall mature upon the earliest occurrence of any of the following Events of Maturity while in the employment of the Employer:
  - (a) His termination of employment with the Employer.
  - (b) Termination of this Plan.

Provided, however, that with the exception of Police and Fire Plans, a transfer from Recognized Employment to employment with the Employer that is other than Recognized Employment or a transfer for the employment of one Employer Participating in this Plan to another such Employer shall not constitute an Event of Maturity, though in all cases, even involving a Police or Fire Plan, a transferring Participant shall cease participating in this Plan if as a result of the transfer he or she becomes eligible under another Plan that is qualified under Code Section 401(a).

- 6.2 <u>Determination of Matured Benefit</u>. Upon the occurrence of an Event of Maturity effective as to a Participant, the Trustee shall determine the value of his Total Account as of the Valuation Date contemporaneous with, following or next preceding the Event of Maturity, and the Vested portion thereof shall be his Matured Account.
- 6.3 Effect of Maturity Upon Further Participation in the Plan. On the occurrence of an Event of Maturity, a Participant shall cease to have any interest in the Plan or the Fund other than his right to receive distribution of his Matured Account, as provided in Sections 6 & 7 hereof; except that (a) his Matured Account or the portion thereof from time to time undistributed shall be increased or decreased, as the case may be, by any increase or decrease in the value of the Fund and by any income thereon or expense allocated thereto, and (b) he shall be entitled to share in the Employer contributions for the Plan Year in which such Event of Maturity occurred only to the extent provided in the Agreement. He shall not share in any Employer contribution for Plan Years after the Plan Year in which such Event of Maturity occurred, unless he shall continue to be or thereafter again become a Participant hereunder, as hereinbefore provided.
- 6.4 <u>Disposition of Nonvested Portion of Account</u>. If any portion of a Participant's Employer Account is not Vested in him upon the occurrence of an Event of Maturity effective as to him, such portion treated as a forfeiture under Section 5.1.3 shall be disposed of as follows:

## 6.4.1. Police and Fire Plans.

- (a) <u>Police Plan</u>. Any Forfeitures shall first be used as of the Reallocation Date to meet the expense charges incurred by the City with respect to administration of the Plan, and the remainder, if any, shall be applied to reduce City contributions which otherwise are required to fund benefits under the Plan.
- (b) <u>Fire Plan.</u> Any Forfeiture shall be deposited as of the Reallocation Date in the Unallocated Employer Account. If the actuarial evaluation of the Fund required herein shows that the assets of the Unallocated Employer Account are sufficient to provide for the projected plan liabilities, such forfeitures shall instead be used to meet the expenses incurred by the City in connections with administering the Plan, and the remainder, if any, shall be applied to reduce City contributions which otherwise would be required to fund benefits under the Plan.

- 6.4.2 <u>All Other Plans Reallocation to Other Participants.</u> As elected in the Adoption Agreement:
  - (a) Added to Employer Contribution. As of the Reallocation Date, any Forfeitures shall be added to the Employer contribution, if any, for such Plan Year and allocated as provided in Section 3.3 hereof to the Employer Accounts of those Participants who are eligible to share in the allocation of the Employer contribution, if any, as of such Reallocation Date.
  - (b) <u>Reduce Employer Contribution</u>. As of the Reallocation Date, any Forfeitures shall be used to reduce Employer contributions for the Plan Year and succeeding Plan Years until totally reallocated under Section 3.4 to other Participants.
- 6.4.3 <u>Reallocation Date</u>. As elected by the Employer in the Adoption Agreement, the Reallocation Date under Section 6.4 shall be:
  - (a) the next succeeding Valuation Date after an amount is treated as a Forfeiture, or
  - (b) the earlier of the Valuation Date of the Plan Year in which a Participant incurs 5 consecutive One-Year Breaks in Service or his Account is no longer subject to restoration; or the date the Participant receives a cash-out distribution of the nonforfeitable percentage of his or her Account as a result of termination of participation in the Plan. (Accounts held under this Section 6.4.3(b) are increased or decreased, as the case may be, by any increase or decrease in the Fund and by any income thereon.)
- 6.4.4 <u>Restoration</u>. Amounts treated as Forfeitures under Section 6.4.3(a) and Accounts held under Section 6.4.3(b) are subject to the restoration provisions of Section 5.1.3 if elected by the Employer in the Adoption Agreement.
- 6.5 <u>Other Distributable Events.</u> Distribution of some or all of the Participant's Total Account will occur:
  - (a) If elected by the Employer in the Adoption Agreement, and subject to the consent of the Participant (and the Participant's spouse) if required, after the Participant's Normal Retirement Date.
    - (b) At and after a Participant's or Beneficiary's Required Beginning Date under Section 7.
  - (c) In the case of a Plan which provides for Participant Loans, default under the terms of a Participant Loan, to the extent of the amount in default.
  - (d) Subject to the consent of the Participant, upon the Participant's attainment of early retirement date, if allowed under the terms of the Plan (in-service distribution upon attaining early retirement age or date not allowed in the case of a pension plan).

Distributions pursuant to this Section 6.5 shall not affect a right of a Participant who continues in Recognized Employment to share in Employer contributions and allocation of Forfeitures as otherwise determined under this Plan.

6.5.1 <u>Termination of Employment - Police and Fire Plans</u>. Notwithstanding any provision in this Agreement to the contrary, a deferred benefit which is payable to a Participant whose employment with the City was terminated prior to his or her death or retirement before his or her Retirement Date shall be payable on the first of the month immediately following the Participant attaining 60 years of age (55 years of age for Fire Plan Participant). The Participant has the option to commence benefit payments as of the first day of the month after the Participant attains 55 years of age and has completed 25 years of service with the City (50 years of age and 21 years of service for Fire Plan Participant). An election for early commencement of benefits shall be made by the Participant in writing prior to the payment of such

- benefits. Benefits shall be paid in a form permitted in this Agreement for payment of benefits upon retirement and properly elected by the Participant in writing prior to the commencement of such benefits. Unless otherwise provided in the Adoption Agreement, The City shall have the option to pay any Participant's Retirement Value in the form of a lump sum payment if such Vested Retirement Value is not greater than \$1,000 ("less than \$3,500" for Police Plans before March 28, 2005) on the date of his or her termination of employment (for Fire Plan Participants, the City can make a lump sum payment upon request of the Participant within one year of termination of employment if the Vested Retirement Value upon such termination is less than \$3,500). Participants also have the option of transferring the Employee Account and the vested portion of the Employer Account if the requirements of Section 3.5 are satisfied.
- 6.5.2 <u>Early Retirement Swing Employment Fire Plans</u>. Notwithstanding any provision in the Agreement to the contrary, A Fire Plan Participant may retire or be retired and receive the applicable benefit upon the attainment of age 55 while employed by the City as a firefighter, in accordance with Neb. Rev. Stat. Section 16-1028. An election for early commencement of benefits shall be made by the Participant, in writing, prior to the payment of such benefits. Benefits shall be paid in a form permitted in the Agreement for payment of benefits upon retirement and properly elected by the Participant in writing prior to the commencement of such benefits.
- (e) The City shall have the option to pay a Participant's Vested Retirement Value in the form of a single lump sum payment to the extent permitted by Section 5.1.3 if such Vested Retirement Value does not exceed \$1,000 on the date of his or her termination of employment, except as otherwise provided in the Adoption Agreement. Participants shall have the option of transferring the Employee Account and the Vested portion of the Employer Account if the requirements of Section 3.5 are satisfied.

### **SECTION 7**

### Distribution

- 7.1 <u>Time of Distribution</u>. Subject to the provisions of this Section 7 and any required Participant consent, upon the occurrence of an Event of Maturity effective as to a Participant and after the Participant's Matured Account has been determined, the Trustee will make or commence distribution of such Matured Account as follows:
  - 7.1.1 <u>Distribution from Matured Account</u>. Unless otherwise provided herein or the Participant elects otherwise, distribution of a Participant's Matured Account shall commence on the first day of the month following the date which is within 31 to 18090 days after an Event of Maturity, or within an administratively practicable time thereafter (before January 1, 2007, distributions commenced on the first day of the month following the date which is 90 days after an Event of Maturity).

# 7.2 <u>Distribution Requirements</u>.

7.2.1 <u>Distributions in calendar years beginning after December 31, 1984 and ending before January 1, 2003.</u>

- 7.2.1.1 <u>Supersession</u>. The requirements of this Section 7.2.1 shall apply to any distribution of a Participant's interest and will take precedence over any inconsistent provisions of this Plan. Unless otherwise specified, the provisions of this Section 7.2 apply to calendar years beginning after December 31, 1984 and ending before January 1, 2003.
- 7.2.1.2 Requirements. Notwithstanding any provision of the Plan to the contrary, all distributions required under this Section 7.2.1 shall be determined and made in accordance with Code section 401(a)(9) and the proposed regulations thereunder published in the Federal Register on July 27, 1987; provided, however, distributions made during the 2002 calendar year were made in accordance with the regulations that were published in the Federal Register on January 17, 2001.

- 7.2.1.3 <u>Required Beginning Date</u>. The entire interest of a Participant must be distributed or begin to be distributed no later than the Participant's required beginning date.
- 7.2.1.4 <u>Limits on Distribution Periods</u>. As of the first distribution calendar year, distributions, if not made in a single sum, may only be made over one of the following periods (or a combination thereof):
  - (a) the life of the Participant,
  - (b) the life of the Participant and a designated beneficiary,
  - (c) a period certain not extending beyond the life expectancy of the Participant, or
- (d) a period certain not extending beyond the joint and last survivor expectancy of the Participant and a designated beneficiary.
- 7.2.1.5 <u>Determination of Amount to be Distributed Each Year</u>. If the Participant's interest is to be distributed in other than a single sum, the following minimum distribution rules shall apply on or after the required beginning date:
- (a) If a Participant's benefit is to be distributed over (1) a period not extending beyond the life expectancy of the Participant or the joint life and last survivor expectancy of the Participant and the Participant's designated beneficiary or (2) a period not extending beyond the life expectancy of the designated beneficiary, the amount required to be distributed for each calendar year, beginning with distributions for the first distribution calendar year, must at least equal the quotient obtained by dividing the Participant's benefit by the applicable life expectancy.
- (b) For calendar years beginning before January 1, 1989, if the Participant's spouse is not the designated beneficiary, the method of distribution selected must assure that at least 50% of the present value of the amount available for distribution is paid within the life expectancy of the Participant.
- (c) For calendar years beginning after December 31, 1988, the amount to be distributed each year, beginning with distributions for the first distribution calendar year shall not be less than the quotient obtained by dividing the Participant's benefit by the lesser of (1) the applicable life expectancy or (2) if the Participant's spouse is not the designated beneficiary, the applicable divisor determined from the table set forth in Q&A 4 of Section 1.401(a)(9) 2 of the proposed Treasury Regulations. Distributions after the death of the Participant shall be distributed using the applicable life expectancy in Section 7.2.1.5(a) above as the relevant divisor without regard to proposed Treasure Regulation section 1.401(a)(9) 2.
- (d) The minimum distribution required for the Participant's first distribution calendar year must be made on or before the Participant's required beginning date. The minimum distribution for other calendar years, including the minimum distribution for the distribution calendar year in which the Employee's required beginning date occurs, must be made on or before December 31 of that distribution calendar year.
- (e) If the Participant's benefit is distributed in the form of an annuity purchased from an insurance company, distributions thereunder shall be made in accordance with the requirements of Code section 401(a)(9) and the proposed regulations thereunder.
- 7.2.1.6 <u>Distribution Beginning before Death</u>. If the Participant dies after distribution of his or her interest has begun, the remaining portion of such interest will continue to be distributed at least as rapidly as under the method of distribution being used prior to the Participant's death.
- 7.2.1.7 <u>Distribution Beginning after Death</u>. If the Participant dies before distribution of his or her interest begins, distribution of the Participant's entire interest shall be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death except to the extent that an election is made to receive distributions in accordance with (a) or (b) below:

- (a) if any portion of the Participant's interest is payable to a designated beneficiary, distributions may be made over the life or over a period certain not greater than the life expectancy of the designated beneficiary commencing on or before December 31 of the calendar year immediately following the calendar year in which the Participant died;
- (b) if the designated beneficiary is the Participant's surviving spouse, the date distributions are required to begin in accordance with (a) above shall not be earlier than the later of (1) December 31 of the calendar year immediately following the calendar year in which the Participant died and (2) December 31 of the calendar year in which the Participant would have attained age 70 ½.

If the Participant has not made an election pursuant to this Section 7.2.1.7 by the time of his or her death, the Participant's designated beneficiary must elect the method of distribution no later than the earlier of (1) December 31 of the calendar year in which distributions would be required to begin under this Section, or (2) December 31 of the calendar year which contains the fifth anniversary of the date of death of the Participant. If the Participant has no designated beneficiary, or if the designated beneficiary does not elect a method of distribution, distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.

For purposes of this Section 7.2.1.7, if the surviving spouse dies after the Participant, but before payments to such spouse begin, the provisions of this Section 7.2.1.7, with the exception of paragraph (b) herein, shall be applied as if the surviving spouse were the Participant.

For the purposes of Sections 7.2.1.6 and 7.2.1.7, any amount paid to a child of the Participant will be treated as if it has been paid to the surviving spouse if the amount becomes payable to the surviving spouse when the child reaches the age of majority.

For the purposes of Sections 7.2.1.6 and 7.2.1.7, distribution of a Participant's interest is considered to begin on the Participant's required beginning date (or, if this Section 7.2.1.7 is applicable, the date distribution is required to begin to the surviving spouse pursuant to this Section 7.2.1.7). If distribution in the form of an annuity irrevocably commences to the Participant before the required beginning date, the date distribution is considered to begin is the date distribution actually commences.

- 7.2.1.8 <u>Applicable Life Expectancy</u>. The life expectancy (or joint and last survivor expectancy) calculated using the attained age of the Participant (or designated beneficiary) as of the Participant's (or designated beneficiary's) birthday in the applicable calendar year reduced by one for each calendar year which has elapsed since the date life expectancy was first calculated. If life expectancy is being recalculated, the applicable life expectancy shall be the life expectancy as so recalculated. The applicable calendar year shall be the first distribution calendar year, and if life expectancy is being recalculated such succeeding calendar year.
- 7.2.1.9 <u>Designated Beneficiary</u>. The individual who is designated as the beneficiary under the Plan in accordance with Code section 401(a)(9), the proposed regulations thereunder and Section 7.5 of this Plan
- 7.2.1.10 <u>Distribution Calendar Year</u>. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year is the calendar year immediately preceding the calendar year which contains the Participant's required beginning date. For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which distributions are required to begin pursuant to Section 7.2.1.7 above.
- 7.2.1.11 <u>Life Expectancy</u>. Life expectancy and joint and last survivor expectancy are computed by use of the expected return multiples in Tables V and VI of Section 1.72 9 of the Income Tax Regulations.

Unless otherwise elected by the Participant (or spouse, in the case of distributions described in Section 7.2.1.7(b) above) by the time distributions are required to begin, life expectancies shall not be recalculated annually. Such election shall be irrevocable as to the Participant (or spouse) and shall apply to all subsequent years. The life expectancy of a nonspouse beneficiary may not be recalculated.

## 7.2.1.12 Participant's Benefit.

- (a) The Account balance as of the last Valuation Date in the calendar year immediately preceding the distribution calendar year (valuation calendar year) increased by the amount of any contributions or forfeitures allocated to the Account balance as of dates in the valuation calendar year after the valuation date and decreased by distributions made in the valuation calendar year after the valuation date.
- (b) Exception for Second Distribution Calendar Year. For purposes of paragraph (a) above, if any portion of the minimum distribution for the first distribution calendar year is made in the second distribution calendar year on or before the required beginning date, the amount of the minimum distribution made in the second distribution calendar year shall be treated as if it had been made in the immediately preceding distribution calendar year.

### 7.2.1.13. Required Beginning Date

- (a) General Rule. The required beginning date of a Participant is the later of (i) first day of April of the calendar year following the calendar year in which the Participant attains age 70 ½, and (ii) April 1st of the calendar year following the calendar year in which the Participant retires.
- 7.2.1.14 <u>Transitional Rule</u>. Notwithstanding the other requirements of this Section 7.2.1, distribution on behalf of any Employee may be made in accordance with all of the following requirements (regardless of when such distribution commences):
- (a) The distribution by the Fund is one which would not have disqualified such Fund under Code section 401(a)(9) as in effect prior to amendment by the Deficit Reduction Act of 1984.
- (b) The distribution is in accordance with a method of distribution designated by the Employee whose interest in the Fund is being distributed or, if the Employee is deceased, by a beneficiary of such Employee.
- (c) Such designation was in writing, was signed by the Employee or the beneficiary, and was made before January 1, 1984.
  - (d) The Employee had accrued a benefit under the Plan as of December 31, 1983.
- (e) The method of distribution designated by the Employee or the beneficiary specifies the time at which distribution will commence, the period over which distributions will be made, and in the case of any distribution upon the Employee's death, the beneficiaries of the Employee listed in order of priority.

A distribution upon death will not be covered by this transitional rule unless the information in the designation contains the required information described above with respect to the distributions to be made upon the death of the Employee.

For any distribution which commences before January 1, 1984, but continues after December 31, 1983, the Employee, or the beneficiary, to whom such distribution is being made, will be presumed to have designated the method of distribution under which the distribution is being made if the method of distribution was specified in writing and the distribution satisfies the requirements in subsections 7.2.1.14(a) and (e).

If a designation is revoked, any subsequent distribution must satisfy the requirements of Section 401(a)(9) of the Code and the proposed regulations thereunder. If a designation is revoked subsequent to the date distributions are required to begin, the Fund must distribute by the end of the calendar year following the calendar year in which the revocation occurs the total amount not yet distributed which would have been required to have been distributed to satisfy Code section 401(a)(9) and the proposed regulations thereunder, but for the Code section 242(b)(2) election. For calendar years beginning after December 31, 1988, such distributions must meet the minimum distribution incidental benefit requirements in section 1.401(a)(9)-2 of the proposed Treasury Regulations. Any changes in the designation will be considered to be a revocation of the designation. However, the mere substitution or addition of another beneficiary (one not named in the designation) under the designation will not be considered to

be a revocation of the designation, so long as such substitution or addition does not alter the period over which distributions are to be made under the designation, directly or indirectly (for example, by altering the relevant measuring life). In the case in which an amount is transferred or rolled over from one plan to another plan, the rules in O&A J 2 and O&A J 3 shall apply.

- 7.2.27.2.1 Distributions in calendar years beginning after December 31, 2002:
- 7.2.27.2.1.1 <u>Supersession</u>. This Section 7.2.27.2.1 will apply for purposes of determining required minimum distributions for calendar years beginning with the 2003 calendar year. Notwithstanding the other provisions of this section, distributions may be made under a designation made before January 1, 1984, in accordance with Section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (TEFRA) and the provisions of the Plan that relate to Section 242(b)(2) of TEFRA.
- 7.2.27.2.1.2 Requirements. The requirements of this Section 7.2.27.2.1 will take precedence over any inconsistent provisions of the Plan. All distributions required under this Plan will be determined and made in accordance with Internal Revenue Code section 401(a)(9), including the incidental death benefit requirement in section 401(a)(9)(G), and the Treasury Regulations thereunder, including the regulations published on April 17, 2002 and June 15, 2004.
- 7.2.27.2.1.2.1 Limits on Distribution Periods. As of the first distribution calendar year, distributions to a Participant, if not made in a single sum, may only be made over one of the following periods:
  - a. the life of the Participant;
  - b. the joint lives of the Participant and a Designated Beneficiary;
  - c. a period certain not extending beyond the life expectancy of the Participant; or
  - d. a period certain not extending beyond the joint life and last survivor expectancy of the Participant and a Designated Beneficiary.
- 7.2.27.2.1.3 <u>Required Beginning Date.</u> The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's required beginning date.
- 7.2.27.2.1.4 <u>Participant's Death Before Distribution Begins.</u> If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
- (a) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, then, except as provided in Sections 7.2.27.2.1.15 and 7.2.27.2.1.16, distributions to the surviving spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age 70½, if later.
- (b) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, then, except as provided in Sections 7.2.27.2.1.15 and 7.2.27.2.1.16, distributions to the Designated Beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died.
- (c) If there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth (5<sup>th</sup>) anniversary of the Participant's death.
- (d) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Section 7.2.27.2.1.4, other than sub-section (a), will apply as if the surviving spouse were the Participant.

For purposes of this Section 7.2.27.2.1.4 and Sections 7.2.27.2.1.8 and 7.2.27.2.1.9, unless sub-

section (d) above applies, distributions are considered to begin on the Participant's Required Beginning Date. If sub-section (d) applies, distributions are considered to begin on the date distributions are required to begin to the surviving spouse under sub-section (a) above. If distributions under an annuity purchased from an insurance company irrevocably commence to the Participant before the Participant's required beginning date (or to the Participant's surviving spouse before the date distributions are required to begin to the surviving spouse under sub-section (a) above), the date distributions are considered to begin is the date distributions actually commence.

7.2.27.2.1.5 Forms of Distribution. Except for defined benefit payments described in Section 7.2.27.2.1.17 below, and unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the required beginning date, as of the first (1<sup>st</sup>) Distribution Calendar Year, distributions will be made in accordance with Sections 7.2.27.2.1.6 through 7.2.27.2.1.9 of this Plan. If the Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions under such annuity will be made in accordance with the requirements of Code section 401(a)(9) and 1.401(a)(9) of the Treasury Regulations. Any part of the Participant's interest which is in the form of an individual account described in section 414(i) of the Code will be distributed in a manner satisfying the requirements of Code section 401(a)(9) and Treasury Regulations section 1.401(a)(9) that apply to individual accounts.

7.2.27.2.1.6 Amount of Required Minimum Distribution for Each Distribution Calendar Year. Except as provided in subsection 7.2.27.2.1.17 if the Participant's interest is paid in the form of an annuity, during the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:

- (a) the quotient obtained by dividing the Participant's Account Balance by the distribution period in the Uniform Lifetime Table set forth in Treasury Regulation section 1.401(a)(9)-9, Q&A-2, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or
- (b) if the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in Treasury Regulation section 1.401(a)(9)-9. Q&A-3, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the Distribution Calendar Year.
- 7.2.27.2.1.7 <u>Lifetime Required Minimum Distributions Continue Through Year of Participant's Death</u>. Required minimum distributions will be determined under Section 7.2.27.2.1.6 beginning with the first (1<sup>st</sup>) Distribution Calendar Year and up to and including the Distribution Calendar Year that includes the Participant's date of death.

### 7.2.27.2.1.8 Death On or After Date Distributions Begin.

- (a) <u>Participant Survived by Designated Beneficiary</u>. If the Participant dies on or after the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining Life Expectancy of the Participant or the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as follows:
- (1) The Participant's remaining Life Expectancy is calculated using the age of the Participant in the year of death, reduced by one (1) for each subsequent year.
- (2) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining Life Expectancy of the surviving spouse is calculated for each Distribution Calendar Year after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving spouse's death, the remaining Life Expectancy of the surviving spouse is calculated using the age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one (1) for each subsequent calendar year.

- (3) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining Life Expectancy is calculated using the age of the beneficiary in the year following the year of the Participant's death, reduced by one (1) for each subsequent year.
- (b) No Designated Beneficiary. If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the Participant's remaining Life Expectancy calculated using the age of the Participant in the year of death, reduced by one (1) for each subsequent year.

## 7.2.27.2.1.9 <u>Death Before Date Distributions Begin.</u>

- (a) <u>Participant Survived by Designated Beneficiary</u>. Except as provided in Sections 7.2.27.2.1.15 and 7.2.27.2.1.16, if the Participant dies before the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as provided in Section 7.2.27.2.1.8.
- (b) <u>No Designated Beneficiary</u>. If the Participant dies before the date distributions begin and there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest will be completed by December 31 of the calendar year containing the fifth (5<sup>th</sup>) anniversary of the Participant's death.
- (c) <u>Death of Surviving Spouse Before Distributions to Surviving Spouse Are Required to Begin</u>. If the Participant dies before the date distributions begin, the Participant's surviving spouse is the Participant's sole Designated Beneficiary, and the surviving spouse dies before distributions are required to begin to the surviving spouse under Section 7.2.27.2.1.4(a), this Section 7.2.27.2.1.9 will apply as if the surviving spouse were the Participant.
- 7.2.27.2.1.10 <u>Designated Beneficiary</u>. The Designated Beneficiary is the individual who is designated by the Participant (or the Participant's surviving spouse) as the beneficiary of the Participant's interest under the Plan and is the Designated Beneficiary under Code section 401(a)(9) and Treasury Regulation section 1.401(a)(9)-4.
- 7.2.27.2.1.11 <u>Distribution Calendar Year</u>. A Distribution Calendar Year is a calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year which contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin under Section 7.2.27.2.1.4. The required minimum distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Beginning Date. The required minimum distribution for the Distribution Calendar Years, including the required minimum distribution for the Distribution Calendar Year in which the Participant's Required Beginning Date occurs, will be made on or before December 31 of that Distribution Calendar Year.
- 7.2.27.2.1.12 <u>Life Expectancy</u>. Life Expectancy means the expectancy as computed by use of the Single Life Table in Treasury Regulation section 1.401(a)(9)-9, Q&A-1.
- 7.2.27.2.1.13 Participant's Account Balance. The Account Balance is the balance in the participant's account as of the last valuation date in the calendar year immediately preceding the Distribution Calendar Year (valuation calendar year) increased by the amount of any contributions made and allocated or forfeitures allocated to the Account Balance as of dates in the valuation calendar year after the valuation date and decreased by distributions made in the valuation calendar year after the valuation

date. The Account Balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the Distribution Calendar Year if distributed or transferred in the valuation calendar year.

7.2.27.2.1.14 <u>Required Beginning Date</u>. The Required Beginning Date is April 1<sup>st</sup> of the calendar year following the later of the calendar year in which the Participant attains age 70 ½ or the calendar year in which the Participant retires.

7.2.27.2.1.15 Participants or Beneficiaries Permitted to Elect 5-Year Rule. Participants or beneficiaries may elect on an individual basis whether the 5-year rule or the life expectancy rule in Sections 7.2.27.2.1.4 and 7.2.27.2.1.9 of this Plan applies to distributions after the death of a Participant who has a Designated Beneficiary. The election must be made no later than the earlier of September 30 of the calendar year in which distribution would be required to begin under Section 7.2.27.2.1.4 of this Plan, or by September 30 of the calendar year which contains the fifth (5<sup>th</sup>) anniversary of the Participant's (or, if applicable, surviving spouse's) death. If neither the Participant nor beneficiary makes an election under this Section, distributions will be made in accordance with Sections 7.2.27.2.1.4 and 7.2.27.2.1.9 and, if applicable, the elections in Section 7.2.27.2.1.16 below.

7.2.27.2.1.16 Election to Apply 5-Year Rule to Distributions to Designated Beneficiaries. For all distributions, if the Participant dies before distributions begin and there is a Designated Beneficiary, distribution to the Designated Beneficiary is not required to begin by the date specified in the Plan, but the Participant's entire interest will be distributed to the Designated Beneficiary by December 31 of the calendar year containing the fifth (5<sup>th</sup>) anniversary of the Participant's death. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to either the Participant or the surviving spouse begin, this election will apply as if the surviving spouse were the Participant.

7.2.27.2.1.17 In the case of a payment that is treated as a payment from a defined benefit plan, unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the Required Beginning Date, as of the first Distribution Calendar Year distributions will be made in accordance with Sections 7.2.27.2.1.17.1 through 7.2.27.2.1.17.3 below. If the Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code section 401(a)(9) and Section 1.401(a)(9) of the Treasury Regulations. Any part of the Participant's interest which is in the form of an individual account described in Code section 414(k), will be distributed in a manner satisfying the requirements of Code section 401(a)(9) and Section 1.401(a)(9) of the Treasury Regulations thereunder that apply to individual accounts.

### 7.2.27.2.1.17.1 Determination of Amount to be Distributed Each Year.

- (a) <u>General Annuity Requirements</u>. If the Participant's interest is paid in the form of annuity distributions under the Plan, payments under the annuity will satisfy the following requirements:
  - (i) the annuity distributions will be paid in periodic payments made at uniform intervals not longer than one year;
  - (ii) the distribution period will be over a life (or lives) or over a period certain not longer than the period described in Section 7.2.27.2.1.17.2 or 7.2.27.2.1.17.3;
  - (iii) once payments have begun, the period will be changed only in accordance with Section 7.2.27.2.1.17.4 below;
  - (iv) payments will either be nonincreasing or increase only as follows:

- (A) by an annual percentage increase that does not exceed the percentage increase in an eligible cost-of-living index for a 12-month period ending in the year during which the increase occurs or a prior year;
- (B) by a percentage increase that occurs at specified times and does not exceed the cumulative total of annual percentage increases in an eligible cost of living index since the annuity starting date, or if later, the date of the most recent percentage increase;
- (C) by a constant percentage of less than 5 percent per year, applied not less frequently than annually;
- (D) as a result of dividend or other payments that result from actuarial gains provided:
  - (i) actuarial gain is measured not less frequently than annually,
  - (ii) the resulting dividend or other payments are either paid no later than the year following the year for which the actuarial experience is measured or paid in the same form as the payment of the annuity over the remaining period of the annuity (beginning no later than the year following the year for which the actuarial experience is measured),
  - (iii) the actuarial gain taken into account is limited to actuarial gain from investment experience,
  - (iv) the assumed interest rate used to calculate such actuarial gains is not less than 3%, and
  - (v) the annuity payments are not increased by a constant percentage as described in "(C)" above.
- (E) to the extent of the reduction in the amount of the Participant's payments to provide for a survivor benefit, but only if there is no longer a survivor benefit because the beneficiary whose life was being used to determine the distribution period described in 7.2.27.2.1.17.2 dies or is no longer the Participant's beneficiary pursuant to a qualified domestic relations order within the meaning of Code section 414(p):
- (F) to provide a final payment upon the Participant's death not greater than the excess of the actuarial present value of the Participant's accrued benefit (within the meaning of Code section 411(a)(7)) calculated using the interest rate on 30 year Treasury securities as specified by the Commissioner and described in section 3.6.9(i)((II)(a) above and mortality table set forth in Rev. Rul. 2001-62, 2001-53, I.R.B. 632 (or, if greater, the total amount of employee contributions) over the total of payments before the Participant's death;

- (G) to allow a beneficiary to convert the survivor portion of a joint and survivor annuity into a single sum distribution upon the Participant's death; or
- (H) to pay increased benefits that result from a plan amendment.
- (b) Amount Required to be Distributed by Required Beginning Date. The amount that must be distributed on or before the Participant's Required Beginning Date (or, if the Participant dies before distributions begin, the date distributions are required to begin under Section 7.2.27.2.1.4(a) or (b)) is the payment that is required for one payment interval. The second payment need not be made until the end of the next payment interval even if that payment interval ends in the next calendar year. All of the Participant's benefit accruals as of the last day of the first Distribution Calendar Year will be included in the calculation of the amount of the annuity payments for payment intervals ending on or after the Participant's Required Beginning Date.
- (c) Additional Accruals After First Distribution Calendar Year. Any additional benefits accruing to the Participant in a calendar year after the first Distribution Calendar Year will be distributed beginning with the first payment interval ending in the calendar year immediately following the calendar year in which such amount accrues.

# 7.2.27.2.1.17.2 <u>Requirements For Annuity Distributions That Commence During Participant's Lifetime.</u>

- (a) Joint Life Annuities Where the Beneficiary is Not the Participant's Spouse. If the Participant's interest is being distributed in the form of a joint and survivor annuity for the joint lives of the Participant and a nonspouse beneficiary, annuity payments to be made on or after the Participant's Required Beginning Date to the Designated Beneficiary after the Participant's death must not at any time exceed the applicable percentage of the annuity payment for such period that would have been payable to the Participant using the table set forth in Q&A-2(c)(1) of Treasury Regulation section 1.401(a)(9)-6 to determine the applicable percentage. If the form of distribution combines a joint and survivor annuity for the joint lives of the Participant and a nonspouse beneficiary and a period certain annuity, the requirement in the preceding sentence will apply to annuity payments to be made to the Designated Beneficiary after the expiration of the period certain.
- (b) Period Certain Annuities. Unless the Participant's spouse is the sole Designated Beneficiary and the form of distribution is a period certain and no life annuity, the period certain for an annuity distribution commencing during the Participant's lifetime may not exceed the applicable distribution period for the Participant under the Uniform Lifetime Table set forth in Treasury Regulation section 1.401(a)(9)-9, Q&A-2, for the calendar year that contains the annuity starting date. If the annuity starting date precedes the year in which the Participant reaches age seventy (70), the applicable distribution period for the Participant is the distribution period for age seventy (70) under the Uniform Lifetime Table set forth in Treasury Regulation section 1.401(a)(9)-9, Q&A-2, plus the excess of seventy (70) over the age of the Participant as of the Participant's birthday in the year that contains the annuity starting date. If the Participant's spouse is the Participant's sole Designated Beneficiary and the form of distribution is a period certain and no life annuity, the period certain may not exceed the longer of the Participant's applicable distribution period, as determined under this subsection or the joint life and last survivor expectancy of

the Participant and the Participant's spouse as determined under the Joint and Last Survivor Table set forth in Treasury Regulation section 1.401(a)(9)-9, Q&A-3, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the calendar year that contains the annuity starting date.

### 7.2.27.2.1.17.3 Requirements for Minimum Distributions After Participant Dies

- (a) <u>Death After Distributions Begin</u>. If the Participant dies after distribution of his or her interest begins in the form of an annuity meeting the requirements of this section 7.2.27.2.1, the remaining portion of the Participant's interest, if any, will continue to be distributed over the remaining period over which distributions commenced.
- (b) <u>Death Before Distributions Begin.</u>
  - (i) Participant Survived By Designated Beneficiary. If the Participant dies before the date distribution of his or her interest begins and there is a Designated Beneficiary, the Participant's entire interest will be distributed, beginning no later that the time described in Section 7.2.27.2.1.4(a) or (b), over the life of the Designated Beneficiary or over a period certain not exceeding:
    - (A) unless the annuity starting date is before the first Distribution Calendar Year, the life expectancy of the Designated Beneficiary determined using the beneficiary's age as of the beneficiary's birthday in the calendar year immediately following the calendar year of the Participant's death; or
    - (B) if the annuity starting date is before the first Distribution Calendar Year, the life expectancy of the Designated Beneficiary determined using the beneficiary's age as of the beneficiary's birthday in the calendar year that contains the annuity starting date.

Election of 5 Year Rule. Participants or beneficiaries may elect on an individual basis whether the five (5) year rule or the life expectancy rule in Section 7.2.27.2.1.4 and Section 7.2.27.2.1.17.2(b) applies to distributions after the death of a Participant who has a Designated Beneficiary. The election must be made no later than the earlier of September 30 of the calendar year in which distributions would be required to begin under Section 7.2.27.2.1.4, or September 30 of the calendar year which contains the fifth (5<sup>th</sup>) anniversary of the Participant's (or, if applicable, surviving spouse's) death. If neither the Participant nor beneficiary makes an election under this subsection, distributions will be made in accordance with Sections 7.2.27.2.1.4 and 7.2.27.2.1.17.2(b).

- (ii) No Designated Beneficiary. If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the year after the year of the Participant's death, distribution of the Participant's entire interest will be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (iii) <u>Death of Surviving Spouse Before Distributions to Surviving Spouse Begin.</u> If the Participant dies before the date distribution of his or her interest begins, the Participant's surviving spouse is the Participant's

sole Designated Beneficiary, and the surviving spouse dies before distributions to the surviving spouse begin, this subsection will apply as if the surviving spouse were the Participant, except that the time by which distributions must begin will be determined without regard to Section 7.2.27.2.1.4(a).

### 7.2.27.2.1.17.4 Changes to Annuity Payment Period.

- (a) Permitted Changes. An annuity payment period may be changed only in association with an annuity payment increase described in Section 7.2.27.2.1.17.1(a)(iv) (or in accordance with subsection (b) below).
- (b) Reannuitization. An annuity payment period may be changed and the annuity payments modified in accordance with that change if the conditions in subsection (c) below are satisfied and:
  - (i) the modification occurs when the Participant retires or in connection with a Plan termination:
  - (ii) the payment period prior to modification is a period certain without life contingencies; or
  - (iii) the annuity payments after modification are paid under a qualified joint and survivor annuity over the joint lives of the Participant and a designated beneficiary, the Participant's spouse is the sole designated beneficiary, and the modification occurs in connection with the Participant's becoming married to such spouse.
- (c) Conditions. The conditions in this subsection (c) are satisfied if:
  - (i) the future payments after the modification satisfy the requirements of Section 409(a)(9), Section 1.401(a)(9) of the regulations, and this Section 7.2 (determined by treating the date of the changes as the new annuity starting date and the actuarial present value of the remaining payments prior to modification as the entire interest of the Participant);
  - (ii) for purposes of Code Sections 415 and 417, the modification is treated as a new annuity starting date;
  - (iii) after taking into account the modification, the annuity (including all past and future payments) satisfies the requirements of Code Section 415 (determined at the original annuity starting date, using the interest rates and mortality tables applicable to such date); and
  - (iv) the end point of the period certain, if any, for any modified payment period is not later than the end point available to the Employee at the original annuity starting date under Code Section 401(a)(9) and this Section 7.2.

# 7.2.27.2.1.17.5 Payments to Surviving Child.

- (a) Special Rule. For purposes of this Section 7.2, any payments made to a Participant's surviving child until the child reaches the age of majority (or dies, if earlier) shall be treated as if such payments were made to the surviving spouse to the extent the payments become payable to the surviving spouse upon cessation of the payments to the child.
- (b) Age of Majority. For purposes of this section, a child shall be treated as having not

reached the age of majority if the child has not completed a specified course of education and is under the age of 26. In addition, a child who is disabled within the meaning of Section 72(m)(7) when the child reaches the age of majority shall be treated as having not reached the age of majority so long as the child continues to be disabled.

### <del>7.2.2</del>7.2.1.17.6 Additional Definitions

- (a) Actuarial Gain. The difference between an amount determined using the actuarial assumptions (i.e. investment return, mortality, expense and other similar assumptions) used to calculate the initial payments before adjustment for any increases and the amount determined under the actual experience with respect to those factors. Actuarial gain also includes differences between the amount determined using actuarial assumptions when an annuity was purchased or commenced and such amount determined using actuarial assumptions used in calculating payments at the time the actuarial gain is determined.
- (b) Eligible Cost of Living Index. An index described in paragraphs (b)(2), (b)(3) or (b)(4) of section 1.401(a)(9)-6, Q&A-14 of the regulations.

### <del>7.2.2</del>7.2.1.18 TEFRA section 242(b)(2) Elections

- (a) Notwithstanding the other requirements of this 7.2, distribution on behalf of any Employee who has made a designation under Section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (a "242(b)(2) election") may be made in accordance with all of the following requirements (regardless of when such distribution commences):
  - (i) The distribution by the Plan is one which would not have disqualified such Plan under section 401(a)(9) of the Internal Revenue Code as in effect prior to amendment by the Deficit Reduction Act of 1984,
  - (ii) The distribution is in accordance with a method of distribution designated by the Employee whose interest in the Plan is being distributed or, if the Employee is deceased, by a beneficiary of such Employee,
  - (iii) Such designation was in writing, was signed by the Employee or the beneficiary, and was made before January 1, 1984,
  - (iv) The Employee had accrued a benefit under the Plan as of December 31, 1983,
  - (v) The method of distribution designated by the Employee or the beneficiary specifies the time at which distribution will commence, the period over which distributions will be made, and, in the case of any distribution upon the Employee's death, the beneficiaries of the Employee listed in order of priority.
- (b) A distribution upon death will not be covered by this transitional rule unless the information in the designation contains the required information described above with respect to the distributions to be made upon the death of the Employee.
- (c) For any distribution which commences before January 1, 1984, but continues after December 31, 1983, the Employee, or the beneficiary, to whom such distribution is being made, will be presumed to have designated the method of distribution under which the distribution is being made if the method of distribution was specified in writing and the distribution satisfies the requirements in subsections (a)(i) and (v) above.
- (d) If a designation is revoked, any subsequent distribution must satisfy the requirements of § 401(a)(9) of the Code and the regulations thereunder. If a designation is revoked subsequent to the date distributions are required to begin, the Plan must distribute by the

end of the calendar year following the calendar year in which the revocation occurs the total amount not yet distributed which would have been required to have been distributed to satisfy Section 401(a)(9) of the Code and the regulations thereunder, but for the 242(b)(2) election. For calendar years beginning after December 31, 1988, such distributions must meet the minimum distribution incidental benefit requirements. Any changes in the designation will be considered to be a revocation of the designation. However, the mere substitution or addition of another beneficiary (one not named in the designation) under the designation will not be considered to be a revocation of the designation, so long as such substitution or addition does not alter the period over which distributions are to be made under the designation directly or indirectly (for example by altering the relevant measuring life).

(e) In the case in which an amount is transferred or rolled over from one plan to another plan, the rules in Section 1.401(a)(9)-8, Q&A-14 and Q&A-15 of the regulations shall apply.

7.2.2.19 <u>Transition Rules</u>. The transition rules for required minimum distributions before January 1, 2003 are specified in section 7.2.1 above.

7.2.1.20 2009 Required Minimum Distributions.

- (a) Suspension of Required Minimum Distributions for 2009. Notwithstanding anything in this Plan to the contrary, a Participant or Beneficiary who would have been required to receive required minimum distributions for 2009 but for the enactment of Code Section 401(a)(9)(H) ("2009 RMDs"), and who would have satisfied that requirement by receiving distributions that are (i) equal to the 2009 RMDs or (ii) one or more payments in a series of substantially equal distributions (that include the 2009 RMDs) made at least annually and expected to last for the life (or life expectancy) of the Participant, the joint lives (or joint life expectancy) of the Participant and the Participant's designated "Beneficiary, or for a period of at least 10 years ("Extended 2009 RMDs"), will not receive those distributions for 2009 unless the Participant or Beneficiary chooses to receive such distributions. Participants and Beneficiaries described in the preceding sentence will be given the opportunity to elect to receive the distributions described in the preceding sentence; and
- (b) **Direct Rollovers**. For purposes of applying the direct rollover provisions of the Plan, a direct rollover will be offered only for distributions that would be eligible rollover distributions without regard to Code Section 401(a)(9)(H).
- $\frac{7.2.2}{\text{to the contrary, a governmental Plans.}} \frac{7.2.2}{\text{to the contrary, a governmental plan within the meaning of Code section 414(d), or an eligible governmental plan described in Treasury Regulations section 1.457-2(f), is treated as having complied with Code section 401(a)(9) for all years to which section 401(a)(9) applies to the Plan if the Plan complies with a reasonable good faith interpretation of section 401(a)(9).}$
- 7.3 <u>Minimum Benefits Upon Retirement, Death and Disability of Certain Police and Fire Plan Participants.</u> Notwithstanding any provision of this Agreement to the contrary, this Section 7.3 shall in certain circumstances provide minimum benefits to Police and Fire Plan Participants as follows:
  - 7.3.1 <u>Minimum Retirement Benefits</u>. Participants of Police and Fire Plans, if employed on January 1, 1984 and continuously employed by the City from such date through the date of their retirement, shall receive a benefit which, when determined on a Straight Life Annuity basis, shall not be less than:
    - (a) <u>Police Plan</u>.
    - (i) <u>Effective April 16, 2012, 50</u>% of Regular Pay if retirement occurs after reaching 60 years of age and the Participant has completed 25 years of service with the City. <u>Before April 16, 2012</u>, the following parenthetical was included at the end: (or 21 Years of Service if hired prior to November 18, 1965); or

(ii) 40% of Regular Pay if retirement occurs after reaching 55 years of age, but before reaching 60 years of age, and the Participant has completed 25 Years of Service with the City.

The Minimum Benefit provided in this Section (a) shall be paid in any form of benefit payment otherwise provided for in this Section 7. If the Minimum Benefit is paid in a form other than a straight life annuity, such benefit shall be the Actuarial Equivalent of the Minimum Benefit payable as a Straight Life Annuity.

If the Participant chooses the single lump-sum payment option, the Participant can request that the Actuarial Equivalent be equal to the average of the cost of three Annuity Contracts purchased on the open market. The Participant, Retirement Committee and City each shall submit the cost of an appropriate Annuity Contract to determine the Actuarial Equivalent.

### (b) <u>Fire Plan</u>.

- (i) 50% of Regular Pay if retirement occurs after reaching 55 years of age and completing 21 years of service with the City; or
- (ii) the Actuarial Equivalent of the benefit which otherwise would be provided in (i) above at 55 years of age if retirement occurs after reaching 50 years of age, but before reaching age 55, and Retirement occurs after completing 21 Years of Service with the City; or
- (iii) 50% of the Salary received at the time of retirement multiplied by the ratio of Years of Service to 21 if retirement from the City occurs on or after reaching 55 years of age with less than twenty-one Years of Service with the City; or
- (iv) For termination of employment after September 9, 1993, 50% of Regular Pay if such termination of employment occurs prior to 55 years of age but after completion of 21 Years of Service with the City.

Unless an optional annuity benefit is selected by the Participant, at the death of the Participant, the same rate of pension as is provided for in this Section (b) shall be paid to the surviving spouse of such deceased Participant during such time as the surviving spouse remains unmarried and, in case there is no surviving spouse, then the minor children, if any, of such deceased Participant shall equally share such Minimum Benefit during their minority. As soon as a child of a deceased Participant ceases to be a minor, such benefit to such child shall cease.

In the event a Participant or his or her surviving beneficiaries die before the aggregate amount of Minimum Benefit payments distributed under this Section (b) equals the total amount in the Participant's Employee Account at the time of the first payment, the difference between the total amount in the Employee Account and the aggregate amount of Minimum Benefit payments distributed shall be paid in a single sum to the Participant's estate.

The Minimum Benefit provided for in this Section (b) shall be paid in any form permitted under Section 7. If the Minimum Benefit is paid in an optional annuity benefit or a single lump-sum payment, such benefit or payment shall be the Actuarial Equivalent of the annuity that would otherwise be paid to the Participant.

If the Participant chooses the single lump-sum payment option, the Participant may request that the Actuarial Equivalent be equal to the average of the cost of two Annuity Contracts purchased on the open market, if the difference between the cost of the two Annuity Contracts does not exceed 5%. The Participant and the City each shall choose one of the Annuity Contracts used for determining the Actuarial Equivalent. If the difference between the two Annuity Contracts chosen exceeds 5%, the Retirement Committee shall review the costs of the two

contracts and make a recommendation to the City Council as to the amount of the lump-sum payment to be made to the Participant. The City Council shall determine the amount of the single lump-sum payment after a hearing thereon.

- (c) <u>Retirement Benefits for Certain Firefighters Employed on August 7, 1965.</u>
- (i) All firefighters of the paid fire department of a city of the first class in Nebraska who:
  - (1) were serving as such on August 7, 1965;
  - (2) did not elect coverage under the provisions of Neb. Rev. Stat. Sections 35-204 to 35-215 as in existence prior to January 1, 1984; and
    - (3) served in the fire department for a period of 21 years

shall elect to retire from active service, go upon the retired list, and receive a pension of at least 50% of the amount of Salary such retiring firefighter is receiving at the time he or she goes upon the pension list. Such benefits shall be paid by the City in the same manner as firefighters upon the active list are paid.

- (ii) Any such firefighter who retires on or after attaining 55 years of age with less than 21 years of service with the City shall receive a benefit of at least 50% of the Salary he or she was receiving at the time of his or her retirement multiplied by the ratio of the years of service to 21.
- (iii) At the death of any such retired firefighter, the same rate of pension, as provide for herein, shall be paid to the surviving spouse of such deceased firefighter during such time as the surviving spouse shall remain unmarried and, in case there be no surviving spouse, then the minor children, if any, of such deceased firefighter, shall be paid such benefit during their minority to the age of eighteen years, except that as soon as a child of such deceased firefighter shall become eighteen years of age, such pension as to such child shall cease.
- (iv) Firefighters described in subsection (c) above shall be subject to Sections 16-1029 to 16-1032 of the Nebraska statutes, but shall be exempt from Sections 16-1024, 16-1025, 16-1027, 16-1028 and 16-1033.
- (v) After August 7, 1965, every firefighter subject to the provisions of Sections 35-201 to 35-203 as in existence prior to January 1, 1984 shall contribute to the City an amount equal to 5% of his or her Salary until he or she shall be entitled to retire or otherwise become eligible for a pension. No such firefighter continuing in the employment of the City as a member of such department after becoming eligible to retire shall be required to make any further contribution. Any such firefighter whose employment shall terminate, whether by discharge or otherwise, prior to the time he or she shall become entitled to a pension, and who shall have made contributions from his or her Salary as provided in this subsection shall, upon demand, be reimbursed by the City for the amount of such contributions plus interest at 5% per annum.
- (vi) Nothing in the Nebraska statutes regarding the Firefighters Retirement System Fund nor any provision of this Plan shall in any manner affect the right of any person now receiving or entitled to receive, now or in the future, pension or other benefits provided for in Section 35-201 to 35-216, as they exist immediately prior to January 1, 1984, to receive such pension or other benefits in all respects the same as if such Sections remained in full force and effect.

### 7.3.2 Minimum Death Benefits.

- (a) <u>Death in the Line of Duty Police and Fire Plans</u>. A Participant's surviving spouse or minor children shall be paid a benefit of 50% of Regular Pay if the Participant (who with respect to a Fire Plan, is participating in a Police or the Fire Plan) died in the line of duty or as a result of injuries received while in the line of duty ("Minimum Death Benefit in the Line of Duty"). The Minimum Death Benefit in the Line of Duty shall be paid to the surviving spouse, and upon his or her remarriage or death, to the minor children during each child's minority, subject to deduction of the amounts paid as workers' compensation benefits on account of death. Each child eligible to receive benefits hereunder shall share equally in the Minimum Death Benefit in the Line of Duty until he or she reaches the age of majority. The Minimum Death Benefit in the Line of Duty to a child shall cease upon his or her reaching the age of majority.
  - (i) <u>Police Plan</u>. To the extent that the Retirement Value of the Participant's Account on the date of death exceeds the amount required to purchase or provide the Minimum Death Benefit in the Line of Duty, the excess shall be paid pursuant to Sections 6 and 7 of this Plan.
  - (ii) <u>Fire Plan</u>. In the event the surviving spouse or minor children of a deceased Participant die before the aggregate amount of payments from the Plan received by the Participant and his or her survivor beneficiaries, if any, equals the total amount in the Participant's Employee Account at the time of the first benefit payment, the difference between the total amount in the Employee Account and the aggregate amount of benefit payments made by the Plan, if any, shall be paid in a single sum to the Participant's beneficiary or, in the absence thereof, to the Participant's estate.
  - (iii) The Retirement Value of a Participant's Account shall be paid to the Beneficiary designated by the Participant prior to his death, or to the Participant's Estate in the absence of a designated Beneficiary, in the event that a Participant who dies in the line of duty, or as a result of injuries received in the line of duty, is not survived by a spouse or minor children. Such payment shall be made to the Beneficiary in a straight life annuity, single lump-sum or other optional form of benefit specified in this Agreement and elected by the Beneficiary.

### (b) Death Not in the Line of Duty – Police and Fire Plans.

- (i) <u>Police Plan</u>. A benefit of at least 25% of Regular Pay in the form of a Straight Life Annuity shall be paid to the surviving spouse of a deceased participant of a Police Plan who dies other than in the line of duty or as a result of injuries received in the line of duty if the Participant:
  - (a) was continuously employed by the City from January 1, 1984, except those who shall have been formerly employed by the City who are now in military duty, until the date of his or her death;
    - (b) had reached 55 years of age;
    - (c) had not elected to retire; and
    - (d) had completed at least 21 Years of Service with the City.
- (ii) <u>Fire Plan</u>. A benefit of at least 25% of Regular Pay in the form of a Straight Life Annuity to the surviving spouse of a Participant of a Fire Plan who dies other than in the line of duty or as a result of injuries received while in the line of duty, if the Participant:

Grand Island

- (a) was employed by the City as a firefighter on January 1, 1984, or reemployed thereafter who, while employed as a firefighter, and entered military service and is still in military service;
- (b) dies while employed by the City;
- (c) had attained 50 years of age;
- (d) had not elected to retire; and
- (e) had 21 Years of Service with the City.

If the surviving spouse or minor children of a deceased Participant die before the aggregate amount of benefits paid to the Participant and his or her Beneficiaries equals the total amount in the Participant's Employee Account on the date of the first payment, the difference between the value of the Employee Account and the aggregate amount of benefits paid shall be paid in a single sum to the designated Beneficiary, or to the estate of the Participant in the absence of a surviving designated Beneficiary.

(iii) If the deceased Participant is not survived by a spouse or in the event the surviving spouse dies before the minor children of the Participant attain the age of majority, the pension benefit shall be paid to the Participant's minor children until they have attained the age of majority. Each minor child shall equally share in the Minimum Death Benefit. The Benefit to a child shall cease upon the child reaching the age of majority. The Benefit shall thereafter be allocated among the remaining minor children until the last remaining child dies or reaches the age of majority.

To the extent that the Retirement Value on the date of the Participant's death exceeds the amount required to purchase a Minimum Death Benefit specified in this Section 7.3, the excess shall be paid pursuant to Sections 6 & 7 of this Plan.

- (c) Any payments for the benefit of a minor child shall be made on behalf of such child to the surviving parent or, if there is no surviving parent, to the legal guardian of the child.
- 7.3.3 <u>Minimum Disability Benefits</u>. The following benefits are payable to Participants of Police and Fire Plans who suffer a Disability:
  - (a) Disability in the Line of Duty: A benefit of 50% of Regular Pay shall be paid to any Participant who becomes disabled and it is determined that the disability is permanent ("Permanent Disability").
    - (i) Disabled or Disability Defined: The complete inability of the Participant, for reasons of accident or other cause while in the line of duty, to perform the duties of a firefighter as defined by fire department job descriptions or ordinance, or with respect to a Police Plan, the duties of a police officer.
    - (ii) A benefit payment shall be made under this Section 7.3.3(a) only upon adequate proof of the Disability. Such proof shall consist of a medical examination conducted by a competent, disinterested physician who is duly licensed to practice medicine and surgery in the state of Nebraska and who certifies to the City that the Participant is unable to perform the duties of his or her job.
    - (iii) The City shall have the right during the first three years of Disability payments to require the Disabled Participant to undergo a medical examination at the City's expense to determine the continuance of the Disability claimed. After such three-year period, the City may request the district court to order the Participant to submit proof

of the continuance of the Disability claimed if the City has reasonable grounds to believe the Participant is fraudulently receiving Disability payments. The City shall have the right to demand a physical examination of the Participant by a competent, disinterested physician who is duly licensed to practice medicine and surgery in the state of Nebraska and who is chosen by the City. The expense of such examination shall be borne by the City.

- (iv) Temporary Disability. A Participant who receives a temporary Disability while in the line of duty shall receive his or her salary during the continuance of the Disability for a period not to exceed 12 months. If the City determines within the initial 12 month period that a Temporary Disability has become a Permanent Disability, then salary payments shall cease and the Participant shall be entitled to the payment of benefits for a Permanent Disability pursuant to Section 7.3.3(a).
- (v) All payments of benefits or salary in this Section 7.3.3. shall be subject to deduction of amounts paid under the Nebraska Workers' Compensation Act. With respect to Disability payments under a Police Plan, such payments shall not commence until all credit for unused annual or sick leave and other similar credits have been fully utilized by the Disabled Participant if there will be no impairment to his or her salary during the period of Disability. Total payments to a Disabled Participant, in excess of amounts paid as workers' compensation benefits, shall not be less than the Retirement Value of the Participant's Account on the date of the Disability.

No Participant shall be entitled during any period of Temporary Disability to receive in full both his or her Salary and his or her benefits under the Nebraska Workers' Compensation Act ("Act"). All Nebraska workers' compensation benefits shall be payable in full to such firefighter as provided in the Act, but all amounts paid by the City or its insurer under the Act to any Disabled firefighter entitled to receive a Salary during such Disability shall be considered as payments on account of such Salary and shall be credited thereon. The remaining balance of such Salary, if any, shall be payable as otherwise provided herein.

- (vi) Unless otherwise provided herein, the benefits provided under this Section 7.3.3 shall terminate upon the cessation of a Participant's Disability, and the Retirement Value of the Participant's Account shall be reduced by the amount of benefits distributed to the Participant hereunder, and thereafter such Account shall be held and administered in the same manner as the Account of Participants who are not disabled.
- (b) Disability Not in the Line of Duty: Disability payments shall not be paid to a Participant in the event of a disability which is not received while in the line of duty.
- 7.4 <u>Effect of Reemployment After Distribution Has Been Made or Commenced.</u> In the event that a Participant is reemployed by the Employer after distribution has been made or commenced to him, the following rules shall apply:
  - (a) Further distribution of his Matured Account shall be suspended and the undistributed remainder of his Matured Account shall continue to be held in the Plan until another Event of Maturity effective as to him shall occur after his reemployment, it being the intent hereof that no distributions shall be made while a Participant is maintaining an employment relationship with the Employer.
  - (b) He shall again become a Participant in this Plan upon his return to Recognized Employment as provided in this Plan.
- 7.5 <u>Designation of Beneficiaries</u>. Each Participant may designate, upon forms to be furnished by and filed with the Trustee, a Beneficiary or Beneficiaries to receive his Matured Account in the event of his death and may change or revoke any such designation and appointment from time to time. No such designation, change or

revocation shall be effective unless executed by the Participant and delivered to the Trustee during the Participant's lifetime. In the event:

- (a) the Participant shall have failed to designate a Beneficiary,
- (b) such a designation shall have been made and revoked and the Participant shall have failed to designate another, or
  - (c) a Beneficiary or Co-Beneficiaries so designated shall fail to survive the Participant;

Participant's Matured Account, or the part thereof as to which such Participant's designation shall fail, as the case may be, shall be payable at the time of the failure to the first class of the following classes of automatic Beneficiaries then surviving and (except in the case of his surviving issue) in equal shares if there are then more than one in each class:

Police and Fire Plans: Participant's Estate
Participant's surviving spouse
Participant's surviving issue per stirpes and not per capita
Participant's surviving parents
Participant's surviving brothers and sisters
Representative of Participant's estate

When used herein and, unless the Participant has otherwise specified in his Beneficiary designation, when used in a Beneficiary designation, "per stirpes" means in equal shares among living children and the issue (taken collectively) of each deceased child, with such issue taking by right of representation; "children" means issue of the first generation; and "issue" means all persons who are descended from the person referred to, either by legitimate birth to or legal adoption by him or any of his legitimately born or legally adopted descendants. The automatic Beneficiaries specified above and, unless the Participant has otherwise specified in his Beneficiary designation, the Beneficiaries designated by him shall become fixed as of the Participant's death so that, if a Beneficiary survives the Participant hereunder and then dies before receipt of all payments, such remaining payments shall be payable to the representative of such Beneficiary's estate. Any designation of Beneficiary by name that is accompanied by a description of relationship to the Participant shall be given effect without regard to whether the relationship to the Participant exists either then or at the Participant's death. Any designation of a Beneficiary only by statement of relationship to the Participant shall be effective only to designate the person or persons standing in such relationship to the Participant at the Participant's death.

- 7.6 <u>Optional Forms of Distribution</u>. As designated by the Employer in the Adoption Agreement, the optional forms of benefit payment provided by this Plan are as follows:
  - (a) a single sum;
  - (b) a straight life annuity;
  - (c) a straight life annuity with a guarantee of at least 60 monthly payments;
  - (d) an annuity payable for life of Participant and annuity to surviving beneficiary of 100%, 75% or 50%;
    - (e) a combination of (a) through (d);
  - (f) if so indicated in the Adoption Agreement, in kind distribution is permitted subject to Section 7.7; and
    - (g) any other form designated in the Adoption Agreement.

For the portion of the death benefit that is payable to the Participant's surviving spouse, the surviving spouse may elect distribution at any time and in any form (other than a joint and survivor annuity), including a single lump-sum payment, which would have been permitted for the deceased Participant; subject, however, to the right of the City to make a cash-out distribution as described in Section 5.1.3(a) above. Similarly, the Participant's Beneficiary may elect to have the Trustee distribute the death benefit in any form (other than a joint and survivor annuity), including a single lump-sum payment, which would have been permitted for the deceased Participant, and within a period permitted under Section 7; subject, however, to the right of the City to make a cash-out distribution as described in Section 5.1.3(a) above. The Beneficiary's election is subject to any restrictions designated in writing by the Participant and not revoked as of his date of death. Such election shall be by a Participant's surviving spouse or Beneficiary made within 60 days after an Event of Maturity.

The amount of annuity benefit payable in the form of an annuity under this Section 7.6 shall be the amount paid by the Annuity Contract purchased or otherwise provided by the Participant's <u>Vested Retirement Value</u> as of the date of the first payment. Any Annuity Contract purchased by the Trustee may be distributed to the Participant or Beneficiary. Upon distribution of an Annuity Contract or lump-sum payment, all obligations of the Plan to the Participant or Beneficiary shall cease. <u>Any Annuity Contract that is distributed from the Plan must be nontransferable.</u>

A Participant who retires after reaching his or her Retirement Date may elect to defer the date of the first annuity payment or lump-sum distribution to the first day of any specified month prior to the Participant reaching 70 years of age.

Any retiring Participant whose benefit under the Plan is less than \$25 per month on the Straight Life Annuity option shall not be entitled to elect to receive annuity benefits.

Any payments for the benefit of a minor child shall be made on behalf of the child to the surviving parent or, if there is no surviving parent, to the legal guardian of the child.

(h) Effective for distributions with annuity starting dates in Plan Years beginning after December 31, 2007, a married Participant who elects to waive the qualified joint and survivor annuity form of benefit under the Plan, if said form is required under the Plan by applicable law, shall be entitled to elect the "qualified optional survivor annuity" at any time during the applicable election period. Furthermore, the written explanation of the joint and survivor annuity, if required of the Plan by applicable law, shall explain the terms and conditions of the "qualified optional survivor annuity." Provided, however, the following rules apply in the specified circumstances:

# (i) Special Effective Date Rules.

(a) If the Plan permits retroactive annuity starting dates and a Participant elects a distribution with a retroactive annuity starting date (pursuant to Treasury Regulations Section 1.417(e)-1(b)(3)(iv)) that is before the aforementioned effective date, the date of the first actual payment of benefits based on the retroactive annuity starting date is substituted for the annuity starting date for purposes of applying the rules of this subsection.

(b) In the case of a plan that is subject to Code Section 401(a)(11) and that is maintained pursuant to one or more collective bargaining agreements between employee representatives and one or more employers ratified on or before August 17, 2006 (the date of enactment of PPA '06), the changes to Code Section 417 made by Section 1004 of PPA '06 apply to distributions with annuity starting dates during plan years beginning on or after the earlier of (i) January 1, 2008 or, if later, the date on which the last collective bargaining agreement related to the plan terminates (determined without regard to any extensions to a collective bargaining agreement made after August 17, 2006), or (ii) January 1, 2009.

- (ii) Definition of Qualified Optional Survivor Annuity. For purposes of this subsection (h), the term "qualified optional survivor annuity" means an immediate annuity:
  - (a) For the life of the Participant with a survivor annuity for the life of the Participant's spouse which is equal to the "applicable percentage" of the amount of the annuity which is payable during the joint lives of the Participant and the Participant's spouse, and
  - (b) Which is the actuarial equivalent of the normal form of benefit (or if this is a defined contribution plan, the amount of benefit which can be purchased with the Participant's vested Account).

Such term also includes any annuity in a form having the effect of an annuity described in the preceding sentence.

- (iii) For purposes of this subsection (h), the "applicable percentage" is based on the survivor annuity percentage (i.e., the percentage which the survivor annuity under the Plan's qualified joint and survivor annuity bears to the annuity payable during the joint lives of the Participant and the spouse). If the survivor annuity percentage is less than seventy-five percent (75%), then the "applicable percentage" is seventy-five percent (75%). If the survivor annuity percentage is equal to or greater than seventy-five percent (75%), the "applicable percentage" is fifty percent (50%)
- (iv) Inapplicability to Governmental Plans. Pursuant to the flush language of Code Section 401(a) and the provisions of Code Section 411(e)(1)(A), the provisions of Code Sections 401(a)(11) and 417, and consequently this subsection (h), shall not apply to this Plan if it is a governmental plan within the meaning of Code Section 414(d).
- 7.7 <u>Distribution in Kind.</u> If so indicated in the Adoption Agreement, in the case of a single sum distribution a Participant may direct the Trustee to cause distribution of a Participant's Matured Account to be made either in a form actually held in the Fund, or in cash by converting assets other than cash to cash, or in any combination of the two foregoing ways. Provided, however, such direction is permitted where it is reasonably possible for the Trustee to comply therewith and not in conflict with the terms of this Agreement and the interests of other Participants or their Beneficiaries.
- 7.8 <u>Facility of Payment.</u> In the case of incompetency or disability, either mental or physical, of a Participant or Beneficiary entitled to receive any distribution under the Plan, payments shall be made, if the Trustee shall be advised of the existence of such condition:
  - (a) to the duly-appointed attorney in fact, guardian or conservator, or to the legal representative of such Participant or Beneficiary; or
  - (b) to a person or institution entrusted with the care or maintenance of the incompetent or disabled Participant or Beneficiary, provided such person or institution has satisfied the Trustee that the payment will be used for the best interest and assist in the care of such Participant or Beneficiary, and provided further that no prior claim for said payment has been made by a duly appointed guardian or any legal representative of such Participant or Beneficiary.

Any payment made in accordance with the foregoing provisions of this Section shall constitute a complete discharge of any liability or obligation of the Trustee and the Plan therefor.

7.9 <u>Withdrawal from Voluntary Account.</u> If the Employer so indicates in the Adoption Agreement, Participants may make withdrawals from time to time from their Voluntary Account. All withdrawals from a Voluntary Account shall come first from the Nondeductible Voluntary Account, and only after the Nondeductible

Voluntary Account is exhausted will a withdrawal come from the Deductible Voluntary Account. No forfeitures will occur solely because a Participant makes a withdrawal from his Voluntary Account.

- 7.10 When a Participant or a Beneficiary Cannot Be Found. In the case of a benefit which is payable and the Participant or Beneficiary to whom the payment is due cannot be found, the Plan Administrator shall make reasonable efforts to locate such Participant or Beneficiary. In the event such Participant or Beneficiary cannot be located, the Trustee shall treat such amount as a forfeiture subject to Section 6.4. If a benefit is forfeited because a Participant or Beneficiary cannot be found such benefit will be reinstated if a claim is made by the Participant or Beneficiary within the period permitted for making a claim under applicable law.
- 7.11 <u>Determination of Required Distributions</u>. The Retirement Committee (or Employer if no Retirement Committee has been designated) shall be responsible for determining any required distribution dates under the Plan. The Retirement Committee shall cause the Trustee to make required distributions by such dates and in an amount satisfying any minimum distribution requirements applicable thereto. The Employer agrees to hold the Trustee harmless from any failure of the Retirement Committee to perform its duties under this Section.
- 7.12 <u>Rollovers.</u> This Section 7.12 applies to distributions made on or after January 1, 2002. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Administrator, to have any portion of an eligible rollover distribution paid directly to an eligible retirement plan specified by the distributee in a direct rollover.

## 7.12.1 <u>Definitions</u>.

(a) Eligible rollover distribution: An eligible rollover distribution is any distribution of all or any portion of the balance to the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under Code section 401(a)(9); the portion of any distribution that is not includable in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities); for distributions made on or after January 1, 1999, any hardship distributions; and any qualified disaster-relief distributions within the meaning of Code section 72(t)(2)(G).

A portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of after-tax employee contributions which are not includible in gross income. However, such portion may be transferred only to (1) an traditional individual retirement account or annuity described in section 408(a) or (b) of the Code ("traditional IRA") or a Roth individual retirement account or annuity described in Code Section 408A ("Roth IRA"); or; (2) for taxable years beginning after December 31, 2001 and before January 1, 2007, to a qualified trust which is part of a defined contribution, defined benefit or annuity plan described in Code Section 401(a) or 403(a), that agrees to separately account for amounts so transferred, including separately account for the portion of such distribution which is includible in gross income and the portion of such distribution which is includible; or (3) for taxable years beginning after December 31, 2006, to a qualified trust or to an annuity contract described in section 403(b), if such trust plan or contract provides for separate accounting for amounts so transferred (including earnings thereon), including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not so includible.

The Employer on a nondiscriminatory basis may elect in the Adoption Agreement or otherwise in a written addendum to this Plan to require that a distribution must be equal to at least \$200 before it qualifies as an eligible rollover distribution.

(b) Eligible retirement plan: An eligible retirement plan is an eligible plan under Code section 457(b) which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this plan, traditional IRA, a Roth IRAan individual retirement account described in Code section 408(a), an individual retirement annuity described in Code section 408(b), an annuity plan described in Code section 403(a), an annuity contract described in Code section 403(b), or a qualified defined benefit or defined contribution plan described in Code section 401(a), that accepts the distributee's eligible rollover distribution. Additionally, the definition of eligible retirement plan shall also apply in the case of a distribution to a surviving spouse, or to a spouse or former spouse who is the alternate payee under a qualified domestic relation order, as defined in section 414(p) of the Code.

If any portion of an eligible rollover distribution is attributable to payments or distributions from a designated Roth account, an eligible retirement plan with respect to such portion shall include only another designated Roth account of the individual from whose account the payments or distributions were made, or a Roth IRA of such individual.

Effective for distributions after December 31, 2007, Participants and Beneficiaries also shall be permitted to make a direct rollover of an eligible rollover distribution from this Plan to a Roth IRA described in Code section 408A ("Roth IRA"), subject to satisfying applicable requirements of Code section 408A, regulations—and IRS Notice 2008-30 and other guidance. Provided, however, for taxable years beginning before January 1, 2010, an individual cannot make a qualified rollover contribution from an eligible retirement plan other than a Roth IRA if, for the year the eligible rollover distribution is made, he or she has modified adjusted gross income exceeding \$100,000 or is married and files a separate return. For this purpose, the term "eligible rollover distribution" includes a rollover distribution described in subsection "a" above of after-tax employee contributions which are not includible in gross income, if applicable.

- (c) Distributee: A distributee includes an Employee or former Employee. In addition, the Employee's or former Employee's surviving spouse and the Employee's or former Employee's spouse or former spouse who is the alternate payee under a qualified domestic relations order, as defined in Code section 414(p), are distributees with regard to the interest of the spouse or former spouse. Effective for distributions on or after January 1, 2008, a distributee also includes the Participant's nonspouse Beneficiary designated under the Plan in accordance with Code section 401(a)(9)(E) and the regulations thereunder. In the case of a nonspouse Beneficiary, the direct rollover may be made only to a traditional IRA n individual retirement account or annuity under Code section 408(a) or 408(b) or Roth IRA that is established on behalf of the Beneficiary for purposes of receiving the distributionas an inherited IRA pursuant to Code section 402(c)(11) and that will be treated as an inherited IRA pursuant to provisions of Code Section 402(c)(11). Also, in this case, the determination of any required minimum distribution under Code section 401(a)(9) that is ineligible for rollover shall be made in accordance with IRS Notice 2007-7, Q&A 17 and 18. Following are additional requirements of nonspouse beneficiary rollovers:
  - (i) Applicability of Certain Code Requirements. For Plan Years beginning on or after January 1, 2010, and to the extent the particular Code section is applicable to the Plan, any direct rollover of a distribution by a nonspouse beneficiary shall be subject to:
    - (a) The direct rollover requirements of Code Section 401(a)(31) (including Code Section 401(a)(31)(B)),
    - (b) The notice requirements of Code Section 402(f) and
    - (c) The mandatory withholding requirements of Code Section 3405(c).

Before that date, any such distribution shall not be subject to said requirements. Any distribution from the Plan to a non-spouse beneficiary shall not be eligible for a 60-day (non-direct) rollover.

- (ii) Trust Beneficiary. Subject to the subsection 7.12.1(c) above, if the Participant's named beneficiary is a trust, the Plan may make a direct rollover to an IRA on behalf of the trust, provided the trust satisfies the requirements to be a designated beneficiary within the meaning of Code Section 401(a)(9)(E).
- distribution, as determined under applicable Treasury Regulations and other Internal Revenue Service guidance. If the Participant dies before his or her required beginning date and the non-spouse beneficiary rolls over to an IRA the maximum amount eligible for rollover, the beneficiary may elect to use either the 5-year rule or the life expectancy rule, pursuant to Treasury Regulations Section 1.401(a)(9)-3, A-4(c), in determining the required minimum distributions from the IRA that receives the non-spouse beneficiary's distribution.
- (d) Direct rollover: A direct rollover is a payment by the plan to the eligible retirement plan specified by the distributee.

### **SECTION 8**

## **Inalienability of Benefits**

No benefit or interest available hereunder will be subject to assignment or alienation, either voluntarily or involuntarily except as provided under Code section 401(a)(13). The preceding sentence shall also apply to the creation, assignment or recognition of a right to any benefit payable with respect to a Participant pursuant to a domestic relations order, unless such order is determined to be a qualified domestic relations order, as defined in Code section 414(p) ("QDRO"), or any domestic relations order entered before January 1, 1985. This Plan specifically permits distribution to an alternate payee under a qualified domestic relations order QDRO at any time, if the Participant consents in writing thereto, irrespective of whether the Participant has attained his earliest retirement age (as defined in Section 414(p) of the Code) under the Plan. The Plan shall not recognize any domestic relations order which alters or changes benefits, provides for a form of benefit not otherwise permitted under the Plan, increases benefits not otherwise provided by the Plan or accelerates or defers the time of payment of benefits. No Participant or Beneficiary shall have authority to any specific portion of the assets of the Plan. The City or Retirement Committee may require releases from any person as a condition to complying with any such order. Effective on or after April 6, 2007, a domestic relations order that otherwise satisfies the requirements for a QDRO will not fail to be a QDRO: (i) solely because the order is issued after, or revises, another domestic relations order or QDRO; or (ii) solely because of the time at which the order is issued, including issuance after the annuity starting date or after the Participant's death. A domestic relations order described in the immediately preceding sentence shall be subject to the same requirements and protections that apply to any other QDRO.

### **SECTION 9**

# **Amendment and Termination**

9.1 <u>Amendment.</u> The Employer reserves the right to amend this <u>Plan from time to time, including without limitation,</u> the designations and elections made by it under the Adoption Agreement from time to time by adopting and executing a new Adoption Agreement, which shall be delivered to the Trustee. The Employer further

reserves the right to amend its retirement plan in its entirety by the adoption of a successor retirement plan in place of the Plan set forth herein and in the Adoption Agreement, and by entering into such agreement with the Trustee or with the successor trustee, successor trustees, or other successor funding medium selected by the Employer as may be required for the purpose of carrying such successor retirement plan into effect; provided, however, that no such amendment shall be effective so as to increase the duties of the Trustee without its consent and provided further that the right of the Employer to designate a successor retirement plan or funding medium shall be subject to the notice requirements affecting the removal of the Trustee set forth in Section 10.4 hereof.

Unless otherwise allowed or provided by applicable law, regulations or other guidance, nNo amendment shall be effective to reduce or divest the Total Account of any Participant without his consent unless the same shall have been adopted with the consent of the Secretary of Labor pursuant to the applicable provisions of the Employee Retirement Income Security Act of 1974, to the extent applicable to this Plan, or in order to comply with the provisions of the Internal Revenue Code of the United States and the regulations and rulings thereunder affecting the tax-qualified status of the Plan or to comply with the provisions of any stabilization law, regulations, orders or directives that may now or hereafter be in force, or to comply with any state law regarding retirement plans for municipal employees. Effective after September 6, 2000, an amendment may eliminate or restrict an optional form of benefit if the amendment satisfies the following:

- (a) The amendment provides a single-sum distribution form that is otherwise identical to the optional form of benefit eliminated or restricted. For purposes of this condition (a), a single-sum distribution form is otherwise identical only if it is identical in all respects to the eliminated or restricted optional form of benefit (or would be identical except that it provides greater rights to the participant) except with respect to the timing of payments after commencement.
- (b) The amendment is not effective unless the amendment provides that the amendment shall not apply to any distribution with an annuity starting date earlier than the earlier of: (i) the 90<sup>th</sup> day after the date the Participant receiving the distribution has been furnished a summary that reflects the amendment and that satisfies the ERISA requirements at 29 CFR 2520.104b-3 relating to a summary of material modifications or (ii) the first day of the second Plan Year following the Plan Year in which the amendment is adopted.

Effective after September 6, 2000, a $\Delta$ n amendment that meets the requirements of Treasury Regulation section 1.411(d)-3 or 1.411(d)-41.411(d) Q&A 2 and Q&A 3, to the extent applicable to the Plan, may be adopted.

Notwithstanding anything in this Plan to the contrary, pursuant to Code section 411(e)(1)(A) and 412(e)(2), requirements of Code Section 411 or 412, and corresponding provisions of this Plan, shall not apply if this Plan is a governmental plan within the meaning of Code Section 414(d), with the exception of Code section 411(e)(2).

- 9.2 <u>Discontinuance of Contributions and Termination of Plan</u>. The Employer also reserves the right, by action of its City Council, to reduce, suspend, or discontinue its contributions to this Plan and to terminate the Plan in its entirety.
- 9.3 <u>Limitations</u>. No power of amendment or of termination may be exercised so as to discriminate in favor of City officials or Highly Compensated Employees, or to permit any part of the Fund to be used for or diverted to purposes other than for the exclusive benefit of Participants or their Beneficiaries prior to the satisfaction of all liabilities with respect to such Participants and their Beneficiaries under this Plan.
- 9.4 Merger, Etc., with Another Plan. In the case of merger or consolidation of this Plan with, or transfer of assets and liabilities of this Plan to any other plan, each Participant shall (if such other plan then terminated) receive a benefit immediately after the merger, consolidation, or transfer which is not less than the benefit he would have been entitled to receive immediately before the merger, consolidation or transfer (if this Plan had then terminated).

**SECTION 10** 

### Concerning the Trustee

### 10.1 Dealings with the Trustee.

- (a) No person, firm or corporation dealing with the Trustee shall be required to make inquiry as to the authority of the Trustee to do any act which the Trustee shall do hereunder, and any such person, firm or corporation shall be entitled to assume conclusively that the Trustee is properly authorized to do any act which it shall do hereunder, and any such person, firm or corporation shall be under no liability to anyone whomsoever for any act done hereunder pursuant to the written direction of the Trustee.
- (b) Any such person, firm or corporation may conclusively assume that the Trustee has full power and authority to receive and receipt for any money or property becoming due and payable to the Trustee, and no such person shall be bound to inquire as to the disposition or application of any money or property paid to the Trustee or paid in accordance with the written directions of the Trustee.
- (c) No person, firm or corporation dealing with the Trustee shall be required to see either to the administration of the Plan or to the faithful performance by the Trustee of its duties hereunder (except to the extent required by provisions of the Employee Retirement Income Security Act of 1974, if applicable to this Plan).
- 10.2 <u>Fees and Expenses from Fund.</u> Except as provided in Section 10.3, the Trustee may pay all expenses, fees, costs and other charges reasonably incurred in the establishment, maintenance, continued qualification, reporting and disclosure, trusteeship, investment, administration and termination (including, but not limited to, legal, actuarial, accounting, and other professional fees) of the Plan from the Fund unless the Employer pays the same. In the event such items shall be paid from and out of the Fund, the Trustee, in a fair and equitable manner of its selection and depending upon the particular kind or type of service for which the expense has been incurred, shall allocate said charge to and pay it out of:
  - (a) the annual Employer contribution to the Trust, or
  - (b) the income of the Fund, or
  - (c) the principal of the fund, including any accumulations of income that have been added thereto, or
  - (d) to or out of any combination of the foregoing sources in the event the service in question has been for the benefit, protection or administration of more than one such source of payment.

The Trustee's determination in such respect made in good faith of the amount so to be allocated and charged to each such source of payment shall be binding and conclusive upon all persons interested or becoming interested in the Plan or the Fund. Each such charge of the Trustee shall be a lien upon the Fund, and, ratably, in accordance with the method of allocation used as aforesaid, shall be a lien upon the interest of Participants in the source of payment to which the same is charged until the same is paid and discharged in full.

10.3 Fire Plans. The City and the Retirement Committee shall develop a schedule of investment costs relating the investment of funds in each of the accounts in the Fund, which costs shall be paid out of the funds in such accounts or assessed to the Participants as provided in such schedule. The schedule of investment costs shall provide for the allocation of the administrative or record-keeping costs of the various investment options available to the Participants of the Plan and shall assess such costs so that each Participant pays a pro rata share of the costs based upon his or her choice of options and number of transfers among options. The costs of the actuarial evaluation of the plan, as well as all other costs related to the general operation of the Plan and not allocated pursuant to the schedule of investment costs shall be considered administrative costs and shall be paid by the City from the Unallocated Employer Account.

### 10.4 Resignation and Removal of Trustee.

- The Trustee may resign by giving the Employer thirty (30) days (or such shorter period (a) as the Employer may approve in writing) written notice of its resignation by registered mail, such notice period to commence upon mailing thereof. The Employer shall thereupon appoint a successor trustee, successor trustees or other successor funding medium to assume the rights, powers and duties of the Trustee and shall promptly give the Trustee written notice by registered mail of the appointment of such successor funding medium, provided that such notice, to be effective, must be received by the Trustee not later than a date which is sixty (60) days from and after the date on which its notice of resignation was mailed to the Employer. The Trustee shall forthwith deliver to the successor funding medium and as soon as possible thereafter account to the successor funding medium for each and every Fund asset and any and all records of the Fund that are in its possession or control. Notwithstanding any of the foregoing, however, if the Trustee shall not have received written notice of the appointment of a successor funding medium within sixty (60) days after the mailing of its notice of resignation, all as hereinbefore provided, the Employer's plan set forth herein and in the Adoption Agreement upon written approval of the Employer shall terminate in its entirety, effective immediately upon the expiration of such sixty (60) day period, and the Trustee shall thereupon proceed to make distribution of the Fund assets to the Participants entitled thereto.
- (b) The Employer may remove the Trustee by giving the Trustee thirty (30) days (or such shorter period as the Trustee may approve in writing) written notice of his resignation by registered mail, such notice period to commence upon the receipt thereof by the Trustee, and which written notice shall identify the successor trustee, successor trustees or other successor funding medium appointed by the Employer to assume the rights, powers and duties of the Trustee. The Trustee shall forthwith deliver to the successor funding medium and as soon as possible thereafter account to the successor funding medium for each and every Fund asset and all records of the Fund that are in its possession or control.

#### 10.5 Accountings by Trustee.

- (a) The Trustee shall render to the Employer an annual account and report as soon as practicable after the Annual Valuation Date in each year showing all transactions affecting the administration of the Plan and the Fund, including, but not necessarily limited to, such information concerning the Plan and the Fund and the administration thereof by the Trustee as shall be requested in writing by the Employer.
- (b) The Trustee shall also render such further reports from time to time as may be requested by the Employer and shall submit its final report and account to the Employer when it shall cease to be Trustee hereunder, whether by resignation or other cause.
- (c) After giving Participants and other persons interested therein a reasonable opportunity to examine the annual account of the Trustee to the Employer as provided in (a) above, provided that no exceptions are asserted thereto by any person (including the Employer) interested therein, the Employer may settle and allow such accounts by agreement with the Trustee. Except as may be otherwise required by the Employee Retirement Income Security Act of 1974, if applicable, the Trustee shall upon such settlement and allowance be released and relieved of all liability for all matters set forth herein.
- 10.6 Trustee's Power to Protect Itself on Account of Taxes. The Trustee, as a condition to the making of distribution of a Participant's Matured Account during his lifetime, may require the Participant, or, in the event of his death, may require the person or persons entitled to receive his Matured Account in such event to furnish the Trustee with proof of payment of all income, inheritance, estate, transfer, legacy, and/or succession taxes, and all other taxes of any different type or kind that may be imposed under or by virtue of any state or federal statute or law upon the payment, transfer, descent or distribution of such Matured Account and for the payment of which the Trustee may, in its judgment, be directly or indirectly liable. In lieu of the foregoing, the Trustee may deduct, withhold and transmit to the proper taxing authorities any such tax which it may be permitted or required to deduct and withhold and the Matured Account to be distributed in such case shall be correspondingly reduced.
- 10.7 <u>Fiduciary Duties</u>. The Trustee and each fiduciary hereunder, in the exercise of each and every power or discretion vested in them by the provisions of this Agreement, shall be governed by the principle that no

discrimination in favor of highly compensated Employees who are Participants from time to time hereunder shall result and shall (subject to the provisions of the Employee Retirement Income Security Act of 1974 to the extent applicable to this Plan) discharge their duties with respect to the Plan solely in the interest of the Participants and Beneficiaries and

- (a) for the exclusive purposes of:
  - (i) providing benefits to Participants and their Beneficiaries; and,
  - (ii) defraying reasonable expenses of administering the Plan;
- (b) with the care, skill, prudence and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims;
- (c) by diversifying the investments of the Plan (according to the policies established by the Nebraska Investment Council for Police and Fire Plans) so as to minimize the risk of large losses, unless under the circumstances it is clearly prudent not to do so; and,
- (d) in accordance with the Plan, insofar as the Plan is consistent with the provisions of the Employee Retirement Income Security Act of 1974, to the extent the same are applicable to this Plan.

Notwithstanding anything in this Agreement to the contrary, any provision hereof which purports to relieve a fiduciary from responsibility or liability for any responsibility, obligation or duty under Part 4 of Subtitle B of Title I of the Employee Retirement Income Security Act of 1974, to the extent Title I is applicable to this Plan, shall, to the extent the same is inconsistent with said Part 4, be deemed void.

- 10.8 <u>Prohibited Transactions</u>. Except as may be expressly permitted by law, neither the Trustee nor any other fiduciary hereunder shall permit the Plan to engage, directly and indirectly, in any of the following transactions with a disqualified person (as defined in Section 4975 of the Internal Revenue Code of 1986):
  - (a) sale or exchange, or leasing, of any property between the Plan and a disqualified person;
  - (b) lending of money or other extension of credit between the Plan and a disqualified person;
  - (c) furnishing of goods, services or facilities between the Plan and a disqualified person;
  - (d) transfer to, or use by or for the benefit of, a disqualified person of the income or assets of the Plan:
  - (e) act by a disqualified person who is a fiduciary whereby he deals with the income or assets of the Plan in his own interest or for his own account; or
  - (f) receipt of any consideration for his own personal account by any disqualified person who is a fiduciary from any party dealing with the Plan in connection with a transaction involving the income or assets of the Plan.
- 10.9 <u>Indemnity</u>. The Trustee shall be indemnified and held harmless by the Employer from any and all liabilities, costs and expenses (including legal expenses) arising out of any action taken by the Trustee as Trustee, fiduciary or in any other capacity with respect to this Plan, whether imposed under the Employee Retirement Income Security Act of 1974, or otherwise, except for the Trustee's negligent acts or omissions.
- 10.10 <u>Investment in Insurance</u>. If the Employer shall so designate in the Adoption Agreement, a Participant may, with the consent of the Employer and subject to such conditions as the Employer may impose, elect to have a portion of his Vested Total Account, excluding the Participant's Deductible Voluntary Account, invested in

life insurance contracts (any said insurance contract being referred to as a "contract") issued by any insurance company licensed to do business in any state subject to the following rules and conditions:

- (a) <u>Ordinary Life</u> For purposes of these incidental insurance provisions, ordinary life insurance contracts are contracts with both nondecreasing death benefits and non-increasing premiums. If such contracts are purchased, less than one-half of the aggregate Employer Contributions allocated to any Participant will be used to pay the premiums attributable.
- (b) <u>Term and Universal Life</u> No more than one-fourth of the aggregate Employer Contributions allocated to any Participant will be used to pay the premium on term life insurance contracts, universal life insurance contracts, and all other life insurance contracts which are not ordinary life.
- (c) <u>Combination</u> The sum of one-half of the ordinary life insurance premiums and all other life insurance premiums will not exceed one-fourth of the aggregate Employer Contributions allocated to any Participant.
- (d) The Participant shall take such physical examinations and furnish such information as may be necessary to procure a contract.
  - (e) All contracts shall have a uniform premium due date.
- (f) The Trustee shall be the owner of all contracts with full power to execute all insurance applications and to exercise all available options, and shall be the death beneficiary thereunder.
- (g) All amounts used to purchase term life insurance, to fund "P.S. 58" costs or to acquire any other non-cash value benefits under this Section 10.10 shall be deemed to come from Employer Accounts subject to the limits specified in paragraph (a) above. If the Participant's Employer Account is insufficient within the limitations herein contained to pay any premium on a contract when the same becomes due, the Trustee shall, unless the Participant pays to the Trustee a sum sufficient to pay such premium (any such payment being deemed a voluntary contribution hereunder), cause such contract to be rewritten for its then paid-up value, if any, and retain the same for the Participant, in which event no further premium payments shall thereafter be made thereon.
- (h) All dividends on a contract shall be used to reduce premiums. Any dividends earned on insurance contracts in excess of the amount of the current premium will be allocated to the Participant's Account for whose benefit the contract is held.
- (i) Any charge or expense of the Trustee in handling a contract shall be paid by the Employer but, if not so paid, shall be a charge against the Fund; provided that the Employer may, in its discretion, direct that any such charge or expense be deducted from the Participant's Total Account.
- (j) Any insurance company issuing contracts may deal with the Trustee alone and without the consent of any Participant or beneficiary and shall not be required to examine the provisions of the Plan, nor shall it be responsible for the failure of the Trustee to perform its duties, nor shall it be obligated to see to the application or disposition of any money paid by it to the Trustee and any such payment shall fully discharge such insurance company for the amount so paid.
- (k) For the purpose of determining the value of a contract hereunder, such contract shall be valued at the greater of the premiums theretofore paid thereon or its then cash value, but such contract shall not be considered a part of the Fund for the purpose of allocating income, market gains and losses of the Fund in accordance with Section 4 hereof.
- (l) On maturity of the Participant's Account by reason of the death of the Participant, the proceeds of any contract shall be deemed a death benefit under the Plan and shall be distributed to his Beneficiary or Beneficiaries in the manner prescribed in Section 7 hereof.

- (m) On maturity of the Participant's Accounts for any reason other than death of the Participant, the Trustee shall surrender the contract for cash or an annuity and distribute the proceeds in the manner described in Section 7 hereof, distribute the contract to the Participant, or any combination of the foregoing.
- (n) For the purpose of the valuation and adjustment of accounts by the Trustee and the allocation of the net income of the Fund thereto as provided in Section 4 hereof, no part of the cash value of a contract purchased hereunder shall be considered in determining the value of a Participant's Total Account.
- (o) The Trustee shall apply for and will be the owner of any insurance contract purchased under the terms of this Plan. The insurance contract(s) must provide that proceeds will be payable to the Trustee, however, the Trustee shall be required to pay over all proceeds of the contract(s) to the Participant's designated beneficiary in accordance with the distribution provisions of this Plan. Under no circumstances shall the Trust retain any part of the proceeds. In the event of any conflict between the terms of this Plan and the terms of any insurance contract purchased hereunder, the Plan provisions shall control.

#### 10.11 Directed Investments.

- 10.11.1 Employer Direction. If so indicated in the Adoption Agreement, the Trustee shall be subject in the management and control of the Fund to the directions (to the extent not inconsistent with law) of the person or committee identified in the Adoption Agreement or certified to the Trustee by an official of the Employer. Such direction shall be subject to such restrictions as the Trustee may impose. The Trustee in acting pursuant to and in reliance on such directions shall be fully and completely indemnified and held harmless by the Employer from any liability, loss or expense (including legal fees) arising out of its actions so directed notwithstanding that such directions, and the Trustee's conduct pursuant thereto, may constitute a breach of fiduciary obligations to the Plan, the Participants and Beneficiaries.
- 10.11.2 Participant Direction of Investments. If the Employer shall so indicate in the Adoption Agreement, each Participant may individually direct the Trustee to segregate his Account and instruct the Trustee regarding the investment of his Account and the Trustee shall be bound by such direction and relieved of all liability for loss resulting from the exercise of such individual direction. Such direction shall be subject to such restrictions as the Plan or the Trustee may impose. Any additional fee agreed upon from time to time by the Employer and the Trustee as compensation for the right of individual direction herein granted shall be charged against the Accounts of Participants exercising such right unless paid by the Employer. The Employer may make this right of individual direction subject to such other restrictions as are deemed appropriate according to rules of uniform nondiscriminatory application. Accounts under individual direction shall not share in the allocation of income and market gains and losses as provided for the general fund in Section 4 hereof. At no time shall a Participant direct the Trustee to invest in a work of art, rug, antique, metal, gem, stamp, coin, alcoholic beverage or any other item or class of items defined as a collectible by the Secretary of Treasury.

#### 10.12 Investment Managers.

- (a) <u>Police Plan</u>. The City or Retirement Committee may select and contract with investment managers registered under the Investment Advisers Act of 1940 to invest, reinvest, and otherwise manage such portion of the assets of the Plan as may be assigned by the City or Committee.
- (b) <u>Fire Plan</u>. The Retirement Committee, subject to the approval of the City, may select an investment manager. The City, subject to approval of the Retirement Committee, may contract with investment managers registered under the Investment Advisers Act of 1940 to invest, reinvest, and otherwise manage such portion of the assets of the Plan as may be assigned by the City or Retirement Committee.
- (c) <u>All Other Plans</u>. The Employer reserves the power to appoint from time to time one or more Investment Managers, who may be a person, firm, association or corporation, to direct the Trustee or

to assume the duties of the Trustee in the investment of all or any portion of the Fund. An Investment Manager shall be any person or firm which is either (1) registered as investment adviser under the Investment Advisers Act of 1940, (2) a bank, or (3) an insurance company which is qualified to perform the services of an Investment Manager under the laws of more than one state. An Investment Manager shall have such rights and responsibilities only with regard to that portion of the Fund designated by the Employer.

- All Plans. The Employer or Retirement Committee, as the case may be, may remove any such Investment Manager and shall have power to appoint a successor or successors from time to time in succession to any Investment Manager who shall be removed, die, resign or otherwise cease to serve hereunder. The Trustee shall follow and comply with all investment directions given to the Trustee by such Investment Manager with respect to the designated portion of the Fund, and the Trustee shall be released, indemnified and held harmless for all actions taken, or things done or omitted to be done by such Investment Manager or by the Trustee in the investment or reinvestment of the Fund pursuant to and in accordance with the directions of the Investment Manager. The reasonable fees and expenses of the Investment Manager, as agreed upon in writing by the Employer and the Investment Manager, shall be an expense chargeable to the Fund and the income derived therefrom, and shall be paid therefrom by the Trustee in such shares as between income and principal as the Trustee deems reasonable and proper, or the Employer, in its discretion, may pay the amount of such fees and expenses directly to the Investment Manager. The appointment of an Investment Manager, original or successor, by the Employer shall be made by resolution adopted by the City or Retirement Committee, as the case may be. Each such appointment shall become effective upon delivery to the Trustee of a certified copy of such resolution and a written acceptance of such appointment signed by the Investment Manager, acknowledging that it is a "fiduciary" with respect to the Plan, as defined in the Employee Retirement Income Security Act of 1974; even though ERISA is not applicable to this Plan if it is a governmental plan within the meaning of Code section 414(d), the ERISA definition of "fiduciary" shall be used. Such resolution shall state what portion of the Fund shall be the investment responsibility of the Investment Manager. The removal of an Investment Manager shall be and become effective upon receipt by the Trustee of a certified copy of the resolution of the City or Retirement Committee, as the case may be, removing such adviser, accompanied by a written statement signed by an official of the City or Retirement Committee that notice of such removal has been given to such Investment Manager. The resignation of an Investment Manager shall be effective (if not, by its terms, made effective at a later date) upon receipt by the Trustee of such resignation in writing signed by the Investment Manager, and unless such resignation states on its face that notice thereof has been given to the Employer, the Trustee shall notify the Employer in writing forthwith of such resignation. Whenever and for as long as there shall be no Investment Manager appointed or acting hereunder, the powers of the Investment Manager shall be exercised by the Trustee. The powers of the Investment Manager shall be exercised by the Trustee with respect to any portion of the Fund over which the Investment Manager has not been given investment authority. No Investment Manager at any time serving hereunder shall be or become liable for the acts or defaults of any prior Investment Manager, for the acts or defaults of another Investment Manager who has investment responsibility for a separate portion of the Fund, or for the acts or defaults of the Trustee. Neither the Employer, Retirement Committee, nor any member of their governing boards shall be or become liable for the acts or omissions of any Investment Manager appointed pursuant to this Section (except to the extent that liability is imposed under the Employee Retirement Income Security Act of 1974, if such Act is applicable to this Plan).
- 10.13 <u>Participant Loans</u>. If the Employer so indicates in the Adoption Agreement that Participant loans will be allowed, effective for Participant loans made or renewed on and after the last day of the 1989 Plan Year, the Employer shall establish a Participant loan policy, separate from this Plan which shall comply with the requirements of ERISA and the Code and the Regulations issued thereunder, as amended from time to time, to the extent such requirements and Regulations are applicable to this Plan. Such loan policy shall be subject to the following:
  - (a) Loans shall be made available to all Participants and beneficiaries on a reasonably equivalent basis.
  - (b) Loans shall not be made available to Highly Compensated Employees (as defined in Section 414(g) of the Code) in an amount greater than the amount made available to other Employees.

- (c) Loans must be adequately secured and bear a reasonable interest rate.
- (d) No Participant loan shall exceed the present value of the Participant's vested accrued benefit.
- (e) In the event of a default, foreclosure on the note and attachment of security will not occur until a distributable event occurs in the Plan.
- (f) Loan repayments will be suspended under this Plan as permitted under Code section 414(u)(4).

Notwithstanding any other provision of this Plan, the portion of the Participant's vested Account balance used as a security interest held by the Plan by reason of a loan outstanding to the Participant shall be taken into account for purposes of determining the amount of the Account balance payable at the time of death or distribution, but only if the reduction is used as repayment of the loan. If less than 100% of the Participant's vested Account balance (determined without regard to the preceding sentence) is payable to the surviving spouse, then the Account balance shall be adjusted by first reducing the vested Account balance by the amount of the security used as repayment of the loan, and then determining the benefit payable to the surviving spouse.

The Plan Administrator should advise any Participant or Beneficiary that aA loan, which when added to the outstanding balance of all other loans to the Participant or beneficiary, would exceed the lesser of (a) \$50,000.00, reduced by the excess (if any) of the highest outstanding balance of loans during the one year period ending on the day before the loan is made, over the outstanding balance of loans from the Plan on the date the loan is made, or (b) one-half of the present value of the Participant's vested accrued benefit, or if greater, his total accrued benefit up to \$10,000.00 will constitute a taxable distribution to the Participant to the extent such loan exceeds these limits pursuant to Section 72(p) of the Internal Revenue Code. For the purpose of the preceding sentence, all loans from all plans of the Employer are aggregated. The Plan Administrator should also advise any Participant or beneficiary that, uUnless the express terms of the loan require repayment (principal and interest) is amortized in level payments, not less frequently than quarterly, over a period not extending beyond five years from the date of the loan (unless such loan is used to acquire dwelling unit which within a reasonable time (determined at the time the loan is made) will be used as the principal residence of the Participant, such loan shall constitute a taxable distribution to the Participant or beneficiary, unless otherwise provided by applicable law. In addition, the Plan Administrator should advise any Participant or beneficiary that the loan must be evidenced by a legally enforceable agreement and the terms of the agreement must demonstrate compliance with the requirements of Section 72(p)(2) and Regulation 1.72(p)-1.

An assignment or pledge of any portion of the Participant's interest in the Plan and a loan, pledge, or assignment with respect to any insurance contract purchased under the Plan, will be treated as a loan under this Section 10.13.

Any loan which was in existence on August 13, 1982, shall be subject to the conditions and terms contained in said loan then in effect until the date of maturity. Any extension, renewal, renegotiation or revision of a loan after August 13, 1982, shall be considered a new loan.

- 10.14 <u>Other Powers</u>. In extension, but not in limitation, of the rights, powers and discretions conferred upon the Trustee herein, the Trustee shall have and may exercise from time to time in the management and custody of the assets of the Fund and, for the purpose of distribution after the termination thereof, and, for the purpose of distribution of Matured Accounts, without order or license of any court, any one or more or all of the following rights, powers and discretions:
  - (a) To invest and reinvest the assets of the Fund with the care, skill prudence and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of like character and with like aims (and to the extent possible consistently with the most recent funding policy and method adopted by the Employer and communicated to the Trustee) without limitation of any statute, rule of law, or regulation of any governmental body prescribing or limiting the investment of trust assets by corporate or individual trustees, in or to certain kinds, types, or classes of investments or prescribing the portion of the Fund which may be

invested in any one property or kind, type, or class of investment; limited, however, to the extent that assets of the Fund are required to be invested pursuant to policies established by the Nebraska Investment Council. Specifically and without limiting the generality of the foregoing, the Trustee may invest and reinvest principal and accumulated income of the Fund in preferred and common stocks of any kind or class of any corporation, including but not limited to investment and small business investment companies of all types; voting trust certificates; interests in investment trusts; shares of mutual funds; interests in a common trust, variable demand note or other type of pooled or collective fund operated by the Trustee; bonds, notes and debentures, secured or unsecured; mortgages on real or personal property; covered call options; deposits in a commercial or savings bank or a savings and loan association; insurance contracts on the life of any "keyman" or shareholder of the Employer; conditional sales contracts; real estate and leases, unless the Plan restricts investments according to policies established by the Nebraska Investment Council. Unless otherwise provided by policies established by the Nebraska Investment Council applicable to this particular Plan, investment of the entire Fund in common stocks shall be deemed appropriate at any phase of the economic business cycle, but it is not, however, the purpose hereof to direct that the Fund shall be invested either entirely or to any extent whatsoever in such common stocks.

- (b) To sell, exchange or otherwise dispose of any asset of whatsoever character at any time held by the Trustee in trust hereunder.
- (c) To segregate any part or portion of the Fund for the purpose of administration or distribution thereof and, in its sole discretion, to hold the Fund uninvested whenever and for so long as, in the Trustee's discretion, the same is likely to be required for the payment in cash of Matured Accounts or of Total Accounts normally expected to mature in the near future, or whenever, and for as long as, market conditions are uncertain, or for any other reason which, in the Trustee's discretion, requires such action or makes such action advisable.
- (d) To retain and employ such attorneys, agents and servants as may be necessary or desirable, in the opinion of the Trustee, in the administration of the Fund, and to pay them such reasonable compensation for their services as may be agreed upon as an expense of administration of the Fund (which may be paid from the principal of the Fund, current income or current forfeitures prior to reallocation or reduction of the Employer's contribution), including power to employ and retain counsel upon any matter of doubt as to the meaning of or interpretation to be placed upon this Agreement or any provisions thereof with reference to any question arising in the administration of the Fund or pertaining to the distribution thereof or pertaining to the rights and liabilities of the Trustee hereunder or to the rights and claims of Participants and Beneficiaries, and the Trustee, in any such event, may act in reliance upon the advice, opinions, records, statements and computations of any attorneys and agents and on the records, statements, and computations of any servants so selected by it in good faith and shall be released and exonerated of and from all liability to anyone in so doing (except to the extent that liability is imposed under the Employee Retirement Income Security Act of 1974 and relevant provisions of such Act are applicable to this Plan).
- (e) To institute, prosecute, and maintain, or to defend, any proceeding at law or in equity concerning the Plan or Fund or the assets thereof or any claims thereto, or the interests of Participants and Beneficiaries hereunder at the sole cost and expense of the Fund and/or at the sole cost and expense of the Total Account of the Participant that may be concerned therein or that may be affected thereby as, in the Trustee's opinion, shall be fair and equitable in each case, and to compromise, settle and adjust all claims and liabilities asserted by or against the Plan or Fund or asserted by or against the Trustee, on such terms as the Trustee, in each such case, shall deem reasonable and proper, but the Trustee shall be under no duty or obligation to institute, prosecute, maintain, or defend any suit, action, or other legal proceedings unless it shall be indemnified to its satisfaction against all expenses and liabilities which it may sustain or anticipate by reason thereof.
- (f) To institute, participate in and join in any plan of reorganization, readjustment, merger, or consolidation with respect to the issuer of any securities held by the Trustee hereunder, and to use any other means of protecting and dealing with any of the assets of the Fund which it believes reasonably necessary or proper and, in general, to exercise each and every other power or right with respect to each asset or investment held by it hereunder as individuals generally have and enjoy with respect to their own assets and investments, including power to vote upon any securities or other assets having voting power which it may

hold from time to time, and to give proxies with respect thereto, with or without power of substitution or revocation, and to deposit assets or investments with any protective committee, or with trustees or depositories designated by any such committee or by any such trustees or any court.

- (g) In any matter of doubt affecting the meaning, purpose or intent of any provision of this Agreement, to determine such meaning, purpose of intent; and the determination of the Trustee in any such respect shall be binding and conclusive upon all persons interested or who may become interested in the Plan or the Fund.
- (h) To require, as a condition to distribution of any Matured Account, proof of identity or of authority of the person entitled to receive the same, including power to require reasonable indemnification on that account as a condition precedent to its obligation to make distribution hereunder.
- (i) To collect, receive, receipt and give acquittance for all payments that may be or become due and payable on account of any asset in trust hereunder which has not, by act of the Trustee taken pursuant thereto, been made payable to others, and payment thereof by the company issuing the same, or by the party obligated thereon, as the case may be, when made to the Trustee hereunder or to any person or persons designated by the Trustee, shall acquit, release and discharge such company or obligated party from any and all liability on account thereof.
- (j) To determine from time to time, as required for the purpose of distribution or for the purpose of allocating Fund income or for any other purpose of the Plan, the then value of the Fund and of the Accounts in the Fund, the Trustee, in each such case, using and employing for that purpose the fair market value of each of the assets constituting the Fund. Each such determination so made by the Trustee in good faith shall be binding and conclusive upon all persons interested or becoming interested in the Plan or the Fund.
- (k) To receive and retain Employer contributions in a form other than cash in the form in which the same are received until such time as the Trustee, in its sole discretion, deems it advisable to sell or otherwise dispose of such assets; to carry all investments of the Fund or any part thereof in its own name or in the name of any nominee selected by it, without designation of the trust capacity in which the same is held, but with the same liability for any act or default of any such nominee as for its own act or default.
- (l) Except to the extent otherwise provided in Section 4.4 or herein, to co-mingle, for investment purposes, the assets of the Fund with the assets of any other qualified retirement plan fund of the Employer, provided that the records of the Trustee shall reflect the relative interests of the separate trusts in such commingled fund.
- (m) To grant an option or options for the sale or other disposition of a Fund asset, including the issuance of options for the purchase of common stock held by the Fund in return for the receipt of a premium from the optionee (it being expressly intended that said options may be in form in terms to permit their being freely traded on an option exchange) and including the repurchase of any such option granted, or in lieu thereof, the repurchase of an option identical in terms to the one issued.
- (n) To have and to exercise such other and additional powers as may be advisable or proper in its opinion for the effective, economical and equitable administration of the Fund.
- (o) If so provided in the Adoption Agreement, one (1) or more declarations of trust executed by the Trustee (or by banks or trust companies affiliated in ownership with the Trustee) shall be incorporated by reference into this Agreement and notwithstanding any other provision of the Agreement to the contrary, the Trustee may cause all or any part of the Fund, without limitation as to amount, to be commingled with the money of trusts created by others by causing such money to be invested as a part of any or all of the funds created by said declarations of trust and the Fund so added to any of said funds shall be subject to all of the provisions of said declarations of trust as the same may be amended from time to time.

(p) If the Employer has so indicated in the Adoption Agreement, the Trustee is specifically authorized and empowered to invest Plan assets in deposit accounts or securities offered by the Trustee, its affiliates or other designated financial or securities institutions, unless the Plan is subject to the policies of the Nebraska Investment Council and such investments are contrary to such policies.

#### SECTION 11

#### Certifications - Rules and Regulations

- 11.1 <u>Certificates of Fact by Employer</u>. The Employer shall determine and certify to the Trustee from time to time as required by the provisions of this Agreement or as requested by the Trustee all pertinent information required for the administration of the Plan of which the Employer has knowledge, including, without limiting the generality and effect of the foregoing:
  - (a) The names of employees eligible from time to time to become Participants in the Plan;
  - (b) The Recognized Compensation of each Participant;
  - (c) The date of birth, date of hire, retirement date, Eligibility Service, and Vesting Service of each Participant;
    - (d) The occurrence of any Event of Maturity or any One-Year Break in Service;
  - (e) The adoption of any rules or regulations with respect to the Plan and the administration, maintenance, maturity, or distribution of benefits thereunder;
  - (f) The occurrence with respect to Participants of temporary layoffs, leaves-of-absence, service with the Armed Forces of the United States, or transfers to other employment;
  - (g) The most recent written statement of a funding policy and method adopted by the Employer which shall be adopted and reviewed from time to time by the Employer and communicated to the Trustee. The Trustee shall not be required to compel the Employer to adopt, review or communicate said funding policy.
  - (h) Any other information useful or necessary to the Trustee in the administration of the Plan.
- 11.2 <u>Rules and Regulations</u>. Any rule not in conflict or at variance with the provisions hereof may be adopted by the Employer and, upon furnishing a certified copy thereof to the Trustee, the same shall thereupon become effective, unless a later effective date shall be specified therein, in which case it shall become effective on the date so specified.

### 11.3 <u>Method of Executing Instruments.</u>

- (a) Certifications of fact or written notices to be made or consents to be given by the Employer, or the Retirement Committee if delegated such authority, pursuant to any provision of this Plan may be signed in the name of the Employer or Retirement Committee, as may be the case, by any officer thereof who has been authorized to make such certification or to give such notices or consents and may be relied and acted upon by the Trustee as authorized, valid and complete in form and in substance and as made with the authority of the Employer until written notice of termination of such authority with respect to any such signing officer shall be given by the Employer to the Trustee.
- (b) Any instrument, certification of fact, or written notice required, necessary or advisable to be made or given by the Trustee may be signed by any authorized officer or employee of the Trustee, and each and every such instrument, when so signed and delivered, shall be valid and binding.

- 11.4 <u>Claims Procedure</u>. Except as otherwise provided herein, the Employer shall establish a procedure for the resolution of disputes and disposition of claims arising under this Plan. Until modified by the Employer, this procedure is as follows:
  - (a) <u>Original Claim</u>. Any Employee, former Employee or Beneficiary of such Employee or former Employee may, if he so desires, file with the Employer a written claim for benefits under this Plan. Within ninety (90) days after the filing of such a claim, the Employer shall notify the claimant, in writing, whether his claim is upheld or denied in whole or in part, or notify the claimant in writing of the specific special circumstances requiring a specified amount of additional time (but not more than one hundred eighty (180) days from the date the claim was filed) to reach a decision on the claim. If the claim is denied in whole or in part, the Employer shall state in writing:
    - (i) The specific reasons for the denial;
    - (ii) Specific references to the pertinent provisions of the Agreement on which the denial is based;
    - (iii) A description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary; and.
      - (iv) An explanation of the claim review procedure set forth in this Section.
  - (b) <u>Claim Review Procedure</u>. Within sixty (60) days after receipt of notice that his claim has been denied in whole or in part, the claimant may file with the Employer a written request for a review and may, in conjunction therewith, submit written issues and comments. Within sixty (60) days after the filing of such a request for review, the Employer shall notify the claimant, in writing, whether upon review the claim was upheld or denied in whole or in part or furnish claimant a written notice describing specific special circumstances requiring a specified amount of additional time (but not more than one hundred twenty (120 days from the date the request for review was filed) to reach a decision on the request for review. The Employer's decision on the request for review shall be served on the claimant in writing.

#### (c) General Rules.

- (i) No inquiry or question shall be deemed to be a claim for a request for a review of a denied claim unless made in accordance with the claim procedure.
- (ii) All decisions on claims and on requests for reviews of denied claims are made by the Employer. The Employer may require that any claim for benefits and any request for a review hereunder must be filed on forms to be furnished by the Employer upon request.
- (iii) The Employer may, in its discretion, hold one or more hearings on a claim or a request for a review of a denied claim.
- (iv) Claims may be represented by a lawyer or other representative, but the Employer reserves the right to require the claimant to furnish written authorization.
- (v) If a decision or notice is not received by a claimant within the time specified, the claim or request for review of a denied claim shall be deemed to have been denied.
- (vi) Prior to filing a claim, or a request for a review of a denied claim, the claimant or his representative shall have a reasonable opportunity to review a copy of the Plan and all other pertinent documents in the possession of the Employer.
- 11.5 <u>Information Furnished by Participants.</u> Neither the Employer, nor the Retirement Committee, nor the Trustee shall be liable or responsible for any error in the computation of the Total Account or Matured Account of a Participant resulting from any misstatement of fact made by the Participant, directly or indirectly, to the

Employer, the Retirement Committee, or the Trustee and used by them in determining his Total Account, or Matured Account, and neither the Employer, nor the Retirement Committee, nor the Trustee shall be obligated or required to increase the Total Account or Matured Account of such Participant which, on discovery of the misstatement, is found to be understated as a result of such misstatement by the Participant, but the Total Account or Matured Account of any Participant which is overstated by reason of any such misstatement shall be reduced to the amount appropriate for him in view of the truth. Any refund received on reduction of a Total Account or Matured Account so made shall be treated in the same manner as Forfeitures in the Adoption Agreement.

#### **SECTION 12**

#### Plan Administration

- 12.1 <u>Trustee</u>. Except to the extent provided herein or in Section 4 or 10, the Trustee shall have the exclusive authority to manage and control the assets of the Fund and their custody.
- 12.2 <u>Delegation of Duties</u>. Functions generally assigned to the Employer shall be discharged by its officers or may be delegated and allocated as provided herein. Except as provided in Section 12.5, the Employer may delegate or redelegate and allocate and reallocate to one or more persons or to a committee of persons jointly or severally, and whether or not such persons are officers or employees, such functions assigned to the Employer hereunder as it may from time to time deem advisable.

Without limiting the generality of the foregoing, functions which may be delegated or redelegated include:

- (a) To act for the Employer to employ and supervise the doctor of medicine or any other person or employee for the purposes of this Plan;
- (b) To act for the Employer to make the determinations and certifications contemplated in this Agreement;
- (c) To act for the Employer to prepare, distribute, receive and maintain the forms and records required by the Plan;
- (d) To act for the Employer to prepare and file the necessary reports and documents with governmental agencies or make required disclosures to Participants, Beneficiaries or other employees;
- (e) To act for the Employer to adopt and establish the rules and procedures authorized under this Agreement and to interpret them;
- (f) To act for the Employer in consultation with the Trustee or other qualified person to establish and review a funding policy and method and to communicate the same to the Trustee; and,
- (g) To contract with or appoint others in writing (on behalf of the Plan) to assume functions expressly delegated or allocated or redelegated or reallocated to them and to specify in writing the scope of and any limitations of their authority.
- 12.3 <u>Retirement Committee</u>. The Employer reserves the power to create and establish at any time a Retirement Committee, which shall serve as the Plan Administrator, of such size and composition as the Employer may from time to time determine to carry out fiduciary and administrative responsibilities under the Plan in accordance with the following rules:
  - (a) If a Retirement Committee is created, the Employer may delegate to it any delegable fiduciary and administrative responsibilities (other than responsibility to manage or control the assets of the Plan) to the Retirement Committee. Without limiting the generality of the foregoing, the Employer may delegate to the Committee any or all of the fiduciary and administrative responsibilities permitted to be delegated or allocated pursuant to Section 12.2. In addition, the Retirement Committee (or Employer if no

Retirement Committee has been established) shall have the power described in Sections 10.14(g) and 10.14(n).

- (b) The exercise of any such power by the Committee and the certification thereof by a member of that Committee shall have the same force and effect as if such action were taken by the Employer.
- (c) With respect to each action which the Retirement Committee may take pursuant to the provisions hereof, the Trustee shall be relieved of all liability and be fully protected in acting in reliance on any advice of the Retirement Committee so given by it or in conformity to any rule or regulation so adopted by it or on the basis of any certification made by it.
- (d) Until authorization and creation of a Retirement Committee and thereafter, to the extent that rule making and other powers as aforesaid shall not be granted to such Committee, or in the event the Employer, after creating a Retirement Committee, shall discontinue the same or reduce its power and authority (the right so to do being hereby reserved to the Employer), such power shall be vested or become revested, as the case may be, in the Employer.
- (e) Any Retirement Committee created as aforesaid shall have power to organize and to delegate to such of its members as it shall select authority to make certifications of fact hereunder and otherwise execute or authenticate rules, advisory opinions or instructions, and other instruments adopted or authorized by the Committee.
- (f) The Retirement Committee may adopt such bylaws or regulations as it deems desirable for the conduct of its affairs and may appoint a secretary, who need not be a member of the Committee, to keep its records and otherwise assist the Committee in the performance of its duties. The Committee shall keep a record of all its proceedings and acts and shall keep all books of account, records and other data as may be necessary for the proper administration of the Plan. The Committee shall notify the Trustee and Employer of any action taken by the Committee, and when required, shall notify any other interested person or persons.
- (g) Members of the Retirement Committee shall serve without compensation, but their reasonable expenses shall be an expense of the administration of the Fund and shall be paid by the Employer or, if not, paid by the Trustee from and out of the Fund.
- (h) Neither the Employer, any member of its Board of Directors, nor the Trustee shall be or become liable for any acts or omissions of the Retirement Committee or any member thereof appointed pursuant to this Section (except to the extent that the Employer, the Board of Directors or the Trustee, as the case may be, may be liable pursuant to the Employee Retirement Income Security Act of 1974, if applicable).
- (i) Police and Fire Plans: Notwithstanding any provision of this Section 12.3 to the contrary, a Retirement Committee shall be established to supervise the general operation of the Plan. The number and members of the Retirement Committee shall be in accordance with the relevant statutes of Nebraska, as amended from time to time. The City shall continue to be responsible for the general administration of the Plan unless specific functions or all functions with regard to the administration of the Plan are delegated, by ordinance, to the Retirement Committee. Whenever duties or powers are vested in the City or the Retirement Committee under the Plan or applicable state law, or whenever the Plan or such law fails to specifically allocate the duties or powers of administration of the Plan, such powers or duties shall be vested in the City unless such powers or duties have been delegated by ordinance to the Retirement Committee.

In addition to those duties delegated to the Retirement Committee in the Agreement, by law or by ordinance, the Retirement Committee shall perform the following duties:

(1) Provide each employee a summary of plan eligibility requirements and benefits provisions;

- (2) Provide, within 30 days after a request is made by a Participant, a statement describing the amount of benefits such Participant is eligible to receive;
- (3) Make available for review an annual report of the Plan's operations describing both the amount of contributions to the Plan from Employee and Employer sources, and an identification of the total assets of the Plan;
- (4) Beginning December 31, 1998, file such reports with the State of Nebraska or its instrumentalities as required from time to time by applicable law. Commencing in 1999, the annual report required to be filed with the Public Employees Retirement Board and the members of the Nebraska Retirement Systems Committee of the Legislature shall include:
  - (i) The number of persons participating in the retirement Plan;
  - (ii) The contribution rates of Participants in the Plan;
  - (iii) Plan assets and liabilities:
  - (iv) The names and positions of persons administering the Plan;
  - (v) The names and positions of persons investing Plan assets;
  - (vi) The form and nature of investments;
  - (vii) A full description of investment policies and options available to Plan participants;
  - (viii) For the defined benefit component of the Plan, if any, the levels of benefits of Participants, the number of Participants eligible for benefits and the total present value of such Participants' benefits, as well as the funding source to pay for such benefits.
- (5) Beginning December 31, 1998, and every four years thereafter, the Retirement Committee shall have a quadrennial report prepared with respect to the defined benefit component of the Plan, if any, and file the same with the Public Employees Retirement Board, with a copy submitted to the members of the Nebraska Retirement Systems Committee of the Legislature. Such report shall consist of a full actuarial analysis of the Plan. The analysis shall be prepared by an independent private organization or public entity employing actuaries who are members in good standing of the American Academy of Actuaries, and which organization or entity has demonstrated expertise to perform this type of analysis and is unrelated to any organization offering investment advice or which provides investment management services to the Plan.

The Retirement Committee of Fire Plans shall also have the following additional duties:

- (1) Elect a chairperson, a vice-chairperson, and such other officers as the Committee deems appropriate;
- (2) Hold regular quarterly meetings and special meetings upon the call of the chairperson;
- (3) Conduct meetings pursuant to Open Meetings Act of the Nebraska Revised Statutes; and
- (4) Provide each Employee a summary of plan eligibility requirements, benefit provisions, and investment options available to such Employee.

Members of the Retirement Committee of a Fire Plan shall, subject to the approval of the City Council, be reimbursed for their actual and necessary expenses incurred in carrying out their duties. Such reimbursement shall be paid from the Unallocated Employer Account to the extent not allocated or assessed pursuant to the schedule of investment costs. Retirement Committee members of a Police Plan shall not be reimbursed for their expenses incurred in carrying out their duties.

- 12.4. <u>Benefits Errors</u>. If the Retirement Committee determines that the Plan has overpaid or underpaid a benefit to a Participant or Beneficiary, it shall have authority to correct the error in accordance with EPCRS or other applicable guidance. In the event of an overpayment, the Committee shall be authorized to, in addition to any other remedy, offset future benefits payments by the amount of the overpayment with Regular Interest. A Participant whose benefit under the Plan is adjusted by the Retirement Committee may request a review by the City Council of such adjustment.
- 12.5 <u>Employer Action</u>. The City Council of the Employer <u>or its designee</u> shall have the exclusive authority, which authority may not be delegated to:
  - (a) Amend this Agreement.
  - (b) Terminate the Plan.
  - (c) Except as otherwise provided elsewhere herein, appoint or remove a Trustee or an Investment Manager.
- 12.6 <u>Limitation on Authority.</u> The Trustee shall not be liable for or on account of any payment made by it pursuant to order of the Employer; or for any investment, retention or sale made by it in accordance with the provisions of this Agreement; or for any loss to or diminution in the Fund resulting therefrom, unless such liability for loss or diminution arises under the provisions of the Employee Retirement Income Security Act of 1974<sub>3</sub>- eif applicable to this Plan by the terms of the Act. No action taken by any person, board or committee, if authority to take such action has been delegated or redelegated to it hereunder, shall be the responsibility of any person except as may be required by the provisions of the Employee Retirement Income Security Act of 1974, if applicable to this Plan by the terms of the Act, relating to the responsibility of fiduciaries for the acts of other fiduciaries.

The responsibility and obligations of the Trustee shall be strictly limited to those set forth in this Agreement. The Trustee shall have no authority or duty to determine the existence, nature or extent of any individual's rights in the Fund or under the Plan or question any determination made by the Employer regarding the same. Except to the extent imposed by provisions of the Employee Retirement Income Security Act of 1974, if applicable to this Plan, no fiduciary shall have the duty to question whether any other fiduciary is fulfilling all of the responsibility imposed upon such other fiduciary by the Act, as the same may be amended from time to time, or by any regulations or rulings issued thereunder and applicable to the Plan. The Trustee shall not be responsible in any way for the manner in which the Employer carries out its responsibilities under this Agreement.

- 12.7 <u>Dual Capacity</u>. Individuals, firms, corporations or partnerships identified herein or delegated or allocated authority or responsibility hereunder may serve in more than one fiduciary capacity.
- 12.8 <u>Administrator</u>. If no other Plan Administrator is named in the Adoption Agreement, then the Employer shall be the Plan Administrator for purposes of Section 3(16)(A) of the Employee Retirement Income Security Act of 1974, to the extent such Section of the Act is applicable to this Plan.
- 12.9 <u>Named Fiduciaries</u>. The Employer shall be the named fiduciary for the purposes of the Employee Retirement Income Security Act of 1974, to the extent the Act is applicable to this Plan.
- 12.10 <u>Service of Process</u>. In the absence of any designation to the contrary, in any legal proceeding, including arbitration, involving the Plan, the City Clerk of the Employer is designated as agent for the receipt of service of process directed to the Plan.
- 12.11 <u>Conflict of Interest</u>. If any member of the Committee or any officer or Employee of the Employer to whom authority has been delegated or redelegated hereunder shall also be a Participant in this Plan, he shall have

Grand Island

no authority as such member, officer or employee with respect to any matter specially affecting his individual interest hereunder, all such authority being reserved exclusively to the other members, officers, or Employees, as the case may be, to the exclusion of such Participant, and such Participant shall act only in his individual capacity in connection with any such matter.

12.12 <u>Residual Authority</u>. In the event the Employer, Committee, or other person designated as having the authority to act or a duty to act on any matter hereunder, is prevented by death, dissolution, incapacity or other similar cause from acting hereunder and there is no other person then empowered to act on such matter, the Trustee shall be empowered to act in its place.

## 12.13 State Reporting.

- 1.7 (a) General. Beginning December 31, 1998, such reports shall be filed with the State of Nebraska or its instrumentalities as required from time to time by applicable law.
  - 1.8
  - 1.9 (b) Police, Fire and Neb. Rev. Stat. Section 19-3501 Municipal Plans.
  - (1) Annual Report. In addition to immediately preceding provisions of this Section 12.13, commencing in 1999, an annual report shall be filed as required for each Police Plan, Fire Plan or other municipal Plan established pursuant to Neb. Rev. Stat. Section 19-3501. The report shall be filed with the Public Employees Retirement Board and the Auditor of Public Accounts and include:
    - (i) The number of persons participating in the retirement Plan;
    - (ii) The contribution rates of Participants in the Plan;
    - (iii) Plan assets and liabilities;
    - (iv) The names and positions of persons administering the Plan;
    - (v) The names and positions of persons investing Plan assets;
    - (vi) The form and nature of investments;
    - (vii) A full description of investment policies and options available to Plan participants;
    - (viii) For the defined benefit component of the Plan, if any, the levels of benefits of Participants, the number of Participants eligible for benefits and the total present value of such Participants' benefits, as well as the funding source to pay for such benefits.
  - (2) Quadrennial Report. In addition to immediately preceding provisions of this Section 12.13, beginning December 31, 1998, and every four years thereafter any required quadrennial report shall be prepared with respect to the defined benefit component of the Plan, if any, and filed with the Public Employees Retirement Board, with a copy submitted to the Auditor of Public Accounts. Such report shall consist of a full actuarial analysis of the Plan. The analysis shall be prepared by an independent private organization or public entity employing actuaries who are members in good standing of the American Academy of Actuaries, and which organization or entity has demonstrated expertise to perform this type of analysis and is unrelated to any organization offering investment advice or which provides investment management services to the Plan.
  - 1.7 (3) The Auditor of Public Accounts may, but is not required to, prepare a review of the reports described in this Section 12.13 in accordance with Nebraska Statutes.

#### **SECTION 13**

#### In General

## 13.1 <u>Disclaimers</u>.

- (a) Neither the terms of this Plan, nor the benefits hereunder, nor the continuance thereof shall be a term of the employment of any Employee, and the Employer shall not be obliged to continue this Plan.
- (b) The terms of this Plan shall not give any Employee the right to be retained in the employment of the Employer.
- (c) Neither the Trustee, nor the Retirement Committee, nor the Employer, nor its officers in any way guarantee the Fund against loss or depreciation, nor do they guarantee the payment of any benefit or amount which may become due and payable hereunder to any Participant or to any Beneficiary or to any creditor of a Participant, a Beneficiary or the Trustee. Each Participant, Beneficiary or other person entitled at any time to payments hereunder shall look solely to the assets of the Fund for such payments or to the Matured Account distributed to any Participant or Beneficiary, as the case may be, for such payments. In each case where a Matured Account shall have been distributed to a former Participant or a Beneficiary or to the person or any one of a group of persons entitled jointly to the receipt thereof and which purports to cover in full the benefits hereunder, such former Participant, or Beneficiary, or such person or persons, as the case may be, shall have no further right or interest in the other assets of the Fund.
- (d) Neither the Retirement Committee, nor the Employer nor any of its officers shall in any manner be liable to any Participant, Beneficiary, or other person for any act or omission of the Trustee (except to the extent that liability is imposed under the Employee Retirement Income Security Act of 1974 and the operative provisions of such Act which impose liability are applicable to this Plan).
- (e) Neither the Trustee, nor the Retirement Committee, nor the Employer or its officers shall be under any liability or responsibility (except to the extent that liability is imposed under the Employee Retirement Income Security Act of 1974 and the operative provisions of such Act which impose liability are applicable to this Plan), for failure to effect any of the objectives or purposes of this Plan by reason of loss or fluctuation in the value of Fund or for the form, genuineness, validity, sufficiency or effect of any Fund asset at any time held hereunder, or for the failure of any person, firm or corporation indebted to the Fund to pay such indebtedness as and when the same shall become due or for any delay occasioned by reason of any applicable law, order, or regulation or by reason of any restriction or provision contained in any security or other asset held by the Fund.
- (f) Except as is otherwise provided in the Employee Retirement Income Security Act of 1974, to the extent such Act is applicable to this Plan, the Employer, its officers, the Trustee, the members of the Committee and other fiduciaries shall not be liable for an act or omission of another person with regard to a fiduciary responsibility that has been allocated to or delegated to such other person pursuant to the terms of this Plan or pursuant to procedures set forth in this Plan.
- 13.2 <u>Duration of Fund</u>. This Plan and the Fund shall continue, if not terminated prior thereto, for the period necessary to develop the benefits intended to be developed hereunder for all original Participants and for all Employees who shall hereafter become Participants hereunder. The Fund from time to time hereunder shall at all times be separate and apart from the assets of the Employer, and no part thereof shall be or become available to the Employer or to creditors of the Employer under any circumstances.
- 13.3 <u>Continuity</u>. If this Agreement is adopted as an amendment of a Prior Plan Statement, the tenure and membership of any committee previously appointed, the rules of administration adopted and the Beneficiary designations in effect under the Prior Plan Statement immediately before the Supplemental Effective Date shall, to the extent not inconsistent with this Agreement, continue in full force and effect until altered as provided herein.

- 13.4 <u>State Law</u>. This Agreement has been executed and delivered in the state of organization of the Employer and has been drawn in conformity to the laws of Nebraska and shall be construed and enforced in accordance with the laws of the state of incorporation of the Employer to the extent not preempted by federal law.
- 13.5 <u>Execution in Counterparts</u>. The Adoption Agreement to this Agreement may be executed in any number of counterparts, each of which, without production of the others, shall be deemed to be an original.

#### **SECTION 14**

#### Accelerated Distributions

- 14.1 <u>Accelerated Distributions</u>. A Qualified Participant, <u>provided this Plan is not a pension plan</u>, may elect to receive distribution under the then Vested amount of his Total Account <u>as necessary to satisfy an immediate and heavy financial need, and subject to the following provisions as are so indicated in the Adoption Agreement.</u>
- (a) The following financial needs are considered immediate and heavy: expenses incurred or necessary for medical care, described in Code § 213(d), of the Qualified Participant, the Qualified Participant's spouse or dependents; the purchase (excluding mortgage payments) of a principal residence for the Qualified Participant; payment of tuition and related educational fees for up to the next 12 months of post-secondary education for the Qualified Participant, the Qualified Participant's spouse, children or dependents; payments necessary to prevent the eviction of the Qualified Participant from, or a foreclosure on the mortgage of, the Qualified Participant's principal residence; payments for funeral or burial expenses for the Qualified Participant's deceased parent, spouse, child or dependent; and expenses to repair damage to the Qualified Participant's principal residence that would qualify for a casualty loss deduction under Code § 165 (determined without regard to whether the loss exceeds 10 percent of adjusted gross income). The last two needs (funeral expenses and home repair) only apply to Plan Years beginning after 2005.
- (b) A distribution will be considered as necessary to satisfy an immediate and heavy financial need of the employee only if:
  - (i) The distribution is not in excess of the amount of the immediate and heavy financial need (including amounts necessary to pay any federal, state or local income taxes or penalties reasonably anticipated to result from the distribution);
  - (ii) The Qualified Participant has obtained all distributions, other than hardship distributions, and all nontaxable loans under all plans maintained by the Employer; and
  - (iii) All plans maintained by the Employer provide that the Qualified Participant's elective deferrals (and employee contributions), if any, will be suspended for 6 months (12 months, for hardship distributions before 2002) after the receipt of the hardship distribution.
  - (a) To reimburse the Participant for the expenses of medical and hospital care attributable to the sickness, accident or other disabling cause affecting him or a member of his family who is dependent upon him for care and support;
- (b) To alleviate a financial hardship or emergency affecting the Participant or his dependent family; or,
  - (c) To defray the cost of the education of any member of the Participant's family who is dependent upon him for care and support; or,
  - (d) To pay, in whole or in part, for the construction, purchase or improvement of a home or homesite for the Participant and his family or to discharge, in whole or in part, a mortgage or other security interest therein.

Each and every such accelerated distribution shall first be made from and charged to the Participant's Voluntary Account, if any, pursuant to Section 7.9 hereof, then against his Rollover Account, if any, and lastly, against his Employer Account.

- 14.2 <u>Qualified Participant</u>. For the purpose of this Section 14, the term Qualified Participant means a Participant who has completed two (2) years of participation under this Plan.
- 14.3 <u>Amount of Accelerated Distribution</u>. For the purposes of this Section 14, the maximum amount which may be distributed as an Accelerated Distribution shall not exceed the value of the then Vested <u>Retirement Valuepercentage</u> of the Participant's <u>Employer</u>—Account<u>s</u> reduced by <u>any prior distributions and</u> the aggregate amount of Employer Contributions credited to the Participant's Employer Account during the two-year period preceding such Accelerated Distribution.
- 14.4 <u>Distributions After Normal Retirement Date</u>. In the case of a Money Purchase Pension Plan or a Profit Sharing Plan, if elected by the Employer in the Adoption Agreement, a Participant may, with the consent of his spouse, if required, elect to receive some or all of his Vested Total Account after such Participant's Normal Retirement Date, notwithstanding that an Event of Maturity has not occurred with respect to such Participant. The amount available for distribution, at any time, shall be determined under Section 14.3.
- 14.5 <u>No Forfeitures.</u> No Forfeitures will occur solely because a Participant receives an Accelerated Distribution.

#### **SECTION 15**

### Uniformed Services Employment and Reemployment Rights Act ("USERRA")

Notwithstanding any provisions of this Agreement to the contrary, contributions, benefits and service credit with respect to qualified military service will be provided in accordance with Code section 414(u).

## 15.1 Heroes Earnings Assistance and Relief Tax Act of 2008 ("HEART Act") Provisions.

- 15.1.1 **Death benefits.** In the case of a death of a Participant occurring on or after January 1, 2007, if the Participant dies while performing qualified military service (as defined in Code Section 414(u)), the Participant's Beneficiary is entitled to any additional benefits (other than contributions or benefit accruals relating to the period of qualified military service) provided under the Plan as if the Participant had resumed employment on the day preceding the Participant's death and then terminated employment on account of death. Moreover, the Plan will credit the Participant's qualified military service as service for vesting purposes, as though the Participant had resumed employment under USERRA immediately prior to the Participant's death.
- 15.1.2 **Benefit accrual**. If, pursuant to a written Plan amendment adopted and executed by the City, the City elects to apply this Section 15.1.2, then effective on or after the effective date specified in said Plan amendment, for benefit accrual purposes, the Plan treats an individual who dies or becomes disabled (as defined under the terms of the Plan) while performing qualified military service with respect to the employer as if the individual had resumed employment in accordance with the individual's reemployment rights under USERRA, on the day preceding death or disability (as the case may be) and terminated said employment on the actual date of death or disability; provided, however, that all such individuals performing qualified military service for the employer (as determined under Code sections 414(b), (c), (m), and (o)) who die or become disabled as a result of performing qualified military service prior to reemployment by the employer shall be credited with service and benefits on reasonably equivalent terms.
  - (a) **Determination of benefits**. The Plan will determine the amount of employee contributions, if any, of an individual treated as reemployed under this Section 15.1.2 for purposes of applying Code Section 414(u)(8)(C) on the basis of the individual's average actual employee contributions for the lesser of: (i) the 12-month period of service with the employer immediately prior to qualified military service; or (ii) if service with the employer is less than such 12-month period, the actual length of continuous service with the employer.

# 15.1.3 **Differential wage payments**. For years beginning after December 31, 2008: an individual receiving a differential wage payment, as defined by Code Section 3401(h)(2), shall be treated as an employee of the employer making the payment, (b) the differential wage payment shall be treated as compensation for purposes of Code Section 415(c)(3) and Regulations Section 1.415(c)-2 (e.g. for purposes of Code Section 415, top heavy provisions of Code Section 416 and determination of highly compensated employees under Code Section 414(q), to the extent said provisions are applicable to the Plan), and (c) the Plan shall not be treated as failing to meet the requirements of any provision described in Code Section 414(u)(1)(C) (or any corresponding Plan provisions, including, but not limited to, Plan provisions related to the average deferral percentage or average contribution percentage, to the extent applicable) by reason of any contribution or benefit which is based on the differential wage payment. Differential wage payments (as described herein) shall constitute compensation for all Plan purposes. ---Provided. (i) Nondiscrimination Requirements. however, for purposes of subparagraph (c), all employees of the employer (as determined under Code Section 414(b), (c), (m) and (o)) performing service in the uniformed services described in Code Section 3401(h)(2)(A) shall be entitled to receive differential wage payments on reasonably equivalent terms and, if eligible to participate in a retirement plan maintained by the employer, to make contributions or accrue other benefits, if contributions or other benefit accruals are permitted or provided, based on the payments on reasonably equivalent terms (taking into account the provisions of Code Section 410(b)(3), (4) and (5) to the extent applicable). 15.1.4 **Deemed Severance**. The Plan does not permit distribution upon deemed severance of employment. CITY OF GRAND ISLAND, NEBRASKA, Employer By: \_\_\_\_ (Printed Name) (Title) Date: WELLS FARGO BANK, Trustee By: (Printed Name) (Title)

105

Date:

## APPENDIX A

## **Trustee Fees**

As determined under a service agreement adopted between the City and the Trustee from time to time, which is incorporated herein by this reference.

#### APPENDIX B

## **Actuarial Equivalencies and Actuarial and Mortality Assumptions**

Unless otherwise specified below, in any group annuity contract, custodial account or other permissible arrangement used to fund the retirement system, or in an addendum to this Plan, (any and all of which are incorporated herein by this reference), the Actuarial Equivalent of a pension or benefit payable under this plan shall be determined with the <u>applicable</u> mortality table and applicable interest rate defined in Code Section 417(e)(3); provided, however, that if benefits are obtained through the purchase of an Annuity Contract, the Actuarial Equivalent shall be determined by the amount of benefit that can be purchased or provided by the Participant's Retirement Value, together with any additional amounts which the Employer may be required to contribute as provided under the Plan. Annuity conversion rates provided in an Annuity Contract and all other actuarial and mortality assumptions shall be on a sex-neutral basis. If the benefit is not paid through the purchase of an Annuity Contract, it shall be paid from the Participant's Account(s).

- A. Notwithstanding anything in this Plan to the contrary and, except as provided in regulations or other guidance of the Pension Benefit Guaranty Corporation (PBGC) and IRS, to the extent applicable to the Plan, the following provisions shall apply in determining the amount payable to a Participant having an annuity starting date in a Plan Year beginning on or after January 1, 2008; provided, however, for purposes of Code Section 415 and related provisions of this Plan, changes to the "applicable mortality table" described in subsection "2" below shall be effective for years beginning after December 31, 2008.
  - 1. Applicable interest rate. For purposes of the Plan's provisions relating to the calculation of the present value of a benefit payment that is subject to Code Section 417(e), as well as any other Plan provision referring directly or indirectly to the "applicable interest rate" used for purposes of Code Section 417(e), the definition of "applicable interest rate" under Code Section 417(e) shall apply, and any provision of the Plan prescribing the use of the annual rate of interest on 30-year U.S. Treasury securities shall be deemed amended and implemented by instead using the rate of interest determined by applicable interest rate described by Code Section 417(e) after its amendment by the Pension Protection Act of 2006. Specifically, the applicable interest rate shall be the adjusted first, second, and third segment rates applied under the rules similar to the rules of Code Section 430(h)(2)(C) for the calendar month (lookback month) before the first day of the Plan Year in which the annuity starting date occurs (stability period), or such other lookback month and stability period as elected by the Employer in a written addendum to this Plan. For this purpose, the first, second, and third segment rates are the first, second, and third segment rates which would be determined under Code Section 430(h)(2)(C) if:
    - (a) Code Section 430(h)(2)(D) were applied by substituting the average yields for the month described in the preceding paragraph for the average yields for the 24-month period described in such section, and
    - (b) Code Section 430(h)(2)(G)(i)(II) were applied by substituting "Section 417(e)(3)(A)(ii)(II) for "Section 412(b)(5)(B)(ii)(II)," and
    - (c) The applicable percentage under Code Section 430(h)(2)(G) is treated as being 20% in 2008, 40% in 2009, 60% in 2010, and 80% in 2011.
  - 2. **Applicable mortality assumption**. For purposes of the Plan's provisions relating to the calculation of the present value of a benefit payment that is subject to Code Section 417(e), as well as any other Plan provision referring directly or indirectly to the "applicable mortality table," the definition of "applicable mortality table" under Code Section 417(e)(3)(B) for the calendar year in which the stability period begins shall apply, and any provision of this Plan directly or indirectly prescribing the use of the mortality table described in Revenue Ruling 2001-62 shall be deemed amended to prescribe the use of the applicable annual mortality table within the meaning of Code Section 417(e)(3)(B), as initially described in Revenue Ruling 2007-67.
  - B. Before the applicable effective dates specified in subsection "A" above, the following provisions

applied:
1. "Applicable interest rate" means the interest rate on 30 year Treasury securities as specified by the Commissioner) for the lookback month for the stability period described below, Unless otherwise elected by the Employer in a written addendum to this Plan, (i) the stability period is the successive period of one calendar month which contains the annuity starting date for the distribution and for which the applicable interest rate remains constant, and (ii) the lookback month is the first calendar month preceding the first day of the stability period. Notwithstanding anything in this paragraph or election of the Employer regarding stability period or lookback month, a plan amendment that changes the date for determining the applicable interest rate (including an indirect change as a result of a change in plan year), shall not be given effect with respect to any distribution during the period ending one year after the later of the amendment's effective date or adoption date, if, during such period and as a result of such amendment, the participant's distribution would be reduced.
2. "Applicable mortality table" means the applicable mortality table under Code Section 417 as set forth in Rev. Rul. 2001-62  Interest Rate:
Mortality Table:

## APPENDIX C

## **Investment Choices**

Investment options permitted under the Plan shall be as specified below or from time to time specified by the Plan Administrator or in other governing documents or instruments, which shall be incorporated herein by this reference

## ADOPTION AGREEMENT

## CITY OF GRAND ISLAND,

## NEBRASKA POLICE OFFICERS' RETIREMENT SYSTEM

## PLAN AND TRUST

# TO BE USED WITH BASIC MUNICIPAL EMPLOYEES PLAN AND TRUST AGREEMENT

## THIS IS TO CERTIFY THAT:

The following was adopted by Ordinance by the City of Grand Island,	Nebraska, b	y its	City
Council, at a meeting thereof duly called and held on	, 20	:	

<u>SECTION 1</u>. Pursuant to Nebraska Statutes, Sections 16-1001 through and including 16-1019 ("Police Retirement Plan Statutes"), the City maintains the City of Grand Island, Nebraska Police Officers' Retirement System Plan and Trust embodied in plan documents including an adoption agreement and basic plan document constituting an integral part thereof, as well as various amendments required by applicable law ("Plan").

SECTION 2, The Plan is required by applicable tax law to be amended and restated into a restated plan document incorporating prior amendments and changes to tax laws, regulations and other guidance, including—without limitation the Pension Protection Act of 2006, Heroes, Earnings Assistance and Relief Tax Act of 2008, and Worker, Retiree, and Employer Recovery Act of 2008, and changes to the Police Retirement Plan Statutes. For this purpose, there has been presented to the City a proposed retirement plan and trust embodied in instruments entitled "Adoption Agreement" together with a "Basic Municipal Employees Plan and Trust Agreement" ("Basic Plan Document") as an integral part thereof (together the Adoption Agreement and Basic Plan Document sometimes are referred to herein together as "Agreements"), which Agreements have been reviewed by legal counsel for the City.

<u>SECTION 3</u>. The City does hereby approve and adopt said Agreements as the amendment and restatement of the Plan, and makes the designations and elections with respect to the Plan as indicated in the Adoption Agreement, to be effective on the date(s) specified in the Adoption Agreement or Basic Plan Document.

<u>SECTION 4</u>. That the Mayor is authorized to execute said Adoption Agreement and Basic Plan Document on behalf of the City, and the City Administrator is authorized and directed to provide the same to the Trustee (for its written

acceptance, if determined necessary or appropriate), and if directed in this Ordinance or otherwise determined necessary or advisable, to cause said Agreements to be submitted, together with such supporting data as may be necessary or advisable and applicable application fee, to the Internal Revenue Service for ruling as to whether the same complies with the pertinent provisions of the Internal Revenue Code of the United States and, in particular, Sections 401(a) and 501(a) thereof, with authority to make any changes in or to the designations, elections or provisions under or of said Adoption Agreement or Basic Plan Document and take such further actions as the City Administrator determines necessary or appropriate to obtain a favorable ruling or as otherwise required for the qualified status of the Plan.

This Adoption Agreement is the Adoption Agreement referred to in the foregoing Ordinance, and the designations and elections hereinafter set forth are those made by the City in accordance with said Ordinance, to-wit:

A. ESTABLISHMENT, EFFECTIVE DATE, AND CITY D
--

(1)	 establishes on,, a Retirement Plan Trust to be known as	anc
	Plan and Trust ("Plan") effective,(Effective Date).	 ctive
	OR	

- (2) amends, restates and continues the City of Grand Island, Nebraska Police Officers' Retirement System Plan and Trust, ("Plan"), originally established on January 1, 1984. This amendment and restatement is effective January 1, 20042, unless otherwise specified herein or in the Basic Plan Document or required under applicable law or regulations or guidance thereunder. (Supplemental Effective Date).
- (3) City's Address:

Street: <u>100 E. 1<sup>st</sup> Street</u>, P.O. Box 1968

City, State, Zip Code: Grand Island, Nebraska 68801

Attention: Ms. Tami Herald Telephone: (308) 385-5444

- (4) Retirement Committee: Kerry Cole, Danny Dubbs, Kelly Mossman, Mark Dreher, David Springer and Dick Rabe, Michael Nelson, Eric Olson, James Olson, todd Dvorak and Jaye Monter subject to such changes from time to time pursuant to Section 12.3(i) of the Basic Plan Document.
- (5) Plan Administrator: the City of Grand Island, Nebraska, with the exception of any administrative functions expressly delegated from time to time to the Retirement

<del>273346-2</del>334457v1

Committee herein or in or under the Basic Plan Document or otherwise by direction of the Mayor and City Council.

- (6) City's Taxpayer Identification No.: 47-6006205
- (7) City's Fiscal Year: October 1 September 30
- (8) The Plan serial number ("PN") assigned to this Plan by the City for reporting and disclosure purposes is: <u>002</u>
- (9) The last day of the Plan Year shall be <u>December 31</u> and the Annual Valuation Date shall be <u>December 31</u> [Sections 1.1.26 and 1.1.4]
- (10) The last day of the Plan's Limitation Year shall be <u>December 31</u> [Section 1.1.20] (All qualified retirement plans maintained by the City shall have the same Limitation Year.)

## B. ELIGIBILITY REQUIREMENTS

- (1) <u>Age Requirement</u>. The minimum age which each Employee must attain before becoming a Participant in the Plan is age N/A.
- (2) <u>Service Requirement.</u> To become a Participant in the Plan, each Employee must complete at least <u>N/A</u> year(s) of Eligibility Service. (*Not Applicable for Police and Fire Plans. Not more than five (5) years for other Plans.)* If year(s) of service selected is or includes a fractional year, an Employee will not be required to complete any specified number of Hours of Service to receive credit for such fractional year.
- The computation period for Eligibility Service will be (Not Applicable for Fire and Police Plans) Check One: NA
   as set forth in Section 1.1.9(a)(i), the year beginning with the date the Employee first performs an Hour of Service and then Plan Years.
   as set forth in Section 1.1.9(a)(ii), based upon years commencing on the date the Employee first performs an Hour of Service\* and anniversaries

Upon reemployment, former Participants shall again participate in the Plan under the immediate reentry rule of Section 2.2.

(4) Plan Entry Date shall be (check one):

thereof.

<del>273346-2</del>334457v1

I

	<u>X</u>	(a)	the first day of service in Recognized Employment with the City ( <i>Police and Fire Plans</i> ).
	_	(b)	the first day <u>of the month</u> (specify period e.g., the week, Plan Year, the month, etc., but not more than six months) following the Employee's satisfaction of the Eligibility Requirements [Section 2.l(d)]
		(c)	the first day of the Plan Year in which the Employee first satisfies the Eligibility Requirements. [Section $2.l(c)$ ]
		(d)	the first day of the first month or the first day of the seventh month of the Plan Year, whichever occurs first, following the Employee's satisfaction of the Eligibility Requirements. [Section $2.l(b)$ ]
		(e)	Other
(5)			<b>Employment.</b> Recognized Employment is service in the of the City in those job classifications indicated below (place "X" on
			cating selection): [Section 1.1.30]
	<u>X</u>	(a)	All Employees of the City employed as police officers.
		(b)	All Employees of the City employed as fire fighters.
		(c)	All Employees of the City as that term is defined in Section 1.1.10.
		(d)	All common law Employees of the City.
		(e)	Salaried Employees of the City.
		(f)	Hourly Employees of the City.
		(g)	Employees who are not covered by any retirement plan established by the City.
		(h)	Other (specify):
(6)	<u>Parti</u>	<u>cipatio</u>	n Election: (check one)
	Empl	oyees a	nd Participants
		have	
	<u>X</u>	do no	ot have (Police and Fire)

<del>273346-2</del>334457v1

# C. MANDATORY EMPLOYEE CONTRIBUTIONS

[Section 3.1]

	(1) <u>Amount of Contribution</u> . The Employee contribution to the Trustee for each Plan Year shall be:				
	X (a) (i) Through September 30, 2013, a sum equal to sSix perc (6%) of his or her Salary,  (ii) Beginning October 1, 2013 through September 30, 2015  sum equal to six and one-half percent (6 ½ %) of his or Salary, and  (iii) Beginning October 1, 2015, a sum equal to seven perc (7%) of his or her Salary. (Police)  (b) Six and one-half percent (6 1/2%) of his or her Salary. (Fire)				
			(b)	Six and one-half percent (6 1/2%) of his or her Salary. (Fire)	
			(c)	Other:	
	(2)	Emplo	oyee Co	ontributions:	
		<u>X</u>	shall (	Police and Fire)	
			shall n	not	
		-	_	by the City and treated as Employer contributions as permitted under a) of the Code.	
D.	VOLU	NTAR	Y EMI	PLOYEE CONTRIBUTIONS	
	<u>X</u>	shall (	Police a	and Fire)	
		shall n	ot		
	be peri	mitted t	o the m	aximum amount allowed under the Internal Revenue Code.	
<b>E.</b>	ALLO	CATIO	ON OF	CITY CONTRIBUTIONS AND FORFEITURES	
				[Sections 3.2 and 6.4]	
<del>273346-2</del> <u>3</u>	334457v1			5	I

(1)	contribution	to the Trustee for each Plan Year shall be: (Select one option only. nks as applicable.)
	<u>X</u> (a)	To the Employer Account of each Participant, a sum equal to 100% of the amounts deducted from the Participant's periodic Salary as Mandatory Employee Contributions above (effective April 16, 2012)Six percent (6%) of each Participant's Salary. (Police)
	(b)	Thirteen percent (13%) of each Participant's Salary. (Fire)
	(c)	Other:
		of Forfeitures: (Select one unless Item I(2)(a) is elected in which vision does not apply.) [Sections 6.4.1 and 6.4.2]
	<u>X</u> (a)	Forfeitures shall first be used to pay administration costs of the Plan and then used to reduce City contributions. ( <i>Police</i> )
	(b)	Forfeitures shall be allocated to the Unallocated Employer Account, and if the Unallocated Employer Account is sufficient to meet Plan liabilities, then forfeitures shall first be used to pay expenses of administration and then to reduce City contributions. ( <i>Fire</i> )
	(c)	Forfeitures will be added to the City contribution for allocation.
	(d)	Forfeitures will reduce City contributions.
(2)	Is the Plan in	tegrated with Social Security?
	Yes _X	_ No (Police and Fire)
	(If yes, comp	lete items E, 3-6 and 11; if no, complete items E, 7-11).
		s E, 3-6 and 11 relate to an integrated plan. Contributions are suant to Section 3.3 of the Plan.
(3)	_	Compensation" shall be defined to mean all of each Participant's: <i>indicate selection</i> )
	(a)	W-2 earnings; or,
	(b)	Wages as defined in Code Section 3401(a); or

<del>273346-2</del>334457v1

	(c)	Compe Plan;	ensation as that term is defined in Section 3.6.9(b)(i) of the			
	(d)	shall 3.6.9(b	led, that Recognized Compensation defined in (a) through (c) include amounts described in Sections 3.6.9(b)(vi) and b)-2 as "default provisions" unless otherwise elected below all that apply):			
		(1)	Exclude leave cashouts and deferred compensation (Section $3.6.9(b)-3(b)$ )			
		(2)	Include military continuation payments (Section 3.6.9(b)- $3(c)$ )			
		(3)	Include disability continuation payments (Section 3.6.9(b)- $3(d)$ ):			
			(a) For nonhighly compensated Employees only; or			
			(b) For all Employees and the salary continuation will continue for the following			
		(4)	fixed or determinable period Apply the administrative delay ("first few weeks") rule (Section 3.6.9(b)-4); and/or			
		(5)	Include "deemed" section 125 compensation pursuant to 3.6.9(b)-vi.			
	(e)	Other	5.0.5(b) vi.			
which	is actua	ılly paid	I to the Participant during			
	the Pla	an Year				
	the tax	able ye	ar ending with or within the Plan Year			
	_ the Limitation Year ending with or within the Plan Year.					
	a consecutive 12-month period ending with or within the Plan Year beginning with the day of(enter month).					
Recog	nized C	ompens	sation			
	shall in	nclude				
	shall not include					

<del>273346-2</del>334457v1

402(e)(3), 402(h), 403(b) or 457 of the Code. If an Employee participates in the Plan for only a portion of the year, his (4) Recognized Compensation for the year [check one]: shall shall not include otherwise Recognized Compensation during the portion of the year during which he was not a Participant in the Plan. "Integration Level" is defined as (place "X" next to definition selected and (5) *complete appropriate blanks)* For any Plan Year, an amount equal to \$\_\_\_\_\_(insert stated (a) dollar amount not to exceed the Taxable Wage Base in effect at the beginning of the Plan Year). (b) For any Plan Year, an amount equal to \_\_\_\_% (not more than 100%) of the Taxable Wage Base in effect at the beginning of the Plan Year. For any Plan Year, an amount equal to the lesser of: (i) (c) (insert stated dollar amount); or (ii) the Taxable Wage Base in effect beginning at the Plan Year. (6) Participants who have been credited with a Year of Service for a Plan Year but who terminate employment before the last day of the Plan Year (check one): shall shall not share in the City contribution and reallocation of the forfeitures for that Plan Year. If shall not is elected, designate any exceptions that apply: death retirement at or after Normal Retirement Date disability.

Employee contributions picked up by the City under Section 414(h), and City contributions made pursuant to a salary reduction agreement which are not includable in the gross income of the Employee under Sections 125, 132(f)(4),

(7) Subject to an exclusion in limitations in Item (7) or (8) or as otherwise provided in Section 1 of the Basic Plan Document, "Recognized Compensation" shall be defined to mean all of each Participant's (place "X" to indicate selection, check only one) W-2 earnings; (a) (b) Wages as defined in Code Section 3401(a); or (c) Compensation as that term is defined in Section 3.6.9(b)(i) of the Plan: (d) Provided, that Recognized Compensation defined in (a) through (c) shall include amounts described in Sections 3.6.9(b)(vi) and 3.6.9(b)-2 as "default provisions" unless otherwise elected below (select all that apply): (1) Exclude leave cashouts and deferred compensation (Section 3.6.9(b)-3(b)Include military continuation payments (Section 3.6.9(b)-(2) (3) Include disability continuation payments (Section 3.6.9(b)-*3(d)):* For nonhighly compensated Employees (a) only; or For **Employees** (b) all and the salary continuation will continue for the following fixed or determinable period Apply the administrative delay ("first few weeks") rule (4) (Section 3.6.9(b)-4); and/or (5) Include "deemed" section 125 compensation pursuant to 3.6.9(b)-vi. X Salary as that term is defined in Section 1.1.36(a) of the Plan (e) (Police); Salary as that term is defined in Section 1.1.36(b) of the Plan (f) (Fire); or

NOTE: Items E, 7-11 relate to a nonintegrated plan. Contributions are allocated

directly on Recognized Compensation. (Section 3.3(b)).

		(f)	Other
	which	ı is actu	ally paid to the Participant during
	<u>X</u>	the Pl	an Year.
		the tax	xable year ending with or within the Plan year.
		the Li	mitation Year ending with or within the Plan year.
	-	-	compensation (Police and Fire Plans see definition of "Salary" in 6 of the Basic Plan Document)
		shall i	include
		shall 1	not include
	City of include	contribu lable in	entributions picked up by the City pursuant to Section 414(h), and ations made pursuant to a salary reduction agreement which are not the gross income of the Employee under Sections 125, 132(f)(4), 2(h), 403(b) or 457 of the Code.
(8)		_	Compensation" shall <u>not</u> include: (place "X" to indicate exclusion(s) blank, if applicable) <b>N/A</b>
		(a)	overtime, shift, holiday and vacation pay
		(b)	bonuses
		(c)	commissions, but not more than the first \$thereof (insert dollar limitation desired, if any)
		(d)	overtime, callback pay, clothing allowances and other such benefits reported on Employee federal withholding statement ( <i>Fire</i> ).
(9)		-	yee participates in the Plan for only a portion of the year, his Compensation for the year [check one]:
		shall	
	<u>X</u>	shall 1	not

<del>273346-2</del>334457v1

			e otherwise Recognized Compensation during the portion of the year during he was not a Participant in the Plan.
	(10)		pants who have been credited with a Year of Service for a Plan Year but erminate employment before the last day of the Plan Year. (Check one):
		<u>X</u>	shall (Police and Fire)
			shall not
			in the City contribution and reallocation of forfeitures for that Plan Year. If not is elected, designate any exceptions that apply:
			death
			retirement at or after their Normal Retirement Date
			disability
	(11)	Forfeit	tures will be reallocated [Sections 6.4.2 and 6.4.3]
		<u>X</u>	as of the following Valuation Date
			after a Participant incurs 5 consecutive One Year Breaks in Service or his Account is no longer subject to restoration.
F.	INTE	REST	
	<u>X</u>	shall	
		shall n	ot
	be pai	d on Em	aployer Contributions pursuant to Section 3.2(a)(ii) of the Plan.
G.	WITH	HDRAV	VAL OF PRIOR VOLUNTARY CONTRIBUTIONS
	If this		or a predecessor plan previously permitted Voluntary Contributions,
		are	
	<u>X</u>	are no	t
	-	tted to on 7.9]	withdraw their voluntary contributions before an Event of Maturity.
273346	<del>2</del> 334457v1		11

## H. **ROLLOVERS** Rollover contributions by Participants [Section 3.5] (1) are permitted as specified in Section 3.5. In addition to the plans specified in $\mathbf{X}$ Section 3.5, rollover contributions and direct rollovers may be made from the following types of plans as of the specified effective date(s) (specify all that apply): (a) annuity contract described in Code section 403(b), effective for X distributions after \_\_\_\_\_ (December 31, 2001 if no date specified) (b) eligible plan under Code section 457(b) which is maintained by X a state or political subdivision of a state, or agency or instrumentality of a state or political subdivision of a state, effective for distributions after \_\_\_\_\_(December 31, 2001 if no date specified) Including after-tax employee contributions from the plans or contracts checked above, with separate accounting required for amounts includible and not includible in gross income (select if applicable). are not permitted (2) Eligible rollover distribution [Section 7.12] X must

be distributions that are reasonably expected to total \$200 or more during a year.

## I. VESTING OF REGULAR ACCOUNTS

need not

[Section 5]

(1) <u>Employee Accounts</u>: Each Employee is fully vested in his or her Employee Account at all times.

<del>273346-2</del>334457v1

	(2)	Employer Accounts: Each Participant's Employer Account shall become Vested in him as follows (place "X" next to formula selected and complete appropriate blanks):					
		(a)	Full and Immed	liate Vesting. Eac m at all times.	h Employer	Account shall be	
		<u>X</u> (b)		<ul> <li>Each Participant accordance with th</li> </ul>			
		n the Participan ollowing Vestin	t Has Completed ng Service:		The Vested His Regular Will		
	(i)	Five Year Ve	esting:				
		2 years but le 3 years but le	ess than 2 years ess than 3 years ess than 4 years ess than 5 years		100	_% _% _% _% _%	
	(ii)	Seven Year V	Vesting*:				
		2 years but le 3 years but le 4 years but le 5 years but le	ess than 2 years ess than 3 years ess than 4 years ess than 5 years ess than 6 years ess than 7 years		0 0 0 40 60 80	% % % % % % % %	
	(iii)	Seven Year S	Special Vesting (Fin	·e):			
		5 years but le	ess than 5 years ess than 6 years ess than 7 years			% % % % %	
X	(iv)	Ten Year Ve Amended to Year Vesting	_	Through	Begi	nning	

<del>273346-2</del>334457v1

		June 30, 2012	July 1, 2012
		(10 Yr. Graded)	(7 Yr. Graded)
	Less than 2 years	0%	0%
	2 years but less than 3 years	0%	40%
	3 years but less than 4 years	0 %	40%
	4 years but less than 5 years	40 %	60%
	5 years but less than 6 years	50 %	80%
	6 years but less than 7 years	60 %	80%
	7 years but less than 8 years	70 %	100%
	8 years but less than 9 years	80 %	
	9 years but less than 10 years	90 %	
	10 years or more	100 %	
	Participant		the date upon which the ars. (Insert age, but not 44(e)]
(4)	Notwithstanding any of the forego be 100% Vested in him upon his a		s Employer Account shall
	_ <u>X</u> _ (a) 60 ( <i>Police</i> )		
	(b) 55 (Fire)		
	(c) Other		
	years of age while in the employs Police Plan. (If no age is entere was intended.)		

[Section 5.1.2]

(5) Normal Retirement Date is: (place "X" next to choice selected)

[Section 1.1.22]

\_\_ (a) The Participant's 65th birthday.

		(b) The Participant's 65th birthday or, if later, the 5th anniversary of the date the Participant first becomes a Participant.					
		X (c) Age 60 years (60 for Police; 55 for Fire; Otherwise not greater than the Participant's 65th birthday and not less than age 55.					
	(6)	Early Retirement Date is age <u>N/A</u> years. (Specify age. In-service distribution upon attaining early retirement date is not allowed for a pension plan. Also, this provision is a Not applied by for Relies (Fire Plans, and Section 1.1.24 of Pagis					
		<u>provision is nNot applicable for Police/Fire Plans – see Section 1.1.34 of Basic Plan for definition of Early Retirement Date for Police or Fire Plan.)</u>					
	(7)	An Employee who returns to employment of the City in Covered Employment after terminating service					
		X shall					
		shall not					
		be permitted to restore his or her Employer Account to the amount on the date of distribution. (Section 5.1.3 or 6.4.4)					
J.	INVE	ESTMENT DIRECTIONS					
	(1)	Participants:					
		are					
		X are not					
	permit	nitted to direct the investment of a portion of their accounts into life insurance.					
		[Section 10.10]					
	(2)	Participant Account Investment Direction [Section 10.11.2]					
		(a) Participants:					
		X are					
		are not					
		permitted to direct the investment of their:					
		X Employee Accounts (Required of Police and Fire Plans)					

<del>273346-2</del>334457v1

	X Employer Accounts (Permitted for all types of plans, with the exception of pre-1984 contributions under Police and Fire Plans, and the Employer Account of pre-1984 hires under Fire Plans.)
	The City agrees to indemnify the Trustee and hold it harmless for the Trustee's actions taken pursuant to such direction. (Sections 1.1.35, 4.3 and 10.11)
	(b) Separate Investment Accounts ( <i>Police</i> )
	The City
	X may
	may not
	direct the establishment of separate investment accounts for each Participant to allow each Participant to direct the investment of all or a portion of his or her Employee or Employer Account
	If in the affirmative, enter name or title of person (or committee) authorized to communicate such directions to the Trustee: Retirement Committee. Such directions shall be in writing and the City agrees to indemnify the Trustee and hold it harmless for the Trustee's actions taken pursuant to such directions.
(3)	Investment Direction [Sections 4.3, 10 and 12]
	(a) The
	City
	may
	may not
	X Retirement Committee (Police and Fire)
	<u>X</u> may
	may not
	direct the Trustee in the investment management of Plan assets.
(4)	Participant Loans: [Section 10.13]
	are permitted
<del>273346-2334457v1</del>	16

Special Meeting - 12/30/2013

X	are not	permitted

## K. INTERNAL REVENUE CODE SECTION 415 LIMITATIONS

[Section 3.6]

If the City maintains or ever has maintained another qualified plan in which any Participant in this Plan is (or was) a Participant or could possibly become a Participant, the City must complete this Section. City must also complete this Section if it maintains a welfare benefit fund, as defined in Code section 419(e), an individual medical account, as defined in Code section 415(1)(2), or simplified employer pension, as defined in Code section 408(k) under which amounts are treated as annual additions with respect to any Participant in this Plan. (Designate whether (1) or (2) applies, and complete as appropriate.)

If a Participant is covered by another qualified plan maintained by the City: the provisions of Sections 3.6.4 through 3.6.9 will apply; X (1) OR (2) set forth the method under which the Plans will limit total annual additions or distributions to the maximum permissible amount or benefit, as applicable, and will properly reduce any excess amounts or benefits, in a manner that precludes City discretion. (Use additional continuation pages if alternative limitation rules are to be specified.) 415 Compensation. The definition of Compensation for Code section 415 purposes shall be 415 Safe Harbor Compensation defined in section 3.6.9(b)i of the Plan. unless an alternative definition of compensation is elected below pursuant to section 3.6.9(b)ii of the Plan (select desired alternative definition): (1) W-2 Wages; or X (2) Section 3401(a) Wages Modifications to 415 Compensation - Compensation for 415 purposes shall include amounts described in Sections 3.6.9(b)-2 and 3.6.9(b)(vi) as "default provisions" unless otherwise elected below (select all that apply). Exclude leave cashouts and deferred compensation (Section (1) 3.6.9(b)-3(b)\_\_ (2) Include military continuation payments (Section 3.6.9(b)-3(c))

<del>273346-2</del>334457v1

I

			(4)	(a) (b) Apply	For nonhighly compensated Employees only For all Employees and the salary continuation will continue for the following fixed or determinable period  the administrative delay ("first few weeks") rule (Section 3.6.9(b)-4)  le "deemed" section 125 compensation pursuant to 3.6.9(b)- vi, effective for limitation years beginning on or after January 1, 1998.
L.	CREI	DITING	S SERV	ICE (C	Complete (1) and (2), as appropriate.)
	X	(1)	below Servic Servic in the	, for the se [Sector City contined by	e purpose of determining an Employee's One-Year Breaks in tion 1.1.23], Vesting Service [Section 1.1.44], Eligibility ion 1.1.9] and minimum annual service requirement to share ntribution made for a Plan Year [Section 3.3], service will be y reference to Hours of Service according to the following:
			<u>X</u>	(a)	On the basis of the actual recorded hours for which an Employee is paid or entitled to payment.
				(b)	On the basis that, without regard to his actual recorded hours, an Employee shall be credited with 10 Hours of Service for a day if under Section 1.1.18 such Employee would be certified with at least one hour of service during that day.
				(c)	On the basis that, without regard to his actual recorded hours, an Employee shall be credited under 45 Hours of Service for a calendar week if under Section 1.1.18 such Employee would be credited with at least One Hour of Service during that calendar week.
				(d)	On the basis of semimonthly payroll periods, an Employee shall be credited with 95 Hours of Service for a semimonthly payroll period if under Section 1.1.18 such Employee would be credited with at least one Hour of Service during that semimonthly payroll period.
				(e)	On the basis that, without regard to his actual recorded hours, an Employee shall be credited with 190 Hours of Service for a calendar month if under Section 1.1.18 such Employee would be credited with at least one Hour of Service during that calendar month.

X	(2) <u>Elapsed Time Method</u> . [Section 1.2] Notwithstanding (1) above, service will be credited based upon elapsed time for the following purposes: (check as appropriate)				
			(a)	None	
			(b)	All	
			(c)	Eligibility (and Eligibility Breaks in Service)	
		<u>X</u>	(d)	Vesting (and Vesting Breaks in Service)(Police and Fire)	
			(e)	Minimum Service for benefit accrual for a Plan Year	
INVE	STME	NTS			
				[Section 10.14(p)]	
(1)	All funds of a Plan for police officers or fire fighters must be invested pursuant to the policies established by the Nebraska Investment Council.				
(2)	The Trustee's collective investment fund or funds are incorporated by reference into this Agreement, as indicated in Appendix "C" of the Basic Municipal Employees Plan and Trust, or otherwise agreed by the parties in writing from time to time.				
(3)	The Trustee is hereby specifically authorized and empowered to invest Plan assets in deposit accounts of <u>Wells Fargo Bank N.A.</u> which bear a reasonable rate of interest and securities offered by <u>Wells Fargo Bank N.A.</u> (name of Trustee or financial institution). Such specification shall be permitted in any other applicable document related to funding the Plan, which document shall be incorporated herein by this reference.				
				[Section 4.2]	
(4)				ation of Accounts. Participant accounts will be valued for as follows: (select one)	
			(a)	Annually	
			(b)	Semi-annually	
			(c)	Quarterly	

<del>273346-2</del>334457v1

Μ.

N.	ACCELERATED DISTRIBUTIONS					
	Plan	Particip	ag optional provisions for accelerated distributions may be made available to pants: (Select as many as shall apply. Not applicable for Police and Fire (1) through (4) not applicable if the Plan is a pension plan.) [Section 14]			
		(1)	Medical emergency			
		(2)	Financial hardships			
		(3)	Education expenses			
		(4)	Purchase of home			
		(5)	After Normal Retirement Date			
О.	The capply case in	optional . If the if this H nay not	FORMS OF DISTRIBUTION  forms of benefit payment provided by this Plan are: (Select as many as e requirements of Code section 411(d)(6) are applicable (which is not the Plan is a government plan within the meaning of Code section 414(d)), the teliminate optional forms of payment for benefits which have accrued prior			
	to the	date of	f Plan amendment unless said requirements are satisfied.)			
	<u>X</u>	(1)	a single sum			
	<u>X</u>	(2)	straight life annuity			
	<u>X</u>	(3)	straight life annuity with a guarantee of at least 60 monthly payments			
	<u>X</u>	(4)	annuity payable for life of Participant and annuity to surviving beneficiary of 100%, 75% or 50% as elected by the City.			
		(5)	a combination of (1) through (4)			
		(6)	if this Plan is a transferee plan, an optional form of distribution provided			

(d)

(e)

X

Monthly

Daily

<del>273346-2</del>334457v1

under the transferor plan which is required to be preserved under Code section 411(d)(6) (and the regulations issued thereunder – which is not the case if this is a government plan under Code section 414(d)) with respect

	name of transferor plan and date on which prior accrued benefit distribution options are protected)
(7)	Other (Describe):

# P. MANDATORY DISTRIBUTION ALTERNATIVES (Section 5.1.3(a)(ii)

The following election is made in lieu of the provisions of Section 5.1.3(a)(ii) reducing the mandatory cash out amount to an amount equal to or less than \$1,000 (Option for plans other than Fire Plans — Participant consent to distribution is always required for Fire Plans.) (select one):

- \_\_\_\_ (1) No Mandatory Cash-Outs. An Employee's Vested Retirement Value will not be distributed upon the Employee's termination of service without the Employee's prior written consent, regardless of amount.
- X (2) Increased Mandatory Cash Out Amount. Upon termination of service, the Employee's Vested Retirement Value will be automatically distributed without the Employee's consent if the Vested Retirement Value is less than \$3,500 if a Police Plan (or not in excess of \$5,000 for plans other than Police or Fire Plans). Said distribution, if greater than \$1,000, will be paid in a direct rollover to an "individual retirement plan" designated by the Plan Administrator if the Employee does not elect to have the distribution paid in a direct rollover directly to an "eligible retirement plan" specified by the Employee in accordance with direct rollover provisions of the Plan, or to receive the distribution directly.
- Q. The City shall periodically pay to the Trustee a fee for services rendered according to the Trustee Fee Schedule attached to the Basic Municipal Employees Plan and Trust Agreement as Appendix A or otherwise agreed to by the parties, as incorporated herein by this reference as amended from time to time. The terms of any agreement adopted and executed by the City and Trustee separate and apart from this document and defining rights and duties of the parties to said agreement shall be supplemental and additional to, and incorporated by reference into, this document to the extent not contrary to terms contained herein; and the same, if entered before the date of this Adoption Agreement, shall continue and remain in effect. If any terms of any such separate trust document conflict with the terms of this document, the terms of this document shall control.

#### R. INITIAL DEPOSIT

<del>273346-2</del>334457v1

In the case of establishment of a new Plan, the City hereby delivers to the Trustee the sum of NA as its initial deposit to establish the Trust, and receipt of the stated sum is hereby acknowledged by the Trustee.

S. The completion of this Adoption Agreement creates certain legal relationships and responsibilities. Accordingly, your legal counsel should review the Plan and Trust prior to the execution of this document so as to insure the suitability of the Plan and Trust for your City.

The City acknowledges that it has consulted with and has been advised by its attorney(s) with respect to the effect of entering this Plan and executing this Adoption Agreement.

Terms used in this Adoption Agreement which are defined in the Plan shall have the meaning given them in the Plan.

The City hereby agrees to the provisions of this Plan and Trust, and, in witness whereof, the City and the Trustee have caused this Agreement to be executed on the date(s) set forth below.

THE CITY OF GRAND ISLAND, NEBRASKA

<del>273346-2</del>334457v1

#### NOTICE TO ADOPTING CITY

Failure to properly fill out this Adoption Agreement may result in disqualification of the Plan.

You may contact a Wells Fargo Bank representative at 304 West 3<sup>rd</sup> Street, Grand Island, Nebraska or by calling (308) 389-4225.

In order to obtain reliance with respect to Plan qualification, the City, upon adopting and executing this Plan, must apply to the Employee Plans Determinations of the Internal Revenue Service for a determination letter.

This Adoption Agreement may only be used with the Basic Municipal Employees Plan and Trust Agreement.

By: \_\_\_\_\_\_
Printed Name Title

Date: \_\_\_\_\_

WELLS FARGO BANK, Trustee

By: \_\_\_\_\_\_
Printed Name Title

Date \_\_\_\_\_

CITY OF GRAND ISLAND, NEBRASKA, Employer

<del>273346-2334457v1</del> 23

ORDINANCE NO. 9463

An ordinance of the Mayor and City Council of the City of Grand Island,

Nebraska to amend and restate the City of Grand Island, Nebraska Police Officers' Retirement

System Plan and Trust; to authorize further actions; and to provide for repeal of conflicting

Ordinances, severability and the effective date hereof.

BE IT ORDAINED BY THE MAYOR AND COUNCIL OF THE CITY OF

GRAND ISLAND, NEBRASKA:

SECTION 1. Pursuant to Nebraska Statutes, Sections 16-1001 through and including

16-1019 ("Police Retirement Plan Statutes"), the City maintains the City of Grand Island,

Nebraska Police Officers' Retirement System Plan and Trust embodied in plan documents

including an adoption agreement and basic plan document constituting an integral part thereof,

as well as various amendments required by applicable law ("Plan").

SECTION 2. The Plan is required by applicable tax law to be amended and restated

into a restated plan document incorporating prior amendments and changes to tax laws,

regulations and other guidance, including the Pension Protection Act of 2006, Heroes, Earnings

Assistance and Relief Tax Act of 2008, and Worker, Retiree, and Employer Recovery Act of

2008, and changes to the Police Retirement Plan Statutes. For this purpose, there has been

presented to the City a proposed retirement plan and trust embodied in instruments entitled

"Adoption Agreement" together with a "Basic Municipal Employees Plan and Trust Agreement"

("Basic Plan Document") as an integral part thereof (together the Adoption Agreement and

Basic Plan Document sometimes are referred to herein together as "Agreements"), which

Agreements have been reviewed by legal counsel for the City.

Approved as to Form ¤

## ORDINANCE NO. 9463 (Cont.)

SECTION 3. The City does hereby approve and adopt said Agreements as the amendment and restatement of the Plan, and makes the designations and elections with respect to the Plan as indicated in the Adoption Agreement, to be effective on the date(s) specified in the Adoption Agreement or Basic Plan Document.

SECTION 4. That the Mayor is authorized to execute said Adoption Agreement and Basic Plan Document on behalf of the City, and the City Administrator is authorized and directed to provide the same to the Trustee (for its written acceptance, if determined necessary or appropriate), and if directed in this Ordinance or otherwise determined necessary or advisable, to cause said Agreements to be submitted, together with such supporting data as may be necessary or advisable and applicable application fee, to the Internal Revenue Service for ruling as to whether the same complies with the pertinent provisions of the Internal Revenue Code of the United States and, in particular, Sections 401(a) and 501(a) thereof, with authority to make any changes in or to the designations, elections or provisions under or of said Adoption Agreement or Basic Plan Document and take such further actions as the City Administrator determines necessary or appropriate to obtain a favorable ruling or as otherwise required for the qualified status of the Plan.

<u>SECTION 5</u>. All ordinances and parts of ordinances as previously enacted that are in conflict with this Ordinance or any part hereof are hereby repealed.

SECTION 6. If any section, subsection, sentence, clause or phrase of this Ordinance is, for any reason, held to be unconstitutional or invalid, such unconstitutionality or invalidity shall not affect the validity of the remaining portions of this Ordinance. The Mayor and City Council hereby declare that it would have passed this Ordinance and each section, subsection, sentence,

# ORDINANCE NO. 9463 (Cont.)

clause or phrase thereof, irrespective of the fact that any one or more sections, subsections, sentences, clauses or phrases be declared unconstitutional or invalid.

<u>SECTION 7</u>. This Ordinance shall be in force and take effect from and after passage, approval and publication as provided by law.

Enacted: December 30, 2013.

	Jay Vavricek, Mayor	
Attest:		
RaNae Edwards, City Clerk		